IntechOpen

IntechOpen Series
Education and Human Development, Volume 7

Parenting in Modern Societies

Edited by Teresa Silva



Parenting in Modern Societies

Edited by Teresa Silva

Published in London, United Kingdom

Parenting in Modern Societies http://dx.doi.org/10.5772/intechopen.104241 Edited by Teresa Silva

Contributors

Mulwayini Mundau, Maxwell Chikowe, Lucy Kganyago Mphaphuli, Ethelbert P. Dapiton, Enrique G. Baking, Ranie B. Canlas, Hetty Rooth, Ricanos Jaure, Alfred Henry Makura, Mónica De Martino, Maia Krudo. Anne Marit Vesteraas Danbolt. Teresa Silva

© The Editor(s) and the Author(s) 2022

The rights of the editor(s) and the author(s) have been asserted in accordance with the Copyright, Designs and Patents Act 1988. All rights to the book as a whole are reserved by INTECHOPEN LIMITED. The book as a whole (compilation) cannot be reproduced, distributed or used for commercial or non-commercial purposes without INTECHOPEN LIMITED's written permission. Enquiries concerning the use of the book should be directed to INTECHOPEN LIMITED rights and permissions department (permissions@intechopen.com).

Violations are liable to prosecution under the governing Copyright Law.



Individual chapters of this publication are distributed under the terms of the Creative Commons Attribution 3.0 Unported License which permits commercial use, distribution and reproduction of the individual chapters, provided the original author(s) and source publication are appropriately acknowledged. If so indicated, certain images may not be included under the Creative Commons license. In such cases users will need to obtain permission from the license holder to reproduce the material. More details and guidelines concerning content reuse and adaptation can be found at http://www.intechopen.com/copyright-policy.html.

Notice

Statements and opinions expressed in the chapters are these of the individual contributors and not necessarily those of the editors or publisher. No responsibility is accepted for the accuracy of information contained in the published chapters. The publisher assumes no responsibility for any damage or injury to persons or property arising out of the use of any materials, instructions, methods or ideas contained in the book.

First published in London, United Kingdom, 2022 by IntechOpen IntechOpen is the global imprint of INTECHOPEN LIMITED, registered in England and Wales, registration number: 11086078, 5 Princes Gate Court, London, SW7 2QJ, United Kingdom Printed in Croatia

British Library Cataloguing-in-Publication Data A catalogue record for this book is available from the British Library

Additional hard and PDF copies can be obtained from orders@intechopen.com

Parenting in Modern Societies Edited by Teresa Silva p. cm.

This title is part of the Education and Human Development Book Series, Volume 7

Topic: Human Development

Series Editor: Katherine K.M. Stavropoulos

Topic Editor: Tal Dotan Ben-Soussan

Print ISBN 978-1-80356-788-4 Online ISBN 978-1-80356-789-1 eBook (PDF) ISBN 978-1-80356-790-7 ISSN 2755-9513

We are IntechOpen, the world's leading publisher of Open Access books Built by scientists, for scientists

6,600+

179,000+

195M+

Downloads

Open access books available

International authors and editors

156

Countries delivered to

Our authors are among the

Top 1%

most cited scientists

12.2%

Contributors from top 500 universities



WEB OF SCIENCE™

Selection of our books indexed in the Book Citation Index in Web of Science™ Core Collection (BKCI)

Interested in publishing with us?
Contact book.department@intechopen.com

Numbers displayed above are based on latest data collected.

For more information visit www.intechopen.com



IntechOpen Book Series

Education and Human Development

Volume 7

Aims and Scope of the Series

Education and Human Development is an interdisciplinary research area that aims to shed light on topics related to both learning and development. This Series is intended for researchers, practitioners, and students who are interested in understanding more about these fields and their applications.

Meet the Series Editor



Katherine Stavropoulos received her BA in Psychology from Trinity College, in Connecticut, USA and her Ph.D. in Experimental Psychology from the University of California, San Diego. She completed her postdoctoral work at the Yale Child Study Center with Dr. James McPartland. Dr. Stavropoulos' doctoral dissertation explored neural correlates of reward anticipation to social versus nonsocial stimuli in children with and without autism spectrum

disorders (ASD). She has been a faculty member at the University of California, Riverside in the School of Education since 2016. Her research focuses on translational studies to explore the reward system in ASD, as well as how anxiety contributes to social challenges in ASD. She also investigates how behavioral interventions affect neural activity, behavior, and school performance in children with ASD. She is also involved in the diagnosis of children with ASD and is a licensed clinical psychologist in California. She is the Assistant Director of the SEARCH Center at UCR and is a faculty member in the Graduate Program in Neuroscience.

Meet the Volume Editor



Dr. Teresa Silva is an Associate Professor in Criminology at Mid Sweden University. With a background in health sciences and psychology, the focus of her research has been on child development and the risk factors for children's and adolescents' mental health and problematic behaviour, including delinquency and violence. Initially in Spain, and after in Canada and Sweden, Dr. Silva researched different parenting areas such as child maltreatment,

dysfunctional families, harsh and warm parenting, parental monitoring, parental alienation, and family migration processes, among other topics. In her academic career, she has always advocated for children's rights and the necessity of improving the support and assistance that public administration and private organisations provide for them and their families.

Contents

| Preface | XV |
|---|----|
| Section 1 Outcasts in an Adult World | 1 |
| Chapter 1 Introductory Chapter: Parenting, a Natural and Culturally Defined Role by Teresa Silva | 3 |
| Chapter 2 The Impact of Dysfunctional Families on the Mental Health of Children by Lucy Kganyago Mphaphuli | 11 |
| Chapter 3 Parenting and Work-Family Balance in the Twenty-First Century by Ethelbert P. Dapiton, Enrique G. Baking and Ranie B. Canlas | 23 |
| Chapter 4 Desolate Childhoods: The Surrender of the Adult World – Notes from the Pandemic of SARS-CoV-2 by Mónica De Martino and Maia Krudo | 39 |
| Chapter 5 Children's Rights (Absent) in Universal Parenting Training by Hetty Rooth | 55 |
| Section 2 Migration and Child Rearing | 69 |
| Chapter 6 Communication in Times of Uncertainty and Loss: Refugee Parents in Interaction with Kindergartens and Schools by Anne Marit Vesteraas Danholt | 71 |

| Chapter 7 Mitigating the Unmet Psychosocial Support Needs of Left-behind Learners in Child-Headed Households: Exploring 'Educare' Support at a Zimbabwean School by Ricanos Jaure and Alfred Henry Makura | 83 |
|---|----|
| Chapter 8 The Effectiveness of Foster Care Arrangements for Unaccompanied Migrant Children in Zimbabwe: A Case of Children under Foster Care in Dulibadzimu, Beitbrigde by Mulwayini Mundau and Maxwell Chikowe | 97 |

Preface

Modern societies have evolved to make life easier. Many political systems are (apparently) devoted to attaining a good population level of welfare and social security for families and their children. Over the past 70 years, national and international organisations and agencies have developed with the aim of protecting children's rights, human rights, universal access to education, the health and safety of individuals, and their freedom. Cooperation across nations has made it possible to offer support for those who are most in need as a result of natural or man-made disasters. Furthermore, the continual evolution of technological advances facilitates our ability to communicate across distances. Breaking news almost instantly reaches televisions and mobile devices broadcasted by media corporations and spread on the Internet. Every day we use gadgets developed to simplify daily tasks that used to require great effort in the past. For instance, while they are asleep, parents can now monitor their young children by placing small video and audio-capturing devices in their rooms. Driving children to daycare is currently safer, thanks to special seats. In many countries, education is becoming more digital, and children are taught digital skills from a very young age. In some way, modern societies are societies of the future while, at the same time, dealing with problems of the past. Like in the past, or possibly even more so today, children are adrift in a world that, in many ways, forgets them and does not understand their needs.

This book highlights the challenges that children and their parents face when adjusting to modern life's complexities. The first section opens with an introductory chapter focused on the dual nature of the parental role and how it is defined culturally. The author discusses this duality and highlights elements of evolutionary theory and the importance of the natural and cultural environments.

The impact of being raised in a dysfunctional family on the child's well-being and mental health is the object of debate in Chapter 2, which serves as a timeline reminder of the value of a family system's healthy functioning. Interparental conflict, domestic violence, abuse, neglect, and severed ties between the child and the parent are pointed out as risk factors for a child's behavioural and emotional problems.

Chapter 3 addresses the issue of work–family balance and how today's fast-paced societies, where performance and productivity are highly regarded, affect children and their families. A literature review grounds the discussion of work–family conflict, hustle culture, family vulnerabilities, and the importance of egalitarianism to achieve a necessary balance.

Chapter 4 analyses the plastic expression of Uruguayan children, which uncovers their isolation and loneliness, during the COVID-19 pandemic. Contrary to expectations, the outcomes reveal that children's isolation was not due to government-forced

confinement but rather to the low density of their bonding networks, both internal and external to the family, depleting them of their potential social capital. Adults were portrayed as being sick or armed and ready for violence rather than as being nurturing and supportive.

Chapter 5 examines universal parenting training programs in Sweden, a country with a long tradition of parental assistance, that since the early 2000s have been implemented as a primary preventive effort. The author concludes that the idea of "rights," as they are outlined in the UN Convention on the Rights of the Child, is not an intrinsic value of these programs. Although children may internalise democratic principles as a guide for handling interactions with others, including their parents, the author emphasises how family dynamics and the child's existence, in general, are largely governed by adult norms.

The book's second section collects topics related to themes of migration and child upbringing. Nowadays—perhaps more than ever—many individuals are forced to migrate because of war, violence, poverty, unsustainable living circumstances, and climate change. Moreover, as the world becomes more interconnected, employment and personal ties can also cause relocation to another country. When children are socialised in a culture different from that of their parents, conflict occurs almost inevitably. It is natural for adolescents to rely on their peers, the school, and the community, where they integrate as a reference to construct their system of values, challenging their parents' values. In extreme situations, the parenting role may be severely disrupted and eventually taken over by others when children are left behind in their home countries and towns.

Chapter 6 discusses the challenges that refugee parents encounter when they enter the host country. The inability of parents to communicate with educators due to language barriers makes it difficult for them to interact with the new education system. The author interviewed refugee parents who had recently arrived in Norway and concluded that while both parties have good intentions for cooperation and integration, misunderstandings caused by the situation's complexity create uncertainties. The degree to which children adjust to the parenting behaviour in the new nation depends on whether the uncertainties are resolved in the medium term or result in parents rejecting the new culture.

Migration also comes in other forms. In some countries, parents migrate spurred by poverty, leaving their children behind either in the care of other relatives or, as discussed in Chapter 7, in the care of the older child who assumes parental responsibilities for the young children. In such cases, the community plays a significant role in supporting the child-headed household. Care is provided through schools in addition to education. In this chapter, the outcome of interviews with school personnel and children are discussed regarding the benefits and drawbacks of the "Educare" system, which was put in place in Zimbabwean schools with the intention of ensuring the welfare of children.

Finally, Chapter 8 examines the situation of unaccompanied migrant children in a city that borders South Africa and Zimbabwe, as well as formal and informal

foster care arrangements in place to help the children overcome their more urgent problems. Empathetic cooperation by the community neighbours was a vital resource for the children estranged from their parents. However, the emotional toll of being accidentally or incidentally separated from the care and security that parents are expected to provide will probably leave a permanent impression on the children's lives.

Teresa Silva

Mid Sweden University, Department of Humanities and Social Sciences, Sundsvall, Sweden

Section 1 Outcasts in an Adult World

Chapter 1

Introductory Chapter: Parenting, a Natural and Culturally Defined Role

Teresa Silva

1. Introduction

Parenting is a demanding role. Parental investment in building up a solid parent-child relationship, which warmly supports the child, being constantly attentive to the child's emotional needs and providing consistent discipline, will revert positively in the child's development at the cost of parents' exertion.

For Bowlby [1], parenting is not a job for a single person. In order to do it appropriately and meet the child's needs without getting too exhausted, a caregiver needs assistance. From whom that assistance come varies, depending on societies and cultures. In Northern Europe, the couple is expected to find support in each other, while in South Europe, grandparents play an essential role. In rural areas of China, the grandparents may entirely overtake the parenting role when parents immigrate to find work in the cities, and in some African countries, children of working migrant parents are left in households where the older child takes care of the youngsters with some occasional support from the community. In this regard, we find multiple types of family configurations.

In the framework of family systems theory, families are considered adaptable systems that respond to stressing factors occurring inside and outside the system. The goal of such adaptations is to maintain the family's integrity and autonomy [2]. Three critical dimensions rule the way the family works. First, cohesion among the family members allows the family to be independent and more resilient to adversity. Flexibility, the second dimension, allows a healthy adaptation to changes and, at the same time, keeps stability in a functional manner. The third dimension, communication, facilitates cohesion and flexibility. The family system's integrity and stability are constantly menaced by different factors. Against this background, in modern societies, life entails many challenges for parenting that have the capability to threaten the child's adjustment.

Families cannot thrive alone. As the African proverb says, "It takes a village to raise a child", meaning that the community has a role in providing safety and a healthy environment for children, in order for them to develop appropriately and become well-adjusted adults. Raising a child is not the sole responsibility of their parents or immediate family but rather a collective effort that entails the support and involvement of the wider community [3]. In past traditional societies, every individual in the community had a responsibility to ensure that children were well-fed, healthy, educated and safe. Neighbours and relatives would look out for the youngsters,

3 IntechOpen

offer guidance and share resources to support their well-being and development. Nowadays, the "village" is mainly composed of institutions and organisations in the community, which includes schools, neighbourhood groups, sports clubs, supportive organisations, religious associations, etc. However, we find that the "village" in modern societies is each time more fragmented and disengaged when not broken by political conflicts and war. In many countries, families have to cope with economic pressures, harsh working conditions, professional expectations impossible to achieve, poverty, violence or stigmatisation.

2. An evolutionary perspective

We are biologically prepared to be parents. The selfish gene theory [4] proposes that living organisms are the guardians of the information codified in genes and have evolved to act selfishly to make the information prevail no matter the circumstances. To ensure the survival of the genes and, therefore, the next generation, replication is, in Dawkins's theory, the primary purpose of life. Genes or organisms themselves are not endowed with a conscience that prepares them for strategic replication, but nature will favour those better at replicating.

Strategies for reproduction fall along a continuum [5]. On one extreme of this continuum, we find organisms that reproduce prolifically without much concern or care for the offspring. Many descendants may die, but enough will survive to secure the next generation. On the other extreme, are organisms with a low rate of reproduction but a high investment in raising offspring. The dedication of parents, in this case, maximises the chances that each individual in the progeny survives.

In humans, these different strategies were found by anthropologists to be related to the physical environment. Raine [6] explained this by comparing two ethnographic groups, the!Kung bushmen of the Kalahari desert and the Mundurucú villagers of the Amazon forest. The group living in the extremely difficult conditions of the desert had a low fertility rate and was highly invested in parenting their children, who depended on the quality of parental care for survival. In contrast, the group living in the rich environmental conditions of the Amazon forest, where nature easily provided resources for them, had more descendants, mothers were less caring, men played a minimum role in parenting, and the children had to learn early to fend for themselves.

"Kin selection," a natural tendency to act altruistically in the interest of genetic relatives has been found in many species as well as in humans [7]. Altruistic behaviour is more likely to happen the closer the genetic relationship between individuals is. This explains the assistance that parents may provide for their offspring even when their own lives may be at risk. In this regard, neglecting or intentionally injuring the progeny may seem to defy nature and any evolutionary logic. However, child abuse and neglect in the human species have been overly reported in every culture throughout history. The neo-Darwinian theory identifies four conditions when child abuse and neglect, and in extreme cases, infanticide, by parents, might be favoured: (1) when parents have more children than the necessary resources to rear all of them, (2) when parents do not cooperate with each other in child rearing, (3) in cases of less capable of reproducing offspring, and (4) when the offspring is not genetically related to the parent [8]. In support of these theoretical predictions, research has shown that parental abuse and neglect among humans is more prevalent in situations of single parenthood, large families, multiple birthing and poverty [9–11]. Moreover, children

with severe physical or mental disabilities are more at risk [12], and step-parents are more often perpetrators of violence against the children than biological parents [13].

3. Parenting and environment

Although we can find biological roots in different parenting behaviours, parenting is not the product of a deterministic instinct. Life experiences occurring in parents' childhood, adolescence, before and during the marriage and the experiences with each individual child also determine the way parents raise their children [1]. Environment, as seen earlier, is another strong conditioning factor. For instance, for parents living in war zones, refugee camps, or poverty-stricken areas, the challenges can be overwhelming. The lack of resources, food, clean water or basic medical care forces parents to struggle to provide for their children's basic needs. Parents may have to work long hours in difficult conditions or rely on the generosity of others while exposed to violence and trauma.

Whether it is the threat of war or the daily violence that is commonplace in many impoverished areas, parents must find ways to protect their children from harm while also addressing their emotional needs. At the end of 2021, it was estimated that 36.5 million children worldwide were displaced from their homes due to war, political conflicts and violence [14]. Whether they are dealing with the aftermath of a conflict, the ongoing violence in their communities, or the fear of a potential conflict, parents must navigate difficult terrain when raising their children in such circumstances. In order to ensure their safety, parents may need to take precautions such as finding shelter in safe areas, limiting their children's exposure to traumatic events, and helping them develop coping strategies to deal with the stress and trauma of living in a warzone. Children may also be at risk of violence, exploitation, or recruitment by armed groups. Access to education and opportunities for development is hindered, and countries' economic instability and poverty frustrate any parental endeavour.

Parenting in a world at war demands that parents provide emotional support and help their children develop resilience in the face of adversity. Exposure to significant stress and trauma produced by unsafety environments can lead to depression, anxiety and other mental health problems jeopardising parents' ability to properly parent their children. Eltanamly and colleagues [15] found that in war scenarios, parental warmth decreases while parental harshness increases, compromising children's adjustment, which is linked to an increase in a child's posttraumatic stress symptoms.

Despite these challenges, parents in harsh environments are often incredibly resilient and resourceful. They find ways to adapt to the surrounding and provide their children with the care and protection they need. A strong and supportive social network can favour parenting and child development even in adversarial conditions, and a strong emphasis on cultural traditions and practices helps to maintain a sense of identity and community.

In contrast with harsh living conditions, wealthy societies are characterised by more or less easy access to high-quality education, healthcare and social support systems that make it easier for children to have a stable home as well as opportunities for growth and exploration. Access to high-quality education in well-funded schools where teachers are thoroughly trained and curriculums are well-thought facilitates children's future success. Parents in these environments can support their children's education by staying involved in their academic progress and providing additional resources and support when needed. At the same time, children thrive on structure and routine,

providing them with a sense of security and stability. When parents are able to consistently provide daily routines, including regular meals, bedtimes and study periods, they may find that children react more positively to their parenting practices.

4. Parenting in different cultures

Parenting practices vary widely across cultures, and what is considered "acceptable" in one culture may be viewed differently in others. For instance, parenting in East Asia is often characterised by emphasising academic achievement and discipline. Parents in East Asian cultures focus on children's education and may push them to excel in school while at the same time using strict disciplinary measures, such as physical punishment, to maintain order and obedience. East Asian parents also place a strong emphasis on respect for elders and family hierarchy. Children are often expected to show deference to their parents and grandparents and to prioritise the family's needs over their individual desires.

For parents in Latin America, family and community take the lead. They prioritise social relationships and encourage their children to develop strong bonds with extended family members and neighbours. They often express love and support for their children by demonstrating physical affection, embracing and kissing them. However, parenting in Latin America is also characterised by strict gender roles and expectations. Boys and girls are socialised differently, with boys encouraged to be assertive and independent and girls expected to be nurturing and obedient.

On the other hand, parenting in Africa is diverse and varies widely depending on the specific culture and region. Many African cultures place a strong emphasis on community and family life while promoting respect for elders and traditional values. Children are raised by their parents, and the extended family and community members often play an essential role in their upbringing. Children are expected to show profound deference to their parents and to follow cultural traditions and customs.

Regarding indigenous parenting practices, they also vary widely depending on the specific culture and region. Common among indigenous is the connection to the natural world and to live in harmony with the environment. Children are taught to respect and care for nature and to see themselves as part of a larger community of living beings. Parents in indigenous societies prioritise storytelling and oral tradition as a way of passing down cultural knowledge and values. Children are encouraged to listen to stories and to learn from the experiences of their ancestors.

In contrast to the previous cultures, Western parenting practices are often characterised by affirming individualism and independence. Children's self-esteem is prioritised, and parents encourage them to pursue their individual goals and interests. Positive reinforcement is used as a way of encouraging appropriate behaviour and development. Sometimes, parents in Western cultures are criticised for being overly permissive and lacking discipline. Children may be allowed more freedom and autonomy than in other cultures, which sometimes is seen as a risk factor for behavioural problems and lack of respect for authority.

5. Parenting in modern societies

Modern societies have become more diverse and complex, creating new challenges. Immigration and globalisation have brought people from different backgrounds and values together, making it important for parents to be sensitive to these differences and raise their children accordingly.

Immigrant parents who move to another country with their children face unique challenges. Navigating unfamiliar cultural and social norms while also providing for the family is a daunting experience. Language barriers hinder the ability of parents to communicate effectively with the new social and educational system in the income country. It is, therefore, difficult for parents to advocate for the children's needs and to establish an effective social network with other parents and educators. Immigrant parents also struggle with cultural differences that affect their parenting styles. While in some cultures, parenting is predominantly authoritarian, in others, it is more permissive. Corporal punishment may be a usual educational practice in some cultures, while in Northern Europe, for instance, it is a crime. When immigrant parents are exposed to new cultural norms and expectations, they may feel unsure of how to behave, with a significant risk of social isolation. Moreover, when children are raised in countries with different cultural norms from that of their parents and learn from peers what is acceptable and expected, they may find themselves between two cultures in conflict.

Another significant challenge facing parents in modern societies is the increasing pressure of burdensome work schedules and unattainable professional careers that jeopardise family life. In 2016, the OECD reported that many parents with young children had difficulties balancing their work and family commitments [16]. The report indicated, however, that there were differences between countries. While families from North Europe and Switzerland were more likely to indicate they had little to no trouble juggling work and family, in nations like Chile, Poland, or the Slovak Republic, an average of 70% of parents reported they find it very challenging.

Moreover, in many countries, living costs have risen dramatically, making it necessary for both parents to work. This can generate situations in which parents have limited time and energy to devote to their children, leading to feelings of guilt and stress. Balancing the demands of work and family can be a delicate juggling act that requires careful planning, communication and prioritisation. Maintaining a healthy work-life balance is crucial for creating healthy solid relationships with children, and it helps to reduce stress that otherwise risks impairing both physical and mental health. High demands from work and family can lead to burnout, exhaustion and depression. Children need attention, affection and guidance to develop well-adjusted. Parents who are constantly stressed and overworked find it difficult to provide the emotional support their children need, which may impair the parent-child relationship.

On a different note, modern societies have become more accepting of different family structures and lifestyles, including single-parent households, same-sex families and co-parenting arrangements, which also bring many challenges to parents. The number of children currently living in households with LGBTQ parents has increased sharply over the last few decades, pushed by a sociopolitical change in many countries and a technological development that facilitates assisted reproduction [17]. In recent years, society has started to recognise the rights of LGBTQ families for parenting their children. As more and more same-sex couples adopt children, use assisted reproductive technologies, or have children from previous relationships, parenting by LGBTQ families is becoming increasingly visible. Parenting by LGTBQ families comes with a unique set of challenges, such as discrimination, bullying at school, rejection from extended family members, or even difficulty in finding suitable housing or employment. As Carone and colleagues pointed out, although the family structure seems not

to affect child's developmental outcomes, there are factors regarding stigmatisation and discrimination against their families that affect them [18].

6. Conclusion

In conclusion, parenting is a natural role determined by our biological nature. But parenting is also culturally and environmentally determined. In some cultures, child development is the responsibility of each individual in the community, besides the parents. In more individualistic cultures, parental support is a matter for institutions and organisations to care for. Either way, "it takes a village to raise a child," meaning that families need support to adequately parent their children.

The complexities of modern societies and the multiple environmental challenges parents have to face in their daily lives jeopardise parenting roles and put at risk many children. Prioritising children's development and well-being should be a primary concern for any government and community, a goal not impossible to attain if there is sufficient political commitment to supporting families.

Author details

Teresa Silva Mid Sweden University, Sundsvall, Sweden

*Address all correspondence to: teresa.silva@miun.se

IntechOpen

© 2023 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. CCD BY

References

- [1] Bowlby J. Attachment. New York: Basic Books, Inc. Publishers; 1969. p. 428
- [2] Priest JB. The Science of Family Systems Theory. New York: Routledge; 2021. p. 178
- [3] Reupert A, Straussner SL, Weimand B, Maybery D. It takes a village to raise a child: Understanding and expanding the concept of "village". Frontiers in Public Health. 2022;**10**:1-7. DOI: 10.3389/fpubh.2022.756066
- [4] Dawkins R. The Selfish Gene. Oxford: Oxford University Press; 1976. p. 224
- [5] MacArthur RH, Wilson EO. The Theory of Island Biogeography. Princeton NJ: University Press Princeton; 1967. p. 221
- [6] Raine A. The Anatomy of Violence: The Biological Roots of Crime. New York: Vintage Books; 2013. p. 485
- [7] Hamilton WD. The evolution of social behavior. Journal of Theoretical Biology. 1964;7:1-52. DOI: 10.1016/0022-5193(64)90038-4
- [8] Ellis L. Neodarwinian theories of violent criminality and antisocial behavior: Photographic evidence from nonhuman animals and a review of the literature. Aggression and Violent Behavior. 1998;3:61-110. DOI: 10.1016/S1359-1789(96)00019-5
- [9] Schneider W. Single mothers, the role of fathers, and the risk for child maltreatment. Children and Youth Services Review. 2017;**81**:81-93. DOI: 10.1016/j.childyouth.2017.07.025
- [10] Nikulina V, Widom CS, Czaja S. The role of childhood neglect and childhood

- poverty in predicting mental health, academic achievement and crime in adulthood. American Journal of Community Psychology. 2011;48:309-321. DOI: 10.1007/s10464-010-9385-y
- [11] Land CA, Cox MJ, Flores G. Maltreatment in multiple-birth children. Child Abuse and Neglect. 2013;**3**:1109-1113. DOI: 10.1016/j.chiabu.2013.03.002
- [12] Maclean M, Sims S, Hansen M, Leonard H, Bourke J, Bower C, et al. Abuse and neglect in children with disabilities: Risk varies by type of disability. International Journal of Population Data Science. 2017;1:197. DOI: 10.23889/ijpds.v1i1.197
- [13] Radhakrishna A, Bou-Saada IE, Hunter WM, Catellier DJ, Kotch JB. Are father surrogates a risk factor for child maltreatment? Child Maltreatment. 2001;**6**:281-289. DOI: 10.1177/1077559501006004001
- [14] UNICEF. Nearly 37 million children displaced worldwide highest number ever recorded [Internet]. 2022. Available from: https://www.unicef.org/eap/press-releases/nearly-37-million-children-displaced-worldwide-highest-number-ever-recorded [Accessed: 2023-05-14]
- [15] Eltanamly H, Leijten P, Jak S, Overbeek G. Parenting in times of war: A meta-analysis and qualitative synthesis of war exposure, parenting, and child adjustment. Trauma, Violence, & Abuse. 2021;22:147-160. DOI: 10.1177/1524838019833001
- [16] OECD. Walking in the Tightrope: Background Brief on Parents' Work-Life Balance across the Stages of Childhood.

European Union. Social Policy Division. Better Policies for Better Lifes; 2016. Available from: https://www.oecd.org/social/family/Background-brief-parents-work-life-balance-stages-childhood.pdf

[17] Goldberg AE, Sweeney K, Black K, Moyer A. Lesbian, gay, and heterosexual adoptive parents' socialization approaches to children's minority statuses. Counselling Psychologist. 2016;44:267-299. DOI: 10.1177/001100001562805

[18] Carone N, Bos HMW, Shenkman G, Tasker F. Editorial: LGBTQ parents and their children during the family life cycle. Frontiers in Psychology. 2021;**12**:1-4. DOI: 10.3389/fpsyg.2021.643647

Chapter 2

The Impact of Dysfunctional Families on the Mental Health of Children

Lucy Kganyago Mphaphuli

Abstract

A healthy and nurturing family environment is necessary for the development of mental health in children. A positive atmosphere within the family, such as open communication, strong interpersonal relationships between parents and children, harmony and cohesion, contributes to a conducive and a safe space for children to develop healthy habits. Children who grow up in dysfunctional families are at risk of developing mental illness, which, if not treated, can result in long-term mental health problems such as depression and anxiety. Children who are exposed to constant conflict, aggression, abuse, neglect, domestic violence and separation because of divorce or parents who work long hours away from home are likely to present with behavioural and emotional problems. Parents, whether single, married or divorced, have got the responsibility to protect their children's mental health.

Keywords: dysfunctional families, mental health, mental illness, parenting, parent-child relationship, parental practices

1. Introduction

Mental health of children is a global and persistent concern. It is a multifaceted problem with some of the leading courses being depression, anxiety and behavioural disorders. According to the World Health Organisation [1], one in six people are of ages 10–19, and within this age group, one in seven experience mental health challenges. Children of this age group are at a critical period of developing healthy habits that are necessary for their mental wellness. Being exposed to difficult circumstances at this tender age can compromise their ability to develop healthy mental wellness.

The first year of life is pivotal in the neurological development of children [2]. The childhood experiences during this period can have a positive or negative impact in the development of the brain. Children who are raised in nurturing environments of love, care and support can develop healthy attachments, relationships of trust, security and a good self-esteem. Infants who grow up in unconducive environments characterised by abuse and neglect tend to feel unloved, unappreciated and unwanted. Such children may avoid building intimate and social relationships later in life as they find it difficult to trust other people. They develop fear of their environment and view the world as a dangerous place [2].

11 IntechOpen

Domestic violence is one of the environmental factors that may not be physically directed at children within the family but have a direct impact on them. Children who witness violence at home experience mental, emotional and social challenges that predispose them to mental illness. They are likely to be victims of child abuse and or perpetrators of violence later in their adulthood. The impact of domestic violence on children is likely to manifest in behavioural challenges, low school grades, criminal behaviour and antisocial behaviour [3]. The World Health Organisation [4] estimates that 1 billion children of ages 2–17 have experienced violence of one kind or another, most of which is perpetrated within the home environment. It is in this sense that children are often referred to as silent victims of violence and abuse.

Another environmental factor that affects not only the married couple but children as well is marital breakdown. Divorce brings a lot of devastation, grief and traumatic loss for the children of divorced parents. Logistically and practically, divorce results in single parenting. This is still the case even when in cases of shared custody. The parent who lives with the child carries more responsibility in terms of the day-to-day care and support for the child. More often, parents who bear custody of the children are overburdened financially and logistically, while the other parent might resist and contest reasonable financial contribution towards the needs of the children [5]. The stress of separation between parents can easily be transferred to children, leading to mental health challenges as parents go about creating a new life for themselves, paying less attention to the emotional needs of children. Divorce is, thus, one of the major sources of stress and anxiety in children that can result in mental illness.

Parents have got the responsibility to ensure financial security for their children such as provision of medical care, being able to cater for educational costs, housing, and day-to-day provision for the needs of the family. In most cases, this can be achieved through employment. Parental employment can have both positive and negative effects on parent-child relationships. On the one hand, employment can provide financial stability and a sense of accomplishment that can have a positive impact on the well-being of the family. On the other hand, employment can create stress and time pressures for parents, leading to a strain on parent-child relationships. Parents are likely to bring home the stress of work, which may destabilise the homely environment and further transfer stressful vibes to children.

Growing up in a dysfunctional family has harmful effects that extend to adulthood in children. Children have got no control of the unconducive living conditions created by their parents, caregivers and guardians. Often parents who engage in toxic relationships of violence and abuse are less considerate of the impact of their behaviour on children. They are not aware of the extent of the impact of their actions on children because their aggression is not directed at children, and therefore, they do not think that they are causing emotional harm to children. This unfortunately could not be far from the truth. Negative parenting patterns, such as emotional abuse and neglect, punishment and rejection, create trauma that can result in mental health issue for children.

Some parents come from toxic families themselves where they were exposed to violence, aggression, abuse, neglect, rejection and other negative parenting as children. It becomes difficult for such parents to divorce themselves from their childhood experiences and learn new and positive ways of parenting their own children. Many families are reluctant to accept that they fall in the category of dysfunctional families and thus resist or delay to seek help [6]. Parents are convinced that they are doing well because they are able to provide financially for their children, by so doing, overlooking the negative effects of the toxic environment in which they are raising

children. This circle, if not broken, can be transferred from generation to generation, hurting children up to the edge of mental illness and creating dysfunctional families and communities.

The aim of this chapter is, therefore, to provide information about the relationship between parenting, family dynamics and mental health of children targeting children, parents, families, caregivers and officials who are responsible for proving services to children and families such as social workers, psychologists, and teachers.

2. The impact of growing up in a dysfunctional family

Dysfunctional families have become a huge problem in modern society. While there are no perfect families and people do not choose which family to belong to, the level of dysfunction and lack of coherence in some families are a course for concern. Dysfunctional families are characterised by multiple conflicts, tense relationships, chaos, neglect, abuse, poor communication, lack of empathy and secrecy to an extent that the emotional and physical needs of the family members are not met, especially children. Conflicts are often between parents, parent-child conflict or sibling rivalries. Life in a dysfunctional family is a turbulence of uncertainty and instability as well as an unsafe space for family members. Instead of expressing their concerns and resolving issues in a positive manner, members in some dysfunctional families normalise their situation and get accustomed to condoning unacceptable behaviour such as abuse, victimisation and conflict, and they sweep issue under the carpet. Conflict is an inevitable part of human relationships; however, dysfunctional families model negative ways of managing conflict to children with the biggest problem being lack of effective communication. In dysfunctional families, communication is replaced with shouting, screaming, arguing and silence.

Healthy functioning families, on the other hand, exhibit harmony, love, care and support for each other; the home is the safest environment where they are able to express themselves, and members have a sense of emotional, mental and physical wellness. In healthy functioning families, conflict, disagreements and differences are resolved in a healthy manner that is beneficial to all concerned.

The negative dynamics that are found in dysfunctional families have adverse effects on the growing personality of children and creates a negative viewpoint on life in general; it inflicts pain and leave emotional wounds that are not reversible. This is because the family has got influence on the development of the child and provides a foundation for the growth of the child such as one's identity, values, norms and morals that are acceptable in society by proving the child with a safe space, love, affection as well as instilling social awareness and confidence [7]. This means the family can influence the growth and development of the child in a positive or negative way depending on the lifestyle, parenting, and the level of functionality of that family. Children are likely to carry what they have observed and learned during their childhood into adulthood.

In dysfunctional families, mostly both or one parent exhibits unharmonious, parenting style and behaves in an unpredictable manner resulting in the home environment being unstable [8]. Children as a result are forever on guard because they never know what to expect and when conflict is going to take place. Some parents are emotionally distant towards children, making it hard to create normal family bonds. The impact on children is low self-esteem and the inability to express their feelings in a healthy way and ultimately childhood trauma. Children as a result experience repeated

trauma and pain from their parents' actions, words and attitudes, while parents are generally in denial that they lead a dysfunctional family [4]. Children grow up with multiple traumas that leave them with permanent emotional and mental scars, sadness and distress. Trauma if not treated may lead to physical and psychological illness [9].

Children from dysfunctional families may experience stigma by their peers for the situation at home. This increases the risk of becoming withdrawn and isolated within the family and around their friends. Growing up in a dysfunctional family indeed exposes children to emotional trauma that can lead to mental illness.

3. Mental illness of children

Child mental health is the ability to grow psychologically, socially, intellectually and spiritually, reaching emotional and developmental milestones without a struggle [10]. Children with mental health challenges are at risk of experiencing a delay in age-appropriate development that can affect their normal functioning and the quality of life. Mental health in children is important for their present and future quality of life because childhood experiences have a profound effect on adulthood.

Mental illness in children can be caused by a variety of issues such as stresses relating to domestic violence, being bullied, losing a loved one to death, separation from friends because of moving homes or schools. It can also be caused by separation from parents because of divorce or parents who work long hours away from home as well as child abuse and suffering from a long illness. Mental illness can also be hereditary meaning there is a likelihood that parents can pass the illness to their children. Some of the symptoms in children are, but not limited to, persistent unhappiness and sadness, emotional outbursts and extreme mood swings, difficulties in academic achievement, loss of appetite or overeating, difficulty falling asleep and fear and sudden loss of interest in previously loved activities such as sport [11].

People exist within the family environment from childhood to adulthood meaning the family plays an essential role in the physical and mental well-being of its members especially during the formative development of children. Children need care that promotes resilience, ability to thrive, modelling appropriate behaviour and coping resources. It is, however, difficult to achieve this when children experience inadequate parental care [12]. Parents can minimise the risk of child mental illness by improving the conditions of living at home, the environment in which the child functions and general childhood relationships and experiences.

The family, specifically parents, have got the responsibility to raise their children in the manner that encourages positive emotional health and overall mental health and minimises the risk and exposure to anxiety, depression, fear and helplessness both at home and outside the home environment by providing love and positive affirmations. While some families try to raise children by ensuring healthy development towards a bright future, some instill and model unhealthy and unhelpful practices that will negatively impact the child's life permanently; an example of this is the high percentage of children who are born with foetal alcohol spectrum disorder (FASD). FASD happens when a pregnant woman consumes alcohol, and the baby is exposed to the harsh impact of alcohol before birth. This condition manifests itself in physical learning and behavioural challenges later when the child is born. According to Tomlinson et al. [13], South Africa has got the highest rate of FASD in the world. Children with FASD are at risk of developing mental illness. FASD unfortunately creates a circle that requires resilience and courage to break.

Modelling negative behaviour to children results in children adopting unhealthy life habits. This can be seen in the prevalence of the adolescent who experience with alcohol in South Africa's province of Western Cape [13]. Such children are affected by the behaviour of their parents, the same parents who are supposed to protect them. This is an indication of unstable and unhealthy parenting practices that may ultimately lead to mental health problems in children.

Mentally healthy children, on the other hand, have a positive outlook on life, and they can function optimally emotionally, socially and academically.

4. The impact of divorce on children

Divorce is prevalent in today society across the world. According to the United Nations Organisation [14], 4.08 per 1000 married persons end in divorce worldwide. In 2020, for example, Maldives recorded the highest divorce rate in the world with 2984 divorces out of a population of 540,544, which translates to 5.52 divorce rate per 1000 married persons. In South Africa alone, 23,710 divorces out of the 129,597 marriages were recorded in 2019, according to Statistics South Africa [15]. Divorce, like other environmental factors that affect families, has a dire effect on children, and it undermines the parent-child relationship because of the decline in the quality of relationships, especially with the parent who does not bear custody. Children from divorced families often experience a range of emotions and challenges, including feelings of loss, confusion and insecurity. They lose the family structure that they are accustomed to, and they have to adjust to living in two separate homes and spending time away from one parent at a time.

Divorce creates emotional distance between the child and the parent who does not live with the child on a full-time basis especially in instances where divorce is preceded by conflict, tension and domestic violence between parents [16]. Protracted divorce processes that are characterised by conflict also create emotional distance between children and parents. According to Fagan and Churchill [17] domestic violence weakens and undermines the parent-child relationship. Children of divorced parents may also feel caught in between because of feelings of conflicting loyalty as though they have to choose between their parents. The distance between parents and children causes emotional strain and irreversible harm, which, if not treated, can result in long-term mental health problems. Children of divorced parents are likely to present with weakened health, psychological trauma and behavioural problems because of insufficient emotional support, affection, care and love from both parents. Children as a result struggle to trust and rely on their parents as they develop a sense of fear for the environment around them. Lack of trust hampers family relations.

On the other hand, parents who bear custody of children are faced with difficulties relating to raising children on their own. Juggling work and single parenting may result in lack of sufficient supervision of children. Single parenting because of divorce makes stress inherent as the parent tries to raise children alone. It reduces household income and makes it difficult for the one parent to maintain the standard of living that the children are accustomed to as well as ensuring the maintenance of the home. These challenges can translate into exposure to risk behaviour for children such as embarking on the use of drugs, criminal behaviour and ultimately falling behind academically. Children in broken families may not receive enough encouragement, support and stimulation, and this can affect their ability to focus on school. Active parental involvement of both parents in the child's life is important to prevent the

overload on one parent. Wajim and Shimfe [18] opined that children from divorced families have an increased likelihood of presenting with anti-social behaviour because of the lack of presence of both parents to bring the child up in the norms and values of society, a task that is the responsibility of both parents, playing complimentary roles in their children's lives. Behere et al. [19] elucidate that divorce is a risk factor for mental health problems especially for children.

Divorce paves a way for negative perceptions against marriage and stable relationships. According to Fagan and Churchill [17], boy children from divorced families, for example, are likely to engage in countless and short-term sexual relationships with multiple partners, and they also have a high turnover of failed intimate relationships compared to adults who were raised in intact families. Fagan and Churchill further revealed that children who experience strained relations between parents prefer to leave home earlier to get married, cohabit or live on their own because of the lack of peace and harmony in their homes, instead of continuing to witness the commotion between their parents.

5. The impact of domestic violence on children

Domestic violence is recognised globally as a public health problem and a violation of human rights by organisations such as the United Nations [20] and the World Health Organisation [21] as well as national and international studies such as [22–26]. It is a destructive act of violence and aggression that causes harm physically and mentally as well as neglect and isolation to the family members who are victims. The intention of violence in the family is mostly to wound, intimidate, manipulate, humiliate and gain power over the victim. It affects people globally across the spectrum of race and class, and it is rooted in gender inequality [27, 28]. While violence in the family affect both men and women, its prevalence is higher in violence against women and children, perpetrated within the family or by intimate partners [29]. According to the World Health Organisation [30], exposure to domestic violence, especially intimate partner violence, increases the risk of mental health problems.

Despite a change in the trend in some countries, violence in the family is often still concealed and not reported because it is regarded as a private matter that does not require external intervention [31]. This assumption that family violence is a private matter normalises violence behind closed doors, leading to many families suffering in silence. Children who are raised in homes with family violence may not report it as they see it as a norm, meaning they may not receive help for the emotional trauma suffered. Children who are exposed to violence and aggression of one form or another may suffer psychologically and emotionally with the likelihood of using violence to resolve conflict with their peers and siblings. This is because of the lack of role models on positive conflict management. As teenagers, they may be victimised and stigmatised if they press criminal charges against their own family members; as a result, they continue to suffer in silence. This may lead to the use of unhealthy methods of coping such as self-harm, substances abuse and suicide. In adulthood, they are inclined to argue with their peers, shouting and using physical violence instead of communicating effectively, and they may exhibit signs of anxiety and depression [32].

Domestic violence is detrimental to the children's mental health as it introduces a stressful home environment with a sense of fear, anger, anxiety, nervousness and depression. The home is supposed to be the safest place for children; however, when violence takes place, children find themselves lost emotionally because they no longer

regard their homes as safe environments. Often violence in the family is directed at adults such as wives and girlfriends; however, the emotional impact goes to children who are helpless. Perpetrators of domestic violence fail to appreciate the impact of their actions on children as they believe that they are physically doing nothing wrong to them. A parent cannot claim to love a child whom they continually subject to witnessing violence against the other parent, mostly mothers. When children see their mothers battered, they feel pain, anger and resentment [28]. This means when violence is perpetrated against one member of the family, the entire family system gets affected, with children being the most affected. Parents who were abused as children may not be able to pay attention to nurturing their children as they may still be battling with their own childhood issues, and this can lead to isolation and neglect of their children.

Children need stable environments with responsive parents who are nurturing and protective to grow and explore without fear of failure or harm. Domestic violence is toxic, and it slowly hurts children emotionally.

6. The impact of working parents on parent: Child relationship

Some parents are not directly involved in conflict, but they are simply too busy chasing careers, business or personal activities such as sport and personal entertainment. Working long hours, taking work home and spending a lot of time on their digital devices lead to physical and emotional absence in the home. As a result, providing inadequate parenting neglects the emotional needs of children and creates emotional distance between themselves and children. Parental employment is an essential tool to obtain economic means and fulfilment of material benefits for the family. Lack of income, on the other hand, can hamper the quality of parenting in terms of providing the day-to-day needs of the child, educational needs and provision of stimulating activities and entertainment.

By spending quality time with children, parents can provide a sense of security and stability, which is essential for their mental health, growth and development. The combination of parental employment and parent-child bond creates the foundation for a healthy functioning environment for the well-being of the child. Lau [33] emphasises that there is a need for parents to maintain a healthy family-work balance to ensure financial, material provision and quality family bonds and relationships.

Working parents might find it difficult to fulfil the parental role and participate in building family bonds. Juggling work and family responsibilities can also result in emotional distress for a parent, which can lead to parents not being able to spend quality time with children, participate in their schoolwork and provide support for their emotional growth concurrently. Lack of parental support may result in compromised parent-child relationship.

Working long hours away from home renders parents vulnerable to stress because of competing demands of work and family roles. Work overload can result in parents feeling overwhelmed, and this can lead to the deterioration in the mental health of parents. It is easy for parents to bring home stress from work that can affect the parent's ability to provide emotional support for children; if not managed, it can undermine the atmosphere in the home and transfer to children [34]. This is because the mental health of a parent has got an impact on the mental health of children. Lengthy hours of work also mean children might have to be placed in alternative care such as aftercare programmes resulting in children spending more time with schoolteachers

and aftercare staff members than with their parents. Bishnoi et al. [34] are of the view that the communication and interaction between parents and children is negatively affected when children spend more time with other people such as caregivers and relatives than with parents. On the other hand, poor-quality day-care services can expose children to physical and emotional harm. A good balance between family and work roles and responsibilities is important for the healthy functioning of the family and development of mental health in children.

7. Conclusions

This chapter provides information about the role families play in the mental health of children and the difficulties faced by children who grow up in dysfunctional families. The family provides an environment for children to grow, develop, observe and learn behavioural traits that will enable them to function in society such as norms, values, morals and socially acceptable behaviour. What children learn and experience have a potential to influence their character and mental health. Children with negative experiences such as divorce, domestic violence, parent-child separation and dysfunctional families are prone to develop mental health challenges.

Divorce exposes children to the difficulties of being raised by a single parent as well as emotional distance. Children from broken families tend to experience trust problems with the perception that marriages and relationships are not safe and intimate partners should not be trusted. Divorce separate children from parents and undermines the parent-child bond, which is important for building and sustaining relationships in the family, as well as social and intimate relationships.

Children are affected by the violence and aggression displayed in families that are riddled by domestic violence. Violence in families is often perpetuated in secret, and as a result, children suffer in silence. Witnessing violence by one parent against the other affects children emotionally and psychologically. When they grow up, such children tend to use violence to resolve conflict and use arguments instead of communication.

The inability of parents to spend quality time with children because of work-related commitments impact the parent-child relationship and cause emotional distance as well. The stress of parents from work if not managed can infiltrate the home environment and lead to tensions in the family. Parental employment is necessary to provide financially for children; however, it is necessary for parents to strike a healthy balance between the two.

The challenges discussed above renders the family system dysfunctional. Dysfunctional families are not able to effectively provide for the emotional, psychological, social and academic needs of their children. Children as such are exposed to neglect, abuse, conflicts and poor communication. This can lead to mental health, behavioural and social challenges in children.

The environment in which children grow up has got an impact on their developing mental health. Families should ensure that factors that contribute to a dysfunctional family are avoided so that children can grow up in nurturing and enabling environments for the development of a healthy mental well-being.

Conflict of interest

The author declares no conflict of interest.

The Impact of Dysfunctional Families on the Mental Health of Children DOI: http://dx.doi.org/10.5772/intechopen.110565

Author details

Lucy Kganyago Mphaphuli The National Prosecuting Authority, Witness Protection Programme, Pretoria, South Africa

*Address all correspondence to: lucy.mphaphuli@yahoo.com

IntechOpen

© 2023 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. (CC) BY

References

- [1] World Health Organisation. Adolescent Mental Health. Switzerland: World Health Organization Library Cataloguing-in Publication; 2021. Available from: Adolescent mental health (who.int)
- [2] Ejaikait VI. Effects of Gender-Based Violence among Students in Masinde Muliro University [Theses]. Kakamega: University of Masinde Muliro; 2014. DOI: 10.13140/RG.2.2.22234.08645
- [3] Holt S, Buckley H, Whelan S. The impact of exposure to domestic violence on children and young people: A review of the literature. Child Abuse and Neglect. 2008;32(8):797-810. DOI: 10.1016/j.chiabu.2008.02.004S
- [4] Word Health Organisation. Violence against Women. Geneva: World Health Organization Press; 2021. Available from: Violence against women (who.int)
- [5] Stephen EN, Udisi L. Single parent families and their impact on children: A study of AMASSOMA community in Bayelsa State. European Journal of Research in Social Sciences. 2016;4(9):1-24. Available from: full-paper-single-parent-families-and-their-impact-on-children-a-study-of-amassoma-community.pdf (idpublications.org)
- [6] Ubaidi BA. Cost of growing up in dysfunctional family. Journal of Family Medicine and Disease Prevention. 2017;3(3):1-6. DOI: 10.23937/2469-5793/1510059
- [7] Heinrich CJ. Parents' employment and children's wellbeing. Future of Children. 2014;**24**(1):121-146. Available from: www.futurechildren.org
- [8] Minullina AF. Psychological trauma of children of dysfunctional families.

- Future Academy. 2018;**45**(1):65-74. DOI: 10.15405/epbs.2018.09.8
- [9] Rodrigues VP, Rodrigues AS, Lira MD, Couto M, Diniz NM. Family relationships in the context of genderbased violence. Text and Context Nursing Journal (Brazil). 2016;25(3):2-9. DOI: 10.1590/0104-07072016002530015
- [10] Edwards D, Burnard P. A systematic review of stress and stress management interventions for mental health nurses. Journal of Advanced Nursing. 2003;42(1):169-200. DOI: 10.1046/j.1365-2648.2003.02600.x
- [11] Zubrick SR, Silburn SR, Burton P, Blair E. Mental health disorders in children and young people: Scope, cause and prevention. Journal of Psychiatry. 2000;34(1):570-578. DOI: 10.1080/J.1140-1614.2000.00703.X
- [12] Breiner H, Ford M, Gadsden VS. Supporting Parents of Children Ages 0-8. Washington DC: National Academics Press; 2016. DOI: 10.17226/21868
- [13] Tomlinson M, Kleinjies S, Lake L. Child and Adolescent Mental Health. South African Child Gauge. Children's Institute, University of Cape Town; 2022. Available from: Child Gauge 2021_110822.pdf (uct.ac.za)
- [14] United Nations Organisation. Demographic and Social Statistics on Marriage and Divorce. New York: UNSD-Demographic and Social Statistics; 2022
- [15] Statistics South Africa. Marriages and Divorce in South Africa. Pretoria: South African Government Printing Works; 2021. Available from: www.statssa.gov.za
- [16] Strohschein L. Parenting, divorce and child mental health: Accounting for

- pre-disruption differences. Journal of Divorce and Remarriage. 2012;53(6):489-502. DOI: 10.1080/10502556.2012.682903
- [17] Fagan PF, Churchill A. The Effects of Divorce on Children. Washington DC: Harni Research; 2022. Available from: Marri.frc.org/effects-divorce-children
- [18] Wajim J, Shimfe HG. Single parenting and its effects on the development of children in Nigeria. The International Journal of Social Sciences and Humanities Invention. 2020;7(3):5891-5902. DOI: 10.18535/ijsshi/v7i04.02
- [19] Behere AP, Basnet P, Campbell P. Effects of family structure on mental health of children: A preliminary study. Indian Journal of Psycho Medicine. 2017;39(4):457-463. DOI: 10.4103/0253-7176.211767
- [20] United Nations Organisation. Handbook for Legislation on Violence Against Women. New York: United Nations Publication; 2009
- [21] World Health Organisation. World Report on Violence and Health. Geneva: World Health Organization Cataloguing-in Publication; 2002. Available from: 9241545615_eng.pdf (who.int)
- [22] Gregory S, Holt S, Barter C, Christofides N, Maremela O, Motjuwadi NM, et al. Public health directives in a pandemic: Paradoxical messages for domestic abuse victims in four countries. International Journal of Environmental Research and Public Health. 2002;19(1):2-15
- [23] Krantz G. Violence against women: Global public health issue. Journal of Epidemic Community Health. 2002;**56**(1):242-243. Available from: v056p00242.pdf (nih.gov)
- [24] Chibber K, Krishnan S. Confronting intimate partner violence: A global

- health care priority. Mt Sinai Journal of Medicine. 2011;78(3):449-457. DOI: 10.1002/msj.20259
- [25] De Wet-Billings N, Godongwana M. Exposure to intimate partner violence and hypertension outcomes among young women in South Africa. International Journal of Hypertension. 2021;**2021**:1-8. Article ID 5519356. DOI: 10.1155/2021/5519356
- [26] McCloskey LA, Boonzair F, Steinbrenner SY, Hunter T. Determinants of intimate partner violence in sub-Saharan Africa: A review of prevention and intervention programs. Partner Abuse. 2016;7(3):277-315. DOI: 10.1891/1946-6560.7.3.277
- [27] Kertesz M, Fogden L, Humphreys C. Domestic Violence and the Impact on Children. London: Routledge is Part of the Taylor & Francis Group Publishers; 2021. DOI: 10.4324/9780429331053
- [28] Khemthong O, Chutiphongdech T. Domestic violence, and its impacts on children: A concise review of past literature. Walailak Journal of Social Science. 2021;14(6):1-12. Available from: https://so06.tci-thaijo.org/index.php/wjss
- [29] Rada C. Violence against women by male partners and against children within the family: Prevalence, associated factors and intergenerational transmission in Romania: A cross-sectional study. BMC Public Health. 2014;14(129):1-15. DOI: 10.1186/1471-2458-14-129
- [30] World Health Organisation. Preventing Intimate Partner Violence Improves Mental Health. Geneva: World Health Organization Cataloguing-in Publication; 2022. Available from: Preventing intimate partner violence improves mental health (who.int)
- [31] Chuemchit M, Chernkwanma S, Rugkua R, Daengthern L, Abdullakaim P,

Wieringa S. Prevalence of intimate partner violence in Thailand. Journal of Family Violence. 2018;**33**(5):315-323. DOI: 10.1007/s10896-018-9960-9

- [32] Lloyd M. Domestic violence and education: Examining the impact of domestic violence on young children, children and young people and the potential role of schools. Frontiers in Psychology. 2018;9(11):1-11. DOI: 10.3389/fpsyg.2018.02094
- [33] Lau YK. The impact of fathers' work and family conflicts on children's self esteem: The Hong Kong case. Springer. 2010;95(1):363-376. DOI: 10.1007/s11205-009-9535-5
- [34] Bishnoi S, Malik P, Yadav P. A review of effects of working mothers on Children's development. Akinik Publications. 2020;4(1):41-55. DOI: 10.22271/ed.book.960

Chapter 3

Parenting and Work-Family Balance in the Twenty-First Century

Ethelbert P. Dapiton, Enrique G. Baking and Ranie B. Canlas

Abstract

Parenting in the twenty-first century compel among working parents a great deal of effort and balancing act between having a family and at the same time maintaining a good career. The challenges among working parents are on the balancing act of taking care and attending to family needs while becoming successful in chosen career. In the modern-day society where couples have to strive to make both ends met for the needs of the family, the issue and concern for balancing act is really perplexing. In this fast-paced twenty-first century world, such concern for balancing act has just been relegated to the confines of family resolution without the greater society having so much concern of it. However, it is a topic worthy of concern and study as it pervades not only within the realm of family life but as well as on the aspects of productivity and performance of working professional parents. In this paper, seven general themes with underlying discussions were presented. The themes are not exhaustive of the dynamics between parenting and work-family balance. The bottom line is to provide insights, reflection points and points to ponder about the realm of parenting and work-family balance in the twenty-first century.

Keywords: parenting in the twenty-first century, work-family balance, parenting challenges, parenting struggles, parenting dilemma

1. Introduction

The advent of twenty-first century has brought dramatic changes to social arrangements in general. In particular, parenting and work-family balance has its own series of vignettes worthy of attention since the dawning of the twenty-first century. To start with, parenting in the context of this paper refers to the attributes of activities about rearing and educating a child by parents [1]. On the other hand, work-family balance can be defined as the equal treatment of work and family responsibilities among parents [2, 3]. The dynamics between parenting and work-family balance has created several social contracts and constructs that are already distant in identity compared to that of the immediate prior centuries. Changes in social contracts supported by institutional polices that ranges from expansion and enhancement of child care facilities so that working parents can have the opportunity to attend to their work regularly down to the provisions of reduced work hours for working mothers and

23 IntechOpen

flexible work schedules are already widely practiced among organizations [4, 5]. The social constructs of work-family dynamics have also seen several changes such as the role of mothers being the caregivers and fathers being the breadwinners. These social portrayal of roles in the family has slowly been eroded in the twenty-first century landscape [6]. This change in the societal role of men and women in the context of family responsibilities can be attributed to the ever-growing entry of women in the labor force in the beginning of the twenty-first century [7].

Perhaps the modernity of the twenty-first century is the main reason that has created significant impact to both the aspects of parenting and work-family balance. The evolving nature of society is truly a catalyst of change to the features of parenting and work-family balance that modern humans have seen and observed. At the macro-perspective, mega-factors such as political, social, economic and technological can be considered as the primary sources of significant changes to parenting and work-family balance. Looking into at the micro level, individual attributes such as job satisfaction, personal preferences, motivation, career choice and personal discernment among others are the primary turnaround of the subtleties of parenting and work-family balance in the twenty-first century.

Salient features of modern-day parenting can be seen in shared responsibilities along with the revival of egalitarianism. The delineation of gender roles and its corresponding stereotyped identities between parents are slowly eroding. This can be attributed to the changing consciousness of the current society which is more liberal and fluid. Another lever of change can be attributed to the financial and economic need and stability of the family. Thus, modern-day parenting does not account who will stay at home and who will work as long that opportunities offer economic stability for the family.

Globalization and interconnectivity hastened by the use of handheld technology gadgets has also empowered parents and families in general to learn from other parents and other individuals that share their experiences. Alongside, a plethora of resources about the topic can be readily accessible from the Internet. A lot of lessons can be learned by parents by simply browsing the Internet from the best practices about parenting and work-family balance in this modern-day society. Digital technologies and the Internet have empowered parents to learn from other people across time, space and national borders, making them more resilient and efficient in their parenting roles and managing their work-family balance [8].

The central thesis of this paper revolves around several pressing issues that confronts and pervades parenting and work-family balance in the context of twenty-first century. Anchored on the everyday backdrop of parenting and work-family realities in twenty-first century, this paper explores and explicate recurrent themes that are worthy to be reckoned with from the vantage points of parents, academic researchers and the greater society.

2. Methodology

This research utilized the scope literature review as the methodology to fill-in the discussions of predefined themes relating to parenting and work-family balance in the twenty-first century. The predefined themes are presented in succeeding sections. Google Scholar was the primary platform for the search along with other databases such as ProQuest and Ebscohost. The inclusion was narrowed down subject to the relevance of the papers to this research at the discretion of the authors.

Other search parameters to expand the periphery of the topic was also made. The main term used for the search is "parenting in the twenty-first century" with Boolean operators "AND" and "OR" to connect with other string of words such as "parenting challenges", "parenting struggles", "work-family balance", "parenting and work", and "parenting and career trade-off" among others. Both peer reviewed articles and gray literatures were included. In order for the gray literatures to be included, it has to be either a conference paper, academic notes, or literatures coming from credible web sources and with reputable origin. Other criteria for inclusions are:

- Papers must be written in English.
- Papers must have at least some degree of contextual discussions related to parenting in the twenty-first century.
- Papers must have at least some degree of contextual discussions related to parenting and work-family balance.

3. Performing the balancing act

It is indeed a challenging task for parents to work and at the same time attend to family needs. The struggles of everyday job and taking care of the needs of the family requires a lot of patience, courage and perseverance for parents. This reality is more vividly depicted for starter families wherein young parents has to juggle the balancing act of being an effective and efficient worker and at the same time providing perpetual attention to the newly formed family. Regardless of spatial and geographical context, there is a universal truth that the juggling activities among parents to strike a balance between work and family has made a dent on their emotional, psychological, physical and economic wellbeing [9].

The learning curve of performing the balancing act increases as the family becomes psychologically and emotionally mature and by that time, the children are already grown up with little attention being required to look after their personal needs.

Performing the act of good parenting and maintaining work-family balance is an overarching goal of almost all parents. Good parenting is referred in this paper as the awareness of parents and their ability to provide the needs of their child [10]. Good parenting is also allied to the notion of positive parenting which purports the attributes of parental warmth and responsiveness [11].

At the outset of the balancing act, a variety of negative spillover effects can come into fruition such as stress, family dissatisfaction and poor workplace performance just to name a few [12]. It is of crucial importance that parents has to spend quality time with their children but at the same time exert substantial quantity of efforts in their jobs [13].

The early years of parenting and family life is overwhelmed with mixed emotions and struggles to balance the variety of aspects between family and work. The balancing act require parents to keep up with the rapid growth and development of their children during formative years [14]. On the other hand, the desire to become successful or excel in one's career or job is at the other side of the balancing spectrum. To some extent, trade-offs cannot be avoided in performing the balancing act among parents. There are circumstances that some parents have to take side and chose

priorities between family and career. Empirical evidence shows that women are more likely to trade career opportunities for their family and consequently devote more of their time to child rearing and household activities [15].

The nature of the balancing act between parenting and work is to some degree innate in the very environment where parents work. Organizations and institutions that have high-performance work practices and with some degree of flexibility offer better opportunities for their staff to exercise productivity at work and attending to family obligations [16]. Hence, there is a strong attribution towards the very nature of workplaces or working environments as significant contributory factor for proper balancing act.

Institutional and organizational policies that foster harmonious work and family life balance is an essential catalyst to attain the desired level of satisfaction and success between work and parenting [17]. Situational and work circumstances that are not complimentary in attaining harmonious work and family balance will render dysfunctionality among working parents and thus resulting to negative family outcomes [18]. The negative outcomes brought by dysfunctionality in the family has many root causes. Salient among them are parental stress and quality of relationship due to caregiving burden and role conflict [19, 20].

It can be deduced therefore that the performing of balancing act is not only burdened upon the shoulders of parents. The very organizations where these parents are working have to do their share of helping these parents perform the balancing act. It is the moral duty, obligations and social responsibility of these organizations to help parents achieve the proper balance of work and family [21].

4. Challenges and constraints

It is indicative that young and starting families are expected to face more challenges and constraints as compared to those that are nearing the stage of empty nest. Studies have found out that family life cycles are factors to reckon with when looking into the challenges and constraints of parents towards rearing their children [22–24]. The growing period for a family is beset with challenges and constraints due to the demands of raising a child which solicits time and effort and on the other hand compelling parents to work for economic sustainability and survival [22]. Thus, from the point of family founding up to child launching stage, it is expected that multiple challenges and constraints can be encountered.

Working parents are at dilemma on finding ways on how to balance their parenting responsibilities and being an efficient employee [25]. The reckoning of twenty-first century parenting issues and concerns has taken a visible step recently owing to the fact that more legislative and legal redress has been crafted to address the clamor from citizens to harmonize work and family activities [26].

One of the most prominent and distinct feature relative to challenges and constraints of parenting in the twenty-first century is the high level of perceptual stress among parents [27]. The post-modern work-family ecology has brought tremendous stress among parents in the aspects of financial, social and emotional amidst the demands of good parenting imperative. The pressing demands of good parenthood along with the need to provide the family with stable stream of economic stability has provided parents with so much pressure and stress [28].

Regardless of the setting, there exist some degree of disjuncture between parenting and work-family balance among parents in the twenty-first century [29].

Both parents actually have their respective stress and pressures because of their social roles to portray in the family. Mothers are destined to stay at home in exchange for a career to look after the needs of a growing child. While fathers are socially portrayed and expected to become good providers for the family. This portrayal and expectation of roles among parent were socially constructed by previous era and has been carried over to the present century.

There is a detrimental effect to such social construction among parents with regards to the portrayal of roles. Mothers can experience maternal depression while fathers can experience work-family stress phenomenon due to wage premium expectation. Maternal depression can have a significant effect in the development of a child [30–32]. On the other hand, fathers will tend to be confronted with work-family stress that can be associated with breadwinning demand. This is more pronounced among low-tiered income fathers that have been caught in a quandary between parenting roles and fulfilling the economic needs of the family through stable working environment [33].

In general, all parents can experience parenting stress in one way or another. However, too much of parenting stress can reduce parents psychological and cognitive ability to become optimal parents [34]. There is a need among parents to maintain their proper emotional, cognitive and psychological functioning even when confronted with challenges and difficulties brought by parenting.

5. The hustle Culture and work-family balance

Almost all parents that are wanting to provide their families with substantial economic stability and resources are consumed by the idea and obsession of working harder and longer. The capitalist society is partly the source of the hustle culture and instilling in the mindset of individuals that the idea of rise and grind is the catchall phrase for success. This can be a false panacea that the modern-day society has venerated.

Working parents are obviously sold out to this idea at the expense of work-family balance. Working hard is not bad after all. It fosters productivity and organizational commitment. What is detrimental to work-family balance is going beyond the limits of regular work and having little or no time for family responsibilities. Parents has to be cautious in oversubscribing to hustle culture as there are negative outcomes such as physical and mental health effects in the long run.

Twenty-first century parents have to be cognizant about the negative effects of the hustle culture to their family relations. Hustle culture can take some toll on employees' physical and mental health considering the negative effects of working long works [35]. Hustle culture is a quick path to burnout and can eventually affect the work-family balance structure in the long run.

The only way to counter the cult of the hustle culture is for institutions to implement work-family balance friendly policies. This will enable working parents to cope with their work demands while maintaining their active family roles [36]. Institutions have a key role and responsibility by implementing policies that are work-family friendly so that working individuals can achieve significant level of work-family balance [37].

The hustle culture can definitely take away precious moments that supposedly allotted for spending quality time with family and loved ones [38]. In the twenty-first century parenting context, parents have to work smarter more than working

harder. The rise and grind notion to success advocated by the hustle culture is more of a fallacy than a fact [39, 40]. Moreover, what does working parents can profit from hustle culture at the expense of strained family relations. Excessive working hours has negative effects on workers' mental and physical well-being as well as losing proper balance between work and their families [41].

6. Work: family conflict and family vulnerabilities

Work-family conflict among families are common scenes among working parents [42]. Moreover, it has been found out that parents that are exposed to longer hours of work are prone to experience work-family conflict compared to parents with flexible work schedules [43]. The prevalence of work-family conflict poses a significant risk and danger to families. This may lead to adverse outcomes such as family dissatisfaction and consequently family dissolution. Work-family conflict can make parents and families vulnerable to negative outcomes that will result to strained family relations. Stressful experiences such as fatigue and psychological imbalance are visible manifestations of negative outcomes brought about by work-family conflict [44]. Further, work performance is also reported among scholars to be affected by work-family conflict [45–47]. The balancing of work to family demands can lead to work-interference-with-family making both parents vulnerable to unhealthy family interactions [48].

Moreover, those families that are in lower strata of the economy are more vulnerable to work–family conflict. Parents in this lower stratum of society lack the ability and mechanisms to compensate and mitigate the adverse effect of work–family conflict [49]. Low-income households still experience inequalities in the context of solving their work–family conflict since majority of their workplaces does not foster policies to mitigate such concern [50]. Scholars are already recognizing this phenomenon as a structural issue more than just a personal shortcoming among parents [51]. Moreover, women are often the most vulnerable between the two parents in work–family conflict [52]. The prevalence on the differential bias of gender role and responsibility in the family is still present although subtle in modern times [53, 54]. The result will be most likely that women are to take the burden of child rearing and household responsibilities [55, 56]. Both parents eventually will suffer the effects of work–family conflict and spousal disputes will significantly increase over time with collateral damages such as children's emotional well-being [57].

7. The motherhood penalty

In this theme- 'the motherhood penalty' along with the succeeding theme which is the 'fatherhood premium' are presented consecutively for the purpose of capturing the dynamics of parenting from various standpoint. The purpose of presenting and discussing these two consecutive themes is to provide readers a comprehensive macro perspective about the context of parenting in the twenty-first century. The 'motherhood penalty' and the 'fatherhood premium' can be considered as paradoxes of parenting dynamics and are worthy to be given attention in the discussion of parenting issues. The importance of laying down these two important themes for discussions are not contradictory but rather a point of reckoning wherein in the grand scheme of things, in the realm of parenting such paradoxes exists.

Empirical evidences have showed that women are most likely to experience draw-backs from being a parent in relation to career progression than their male counterparts [58–61]. Gendered norm identities and responsibilities are inherent part of parenting aspects. The aspect of motherhood penalty is one of the most recognizable themes in the sphere of parenting and work-family balance. Income and career inequalities among working mothers are ascribed as the motherhood penalty [62]. Women are more likely to bear the sacrifice of striking the balance between work or career and family matters [63]. In the process, this results to gendered gaps in work opportunities and growth among working mothers which can significantly affect their careers [60].

Women usually experienced drawbacks in career progression the moment they became mothers [64]. This is due to the fact that women would have additional family responsibility during child rearing process that gives them additional work-family balance pressure. There is a common perspective among work institutions that women with children are more likely to reduce their commitment at work due to caregiving responsibilities at home [65]. This social perspective is more likely to result in gender inequality in the workplace as well as a prejudice to working mothers' opportunity for career growth and advancement [66]. This normative discrimination among mothers is associated with the view that they are less committed to their jobs since they have children to look after [64]. Consequently, this amplifies employers' perceptions that mothers have conflicting interests between work productivity and family responsibilities [67].

8. The fatherhood premium hypothesis

The social perception that fathers as breadwinners and mothers as caregivers pervades the realm of parenting discussions. Such would often lead to the delineation of gendered roles and gendered pay gaps in the process of family formation [68]. Fatherhood premiums are associated with the theoretical underpinning wherein fathers are more likely to increase their productivity granting that marriage, family and child are reinforcing factors for such premium gains [69]. The fatherhood premium hypothesis can be considered as an incentive among fathers to strive hard to provide for their family's needs [70]. Occupational characteristics are the usual underlying elements of the fatherhood premium but is just one parameter among the myriad of explanatory mechanisms that can affect wage differentiation [71].

There are however contending arguments about the fatherhood premium hypothesis. Some empirical findings do not support the desired outcome of the fatherhood premium hypothesis which leads to the inference that underlying mechanisms for wage and productivity premiums among fathers are not universally true [72]. If that's the case, the fatherhood premium can be considered as milieu specific contextual results that can change over a period of time in different locales. The fatherhood premium therefore can remain as a working hypothesis subject to further verification. Other parameters aside from occupational characteristics should be explored such as work efforts as a moderating factor to assert the universal existence and veracity of the fatherhood premium hypothesis [73].

9. Importance of egalitarianism to work-family balance

In the twenty-first century parenting scenario, it does not matter anymore whether it will be the mother or the father that contributes more to the family and

household chores. Raising a family is not about counting the contribution from each parent. It is about the convergence and concerted efforts from both parents that makes parenting and work-family balance becomes feasible and attainable [74].

Stereotyping of gender identity in the realm of parenting in the twenty-first century can be a thing of the past. Such stereotype perspective of gender identify parenting has an almost cult-following popularity in previous centuries. The context of parenting in the twenty-first century is more of an egalitarian nature rather than delineation of gender roles in parenting [75]. Basically, it has to be necessary to become egalitarian so as to promote the idea of equality among parents.

Although the notion of egalitarianism in parenting is not a new concept, there is a need to revive the spirit in order to propagate the essence of equality among parents. This can be called neo-egalitarian parenting movement. The bottom line that can be achieved with such notion is the attainment of equality of gender roles, sharing of accountabilities and responsibilities, and to unburden the other parent especially women from the despotic idea of masculinity of the other parent along with the stereotype roles and identities in the family.

The neo-egalitarianism can also be related to neoliberal concept of parenting and work-family management of the twenty-first century [76]. The social constructs of parenting and work-family balance in the twenty-first century is projected towards the functionality of the family along with its coherence in the general society. This can be considered as well as a post-modernism movement of parenting and work-family balance with emphasis on child-focused care together with institutional and social support mechanisms to achieve the desired outcomes of parenting ideals. The practice of egalitarian parenting can be diverse from country to country but the general trajectory points to the same direction [77]. Egalitarianism will prevail in the context of parenting in the twenty-first century.

10. Conclusions

The themes presented in prior sections are not exhaustive of the realities and dynamics of parenting and work-family balance in the twenty-first century. However, to some degree, it has elucidated several pressing issues and topics that needs to be given attention in the context of parenting and work-family balance. Parents, academic researchers and even ordinary people are well aware that those aforementioned topics discussed are the central issues of parenting and work-family balance.

Although so much has been said about those topics, institutional and social support are still necessary to improve the working conditions of parents in the modern work environment. Legislative and legal actions are also vital to mitigate the stress and anxiety of parents due to the juggling demands between work and family responsibilities. The practice of egalitarianism and shared family responsibility roles at the personal and family level can greatly alleviate the burden of the family in terms of work-family balance.

Stereotyped gender roles are not necessary to be practiced in the twenty-first century parenting. It is a dysfunctional social idea that does not serve any merit in the twenty-first century parenting practice.

The academia which is the primary source of knowledge disseminated towards the greater society should constantly encourage researchers to embark on the topics of parenting and work-family balance. This is to further develop and expand the body of knowledge about the topic. There is also a need that this topic has to be revitalized

Parenting and Work-Family Balance in the Twenty-First Century DOI: http://dx.doi.org/10.5772/intechopen.110689

among researchers and scholars. Further, researchers that will venture into this kind of topics for empirical investigations should incorporate or consider other parameters which are vital to the building and development on this body of knowledge.

In succeeding opportunities for publication, the authors wish to explore other dominant and salient themes relevant to parenting and work-family balance in the twenty-first century that can still be developed and expounded.

Acknowledgements

The authors are grateful to IntechOpen for the opportunity given to them to share their knowledge about the subject.

Author details

Ethelbert P. Dapiton*, Enrique G. Baking and Ranie B. Canlas Don Honorio Ventura State University, Philippines

*Address all correspondence to: ethelbertdapiton@yahoo.com

IntechOpen

© 2023 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. CCO BY

References

- [1] Virasiri S, Tunibhand J, Chaiyawat W. Parenting: What are the critical attributes? Journal of the Medical Association of Thailand. 2011;**94**(9):1109-1116. Available from: http://www.mat.or.th/journal
- [2] Grzywacz JG, Butler AB. In: Fink G, editor. Work-Family Balance. Encyclopedia of Stress. 2nd ed. Amsterdam: Elsevier; 2007. pp. 868-871. DOI: 10.1016/B978-012373947-6.00694-2
- [3] Grzywacz JG, Carlson DS. Conceptualizing work-family balance: Implications for practice and research. Advances in Developing Human Resources. 2007;9(4):455-471. DOI: 10.1177/1523422307305487
- [4] Fagnani J. Work-family Life Balance: Future Trends and Challenges. Halshs: OCDE Publishing; 2011. pp. 119-188
- [5] Thomson P. New ways of working in the company of the future. In: Reinventing the Company in the Digital Age. Madrid: OpenMind BBVA; 2015. pp. 248-262. Available from: https://www.bbvaopenmind.com/en/articles/new-ways-of-working-in-the-company-of-the-future/
- [6] Bianchi SM, Milkie MA. Work and family research in the first decade of the 21st century. Journal of Marriage and the Family. 2010;72:705-725. DOI: 10.1111/j. 1741-3727.2010.00726.x
- [7] OECD. Families are changing. In: Doing Better for Families. Paris: OECD; 2011. pp. 17-50. Available from: https:// www.oecd.org/els/soc/47701118.pdf
- [8] Walker SK. Technology Use and Families: Implications for Work-family Balance and Parenting Education.

- New York: United Nations Department of economic and social affairs (UNDESA) division for inclusive social development; 2021. Available from: https://www.un.org/development/desa/family/wp-content/uploads/sites/23/2021/05/Technology-Families-Background.pdf
- [9] O'Brien M. Work-Family Balance Policies. New York: United Nations Department of Economic and Social Affairs (UNDESA); 2012
- [10] Eve PM, Byrne MK, Gagliardi CR. What is good parenting? The perspectives of different professionals. Family Court Review: An Interdisciplinary Journal. 2014;52(1):144-127. Available from: https://ro.uow.edu.au/sspapers/584
- [11] Schofield TJ, Conger RD, Neppl TK. Positive parenting, beliefs about parental efficacy, and active coping: Three sources of intergenerational resilience. Journal of Family Psychology. 2014;**28**(6):973-978. DOI: 10.1037/fam0000024
- [12] Chen Q, Chen M, Lo CKM, Chan KL, Ip P. Stress in balancing work and family among working parents in Hong Kong. International Journal of Environmental Research and Public Health. 2022;**19**(9):5589. DOI: 10.3390/ ijerph19095589
- [13] Milkie MA, Kendig SM, Nomaguchi KM, Denny KE. Time with children, Children's well-being, and work-family balance among employed parents. Journal of Marriage and the Family. 2010;**2010**:1329-1343. DOI: 10.1111/j.1741-3737.2010.00768.x
- [14] Early Child Learning & Knowledge Center [ECLKC). Supporting Parenting: A Balancing Act. 2021. Available from: https://www.eclks.ohs.acf.hhs.gov/

- family-engagement/staff-parent-relationships-honor-support-parenting/supporting-parenting-balancing-act. [Accessed: January 3, 2023]
- [15] Parker K. Women More than Men Adjust Their Careers for Family Life. Washington, D.C.: Pew Research Center. 2015. Available from: https://www.pewresearch.org/fact-tank/2015/10/01/women-more-than-men-adjust-their-careers-for-family-life/. [Accessed: January 3, 2023]
- [16] Berg P, Kalleberg AL, Appelbaum E. Balancing work and family: The role of high-commitment environments. Industrial Relations. 2003;**42**(2):168-188
- [17] Danbolt IL. The Ultimate Balancing Act: Work and Family in the Nordic Region. Copenhagen, Denmark: Nordic Council of Ministers; 2016
- [18] Akuamoah-Boateng C. Balancing work, family and personal life: Perspectives of female staff at the College of Distance Education, University of Cape Coast, Ghana. International Journal of Educational Administration and Policy Studies. 2020;12(1):43-51. DOI: 10.5897/IJEAPS2019.0643
- [19] Thomas PA, Liu H, Umberson D. Family relationships and well-being. Innovation in Aging. 2017;1(3):igx025. DOI: 10.1093/geroni/igx025
- [20] Zagefka H, Jones J, Caglar A, Girish R, Matos C. Family roles, family dysfunction, and depressive symptoms. The Family Journal. 2020;**29**(3):346-353. DOI: 10.1177/1066480720973418
- [21] O'Çonnor MA. Corporate Social Responsibility for Work/Family Balance. St. John's Law Review. 2005;**79**:11. Available from: https://scholarship.law. stjohns.edu/lawreview

- [22] Yan S, Zhou Y. Research of workfamily balance based on family life cycle. Open Journal of Social Sciences. 2016;4:218-224. DOI: 10.4236/jss.2016.411018
- [23] Amstad FT, Meier LL, Fasel U, Elfering A, Semmer NK. A meta-analysis of work-family conflict and various outcomes with a special emphasis on cross-domain versus matching domain relations. Journal of Occupational Health Psychology. 2011;**16**:151-169. DOI: 10.1037/a0022170
- [24] Loscalzo Y, Raffagnino R, Gonnelli C, Giannini M. Work-family conflict scale: Psychometric of the Italian version. SAGE Open. 2019;**9**(3):2158. DOI: 10.1177/ 2158244019861495
- [25] Omran SKA. Work-family balance dilemma among employed parents: An empirical study. International Journal of Business and Economic Development. 2016;4(1):31-46
- [26] Hoffman KA. The work/family balance: New York's struggle to harmonize domestic and employment spheres. Cardozo JL & Gender. 2009;**16**:93
- [27] Buchanan CM, Glatz T. Introduction to special issue "parenting in the 21st century". Social Science. 2022;**11**(10):484. DOI: 10.3390/socsci11100484
- [28] Walper S, Kreyenfield M. The intensification of parenting in Germany: The role of socioeconomic background and family form. Social Science. 2022;**11**(3):134. DOI: 10.3390/socsci11030134
- [29] Walker J. Family life in the 21st century: The implications for parenting policy in the UK. Journal of Children's

- Services. 2008;**3**(4):17-29. DOI: 10.1108/17466660200800023
- [30] Stavrinides P, Nikiforou M. Parenting: Challenges, practices and cultural influences. In: Barberies P, Petrakis S, editors. Parenting. New York: Nova Science Publishers, Inc.; 2013. pp. 59-82
- [31] Connell AM, Goodman SH. The association between psychopathology in fathers versus mothers and children's internalizing and externalizing behavior problems: A meta-analysis. Psychological Bulletin. 2002;**128**(5):746-773. DOI: 10.1037/0033-2909.128.5.746
- [32] Goodman SH. Depression in mothers. Annual Review of Clinical Psychology. 2007;3:107-135. DOI: 10.1146/annurev.clinpsy.3.022806.091401
- [33] Nomaguchi K, Johnson W. Parenting stress among low-income and working-class fathers: The role of employment. Journal of Family Issues. 2014;37(11):1535-1557. DOI: 10.1177/0192513X14560642
- [34] Whiteside-Mansell L, Ayoub C, McKelvey L, Faldowski RA, Hart A, Shears J. Parenting stress of low-income parents of toddlers and Preschoolers: Psychometric properties of a short form of the parenting stress index. Science and Practice. 2007;7(1):26-56. DOI: 10.1080/15295190709336775
- [35] Li Z, Dai J, Wu N, Jia Y, Gao J, Fu H. Effect of long hours on depression and mental well-being among employees in Shanghai: The role of having leisure hobbies. International Journal of Environmental Research and Public Health. 2019;16(24):4980. DOI: 10.3390/ijerph16244980
- [36] Zhao X, Qu H, Liu J. An investigation into the relationship between hospitality

- employees' work-family conflicts and their leisure intentions. Cornell Hospitality Quarterly. 2014;55(4):408-421. DOI: 10.1177/1938965513517171
- [37] Sim AKS. Work-family enrichment and job-family satisfaction among hotel employees. World Applied Sciences Journal. 2013;**22**(12):1775-1781. DOI: 10.5829/idosi.wasj.2013.22.12.393
- [38] Ruggeri C. Research shows hustle Culture does more harm than good. Leaders. 2022. Available from: https://leaders.com/articles/company-culture/hustle-culture/
- [39] Budin C. 'Rise and grind'mentality doesn't determine success. The Daily Illini. 2020. Available from: https://dailyillini.com/special-sections/2020/01/27/rise-and-grind-mentality-doesn't-determine-success/
- [40] Mikhail A. Hustle culture is a dangerous myth, burnout expert says. Here are 6 ways to beat it. Yahoo Finance. 2023. Available from: https://finance.yahoo.com
- [41] Hustle Culture. Definition, impact and how to overcome it. Sampoerna University. 2022. Available from: https://www.sampoernauniversity.ac.id/hustle-culture-definition-impact-and-how-to-overcome-it/
- [42] Cooklin AR, Westrupp EM, Strazdins L, Giallo R, Martin A, Nicholson JM. Fathers at work: Workfamily conflict, work-family enrichment and parenting in an Australian cohort. Journal of Family Issues. 2014;37(11):1-25. DOI: 10.1177/0192513X14553054
- [43] Moreira H, Fonseca A, Caiado B, Canavarro MC. Work-family conflict and mindful parenting: The mediating role of parental psychopathology symptoms and parenting stress in a sample of

- Portuguese employed parents. Frontiers in Psychology. 2019;**10**:635. DOI: 10.3389/fpsyg.2019.00635
- [44] Zakaria N, Ismail Z. The consequences of work-family conflict and the importance of social Suports to reduce work-family conflict among employees. Malaysian Journal of Social Sciences and Humanities. 2017;2(2):24-39
- [45] Mihelic KK, Tekavcic M. Work-Family Conflict: A Review of Antecedents and Outcomes. International Journal of Management & Information Systems. 2014;2014:18
- [46] Eby LT, Casper WJ, Lockwood A, Bordeaux C, Brinley A. Work and family research in IO/OB: Content analysis and review of the literature (1980-2002). Journal of Vocational Behavior. 2005;**66**:124-197. DOI: 10.1016/j. jvb.2003.11.003
- [47] van Steenbergen EF, Ellemers N. Journal of Organizational Behavior. 2009;**30**(5):617-642. DOI: 10.1002/job. v30:510.1002/job.569
- [48] Foucreault A, Menard J, Houlfort N, Trepanier SG, Lavigne GL. How workfamily guilt, involvement with children and spouse's support influence parents' life satisfaction in a context of workfamily conflict. Journal of Child and Family Studies. 2022. DOI: 10.1007/s10826-022-02443-6
- [49] Henly JR, Lambert S. Nonstandard work and child-care needs of low income parents. In: Bianchi SM, Casper LM, King BR, editors. Work, Family, Health, and Well-Being. Hillsdale, New Jersey: Lawrence Erlbaum Associates Publishers; 2005. pp. 473-492
- [50] Hiciano Ramos MG. Work-Family Conflict in Low-Income Households.

- Published Master's Thesis, The City University of New York. CUNY Academic Works. 2020. https://academicworks. cuny.edu/gc_etds/3757
- [51] Kuskoff E, Parsell C, Plage S, Ablaza C, Perales F. Willing but unable: How resources help low-income mothers care for their children and minimise child protection interventions. The British Journal of Social Work. 2022;52(7):3982-3998. DOI: 10.1093/bjsw/bcac027
- [52] Luhr S. Signaling parenthood: Managing the motherhood penalty and fatherhood premium in the US service sector. Gender and Society. April;34(2):259-283. DOI: 10.1177/0891243220905814
- [53] Cerrato J, Cifre E. Gender inequality in household chores and work-family conflict. Frontiers in Psychology. August. 2018;9:1330. DOI: 10.3389/ fpsyg.2018.01330
- [54] Michel JS, Hargis MB. Linking mechanisms of work-family conflict and segmentation. Journal of Vocational Behavior. 2008;73(3):509-522. DOI: 10.1016/j.jvb.2008.09.005
- [55] Ishizuka P. Social class, gender, and contemporary parenting standards in the United States: Evidence from a national survey experiment. Social Forces. 2019;**98**(1):31-58. DOI: 10.1093/sf/soy107
- [56] Collins C, Landivar LC, Ruppanner L, Scarborough WJ. Covid-19 and the gender gap in work hours. Gender, Work and Organization. 2021;28:101-112. DOI: 10.1111/ gwao.12506
- [57] Chai L, Schieman S. Work-tofamily conflict and Children's problems with school, friends, and health: Household economic conditions and couple relationship quality as

- contingencies. Journal of Family Issues. 2022;**43**(6):1555-1578. DOI: 10.1177/ 0192513X211026953
- [58] Dapiton EP, Quiambao DT, Canlas RB. Parenting as a moderating factor for research productivity and work-life balance: Evidence from Philippine women academics. European Journal of Educational Research. 2020;**9**(4):1425-1434. DOI: 10.12973/ eu-jer.9.4.1425
- [59] Gallagher A. Do babies matter? Gender and family in the ivory tower. Children Geographies. 2015;**13**(1):128-129
- [60] Mason MA. The baby penalty. The Chronicle of Higher Education. 2013. Available from: https://www. Chronicle.com/article/The-Baby-Penalty/140813
- [61] Mason MA, Goulden M. Do babies matter? The effect of family formation on the lifelong career of academic men and women. Academe. 2002;88(6):21-27
- [62] Bao X, Ke P, Wang J, Peng L. Material penalty or psychological penalty? The motherhood penalty in academic libraries in China. The Journal of Academic Librarianship. 2021;47(6):102452. DOI: 10.1016/j.acalib.2021.102452
- [63] Drago RW. Striking a balance: Work, family, life. Industrial & Labor Relations Review. 2009;**62**(4):Article 11. http://diigitalcommons.ilr.cornell.edu/ ilrreview/vol62/iss4/11
- [64] Benard S, Correll SJ. Normative discrimination and the motherhood penalty. Gender and Society. 2010;25:616-646. DOI: 10.1177/0891243210383142
- [65] The Motherhood Penalty. The American Association of University Women. Available from: https://www.

- aauw.org/issues/equity/motherhood/ [Accessed: January 25, 2023]
- [66] Gaskell A. How discrimination leads to a motherhood penalty in the labor market. Forbes. September, 7, 2021. Available from: https://www.forbes.com/sites/adigaskell/2021/09/07how-discrimination-leads-to-a-motherhood-penalty-in-the-labor-market/?sh=109c7ceb16bf
- [67] Ishizuka P. The motherhood penalty in context: Assessing discrimination in a polarized labor market. Demography. 2021;58(4):1275-1300. DOI: 10.1215/00703370-9373587
- [68] Yu W, Hara Y. Motherhood penalties and fatherhood premiums: Effects of Parenthood on earnings growth within and across firms. Demography. 2021;58(1):247-272. DOI: 10.1215/00703370-8917608
- [69] Killewald A. A reconsideration of the fatherhood premium: Marriage, Coresidence, biology, and fathers' wages. American Sociological Review. 2013;78(1):96-116. DOI: 10.1177/ 0003122412469204
- [70] Mu Z, Xie Y. 'Motherhood penalty' and 'fatherhood premium'? Fertility effects on parents in China. Demographic Research. 2016;35(47):1373-1410. DOI: 10.4054/DemRes.2016.35.47
- [71] Wang-Cendejas RR, Bai L. The Fatherhood Premium: Mirage or Reality? How Occupational Characteristics Matter. Sociology Study. 2018;8(4):157-170. DOI: 10.17265/ 2159-5526/2018.04.001
- [72] Mari G. Is there a fatherhood wage premium? A reassessment in societies with strong male-breadwinner legacies. Journal of Marriage and the Family. 2019;81(5):1027-1284. DOI: 10.1111/jomf.12600

Parenting and Work-Family Balance in the Twenty-First Century DOI: http://dx.doi.org/10.5772/intechopen.110689

[73] Svedmark, L. The family gaps in pay: Empirical evidence from Japan. Master thesis in Business and Economics. UMEA University. Spring 2021. Available from: http://umu.diva-portal.org/smash/get/ diva2:1569768/FULLTEXT01.pdf

[74] Modern Parenthood. Roles of Moms and Dads Converge as They Balance Work and Family. Washington, D.C.: Pew Research Center; 2013. Available from: https://www.pewresearch.org/social-trends/2013/03/14/modern-parenthood-roles-of-moms-and-dads-converge-as-they-balance-work-and-family/

[75] Espring-Andersen G. Families in the 21st Century. Stockholm, Sweden: SNS Förlag; 2016

[76] Jarvis P, Georgeson J. Parenting and family in the 21st century. In: BERA-TACTYC Early Childhood Research Review 2003-2017. United Kingdom: British Educational Research Association – Association for Professional Development in Early Years; 2017. pp. 30-48

[77] dos Santos CVM, Campana NTC, Gomes IC. Egalitarian parental care: Literature review and conceptual construction. School and Developmental Psychology. 2019;35:e35311. DOI: 10.1590/0102.3772e35311

Chapter 4

Desolate Childhoods: The Surrender of the Adult World – Notes from the Pandemic of SARS-CoV-2

Mónica De Martino and Maia Krudo

Abstract

The weakening of the authority of the adult world is a phenomenon studied for decades. State intervention, whether in its political or technical facet, is no stranger to it. Individualization processes and the predominance of projects from the adult world over the child world also have a fundamental influence. We can point out that most children today live in solitude, without a principle of clear democratic authority on the part of their parents, who also do not feel comfortable placing limits on a technical childhood that they do not recognize. The chapter will basically deal with these topics as an essay, based on our research and university extension activities regarding the impacts of the pandemic in the family environment.

Keywords: childhoods, adolescences, paternities, maternities, pandemic

1. Introduction

The bibliographic review at the European and Latin American level places emphasis on the consequences of the pandemic on families and the isolation measures taken by various governments. We briefly mention increase in situations of intergenerational and gender violence; increase in psychiatric medicalization in children and adolescents; the dramatic situations experienced by the elderly population; difficulties in schooling and access to networks to monitor educational tasks, rigor of teleworking in the domestic space, lag in health care for various pathologies and medical problems not associated with COVID, etc. [1]. However, it is worth asking about the situation prior to the pandemic, the social base in which it has been installed and that forms both the starting point and the social potential to mitigate such impacts. This is what this chapter is about, and it does so in a specific way.

The authors are investigating the impacts of the pandemic at the family level, in the Childhood and Families Academic Area, of the Department of Social Work of the Faculty of Social Sciences of the University of the Republic—Montevideo. The project obtained funding from the Scientific Research Sector Commission and the University Extension Commission of the Faculty of Social Sciences. The research lasts from May

39 IntechOpen

2022 to May 2024, while the Extension Internship Space, from November 2022 to March 2023. The activities of the latter have been developed between the months of November and December 2022, and they will end in March 2023. The reflections that ordered this chapter were provoked by the result of these last activities.

The purpose of the Extension Internship Space was to support, through extension actions, the lines of research linked to the impacts of the pandemic, in the daily life and the psycho-social experiences of children and adolescents in popular sectors of Montevideo.

Table 1 that follows briefly summarizes: 1—the instances of work with children and adolescents and their participants; 2—the institutions for the protection of children and adolescents in which they were developed.

Before beginning with a brief description of the information collected in the instances with children and adolescents, it is convenient to distinguish the organizations where work was done. The activities were carried out in two civil associations in Montevideo. First, the Gurí Children's Club, which belongs to Vida y Educación and is located in the Manga neighborhood. Vida y Educación has among its objectives the development of educational activities that promote the defense of the rights of children and adolescents. On the other hand, the Youth Center and the Papryka projects of the NGO Pablo de Tarso are located in the Borro neighborhood. The NGO Pablo de Tarso is a civil association managed by the Pablo de Tarso Foundation that works with children, adolescents and families of Borough D seeking to promote social inclusion and the construction of full citizenship.

It should be noted that such institutions are located and address the problems of popular neighborhoods in Montevideo, the capital of Uruguay. These neighborhoods are inhabited by unskilled wage workers, with unstable occupations, without legal protection. The houses are built by low quality or light materials. In general, they are neighborhoods that bring together the working class with lower incomes and the

| Date | Applied techniques | Institutions and programs | Participants Thirteen children and one educator participated in the first instance |
|------------|--|---|--|
| 24/11/2022 | Plastic expression through drawings | Gurí Children's Club of the NGO Vida y Educación | |
| 24/11/2022 | Plastic expression through drawings | Gurí Children's Club of the NGO Vida y Educación | Fourteen children and two educators participated in the second instance |
| 7/12/2022 | Plastic expression through drawings | Papryka Project of the Youth Center of NGO Pablo de Tarso. | Eight pre-adolescents, and adolescents participated together with the project coordinator. |
| 7/12/2022 | Plastic expression through drawings | Children's Club and educators of the NGO Pablo de Tarso. | Thirteen children and two educators participated |

Table 1.Summary of the information on the extension internship space.

so-called precarious settlements (a group of light material constructions, without access to basic services). The most pressing social problems are unemployment, low income, access to housing and public services, difficulties to continue educational trajectories, etc.

This does not mean that the difficulties that we will talk about throughout the chapter are proper or ontologically attributable to the popular sectors. We reiterate, the results obtained were the ones that instigated our reflections on the plots of civilizing processes that allow us to speak today of the defenselessness of childhood and adolescence at a general level, in the various social classes. The cited bibliography also points this out.

A work proposal was recommended to each group that sought to promote the opinions and expressions of the participants, mainly on three fundamental axes: impacts of the pandemic on the link between the adult world and the child/adolescent world, and links between peers and educational trajectories during the pandemic. The activities were based on projective techniques. Operationally, we opted for techniques supported by the graphic production of the participants, which reflect, by definition and nature, as we will explain later, their perceptions and personality.

From this epistemic position, a construction of meanings and constructed significations [2] was elaborated together with the children who are the agonists and protagonists of the problem we are addressing. For this, the technique used was oriented to return the word to children and adolescents who, in general, do not have it. An attempt was made to reveal what they tell us, with their representations, their circuits, their transits through public and domestic physical spaces.

This technique is associated with the Freudian concept of projection. At first, Freud defines projection as a defense mechanism underlying paranoid problems. That is, the concept of projection was associated with the clinic of psychological pathologies [3, 4]. Subsequently, Freud himself expanded this concept, making it independent of its pathological character and approaching it as a "placing" outside unconscious elements associated with the various behaviors of the subject even in "normal" situations. In tune with this, this projective technique can be understood, in a very broad way, as tools sensitive to unconscious aspects of all human behavior [5].

When applying this technique, the playful aspects associated with her have also been assessed. This instrument makes it possible to analyze painful aspects through a necessary mediation, such as graphic products. We obviously recognize that distances must be saved with the elaborations of Freud [4] and Winnicott [5] regarding play in clinical settings. That is to say, this technique is not understood as transitional objects, and it is only indicated that in a certain way they are based on a "game"—drawing—that allows a more "mild" handling of what is painful. The technique made available would play the role of signifiers from which the children and adolescent, in this case, allude to meanings associated with the problem in question [2–5].

To this way of understanding projective techniques was added the consideration of the group space as a sort of mediating zone between subjectivity and objectivity, between a painful reality and the capacity to symbolize it. We are not talking about a space of transition between the original experiences of the mother or father and what is real. We speak of the group as a zone between the conscious and the unconscious, between the material and symbolic and between the individual and the collective.

Perhaps Winnicott [6] is the author who allows these last two aspects to be brought together. If for the author the game is always a creative, liberating and essentially symbolic activity, it is also a space different from the external world and the internal world. The author defines a line of accumulation or growth: "there is a development

that goes from the transitional phenomena to the game, from it to the shared game and from it to cultural experiences" (p. 76). In this sense, an attempt was made to reconcile the individual "game" with the group spaces for reflection, trying to feed children and young people other ways of positioning themselves before their situation.

Let us see below in a synthetic way, the results obtained and the theoretical elaborations that they have generated for us. Our reflections have as a central concern to indicate that, more than talking about the pandemic and its impacts, it developed at a certain historical moment, previously installed. We should not only talk about the effects of the pandemic. Instead, we should talk about what still needed to be done at the moment it abruptly interrupted our lives. Not everything is a pandemic, but it is the dialectical tension between the pandemic and the societal context in which it was installed.

Thinking *ex nihilo* about a phenomenon like the pandemic, out of nowhere and with nothing to generate it, is making a big mistake in perspective. That is why we will not place the emphasis on data on the pandemic, which is extremely widespread today, but rather on previous family processes that caused children and adolescents to assume the impact of the pandemic alone, in most cases. In Foucault's words, we try to make "a history of the present," genealogically speaking, "an unspeakable search for the beginnings," in this case, for the desolation of childhood [7].

2. Desolate childhoods and adolescences

2.1 About the results achieved

In a synthetic way, we will gather in three items the results produced by a primary interpretation of the empirical material collected.

The results deserve a first observation. The space of these children and adolescents is a strictly domestic, endogamous space, associated with its natural course within the framework of the house. Beyond the differences between the genders, we could indicate a very marked tendency toward domesticity in children and adolescents of both sexes, without any extension to the already known or daily community or neighborhood. It should be noted that in Uruguay the government did not take measures to confine the population, but only a call for responsible freedom, regarding the care of oneself and others. The only more extreme measure was the suspension of classes throughout the educational system, going on to teach them under a virtual regime. The drawings or narrations do not incorporate activities with friends (except through phone calls in some cases).

The low density of their bonding networks, internal or external to the family, shows us that we find children and adolescents alone, in situations of anguish and isolation, denoting a very elementary family life. For example, the drawings do not represent family members, only children and adolescents with their siblings. In the cases in which the adult world is drawn, it is in its condition as a patient or sick person: in bed, getting vaccinated, etc. There are no drawings in which the family is carrying out common activities or adults carrying out habitual tasks such as cooking, going to or coming from work. This absence of the adult world in the drawings made and the lack of common activities among the various members of the family indicates phenomena that, we believe, go beyond the pandemic. In a way, the limits between the adult and child world are blurred: The few adults that appear are patient, like children. Except

in one case where the drawn adult appears armed and inside the house itself, which is explained by the child in the dialog held. This violent world is also observed in other drawings, where weapons and war transport appear.

Secondly, the works mark a clear logic of power, of domination, where the dominant and masculine social order is so deeply entrenched that it requires no justification, it imposes itself as self-evident and is taken as "natural." This is expressed in the constant presence of doctors or vaccinators, all of them men, in the family space itself. This social order, where the public world that appears is of a sanitary and masculine nature, is achieved thanks to the almost perfect and immediate agreement obtained from social structures such as the social organization of space and time and the sexual division of labor, and on the other hand, of cognitive structures inscribed in bodies and minds. Field and habitus reinforce the patriarchal and adult-centric system. In the words of Bourdieu [8] "history made institution" and "history made flesh." These stereotypes attributed to each sex crystallize in a very radical way in the identities of these children and adolescents.

Finally, we have analyzed the spatial configuration of the drawings where we can highlight figure and background, which are in relation to container and content. The sun, or clouds, or birds do not appear in these drawings. Just big viruses or lots of little ones. Neither do other living beings appear around the participants (except in some cases siblings, fathers as patients and male health personnel).

Neither do the houses rest on the ground. The houses are drawn in the air, without foundations that root them and contain them. The setting is almost empty of content: Isolated houses, very little furniture and the settings are decontextualized, there are no elements that account for the time of day, the ground is not drawn below the houses or the people, nor is the sky above, as we already said. It is practically a childish scenario, no ground, no sky, no time, no space. It would seem that these children and adolescents have a lack of support, elements that frame or contain them. They appear located in blurred spatial-temporal coordinates, a kind of "social limbo." Their houses are like any typical house from any place, at any time, in a popular neighborhood of Buenos Aires, Montevideo or Santiago. The vital-nutritional elements that surround these childhoods and adolescences are almost imperceptible, as if they traveled through circuits of low social intensity and therefore of low vital intensity. The sun/ symbol does not exist. Perhaps because of not waiting for the future to bring different events than today. Time passes and daily life is not invested with relevant and significant events, except for the outbreak of the pandemic: face masks, vaccines, doctors and nurses and many viruses. A sanitary and masculine world faced by children without clear adult references.

It should be noted that some adolescents did not resort to drawing but to sequencing the activities carried out throughout the day. This arrangement under his pandemic life reinforces what has been said: Activities shared with adults do not appear. In the case of girls, there are domestic and care activities, aspects in which the pandemic has had a strong impact in Latin America, burdening the shoulders of women of various age groups with this type of task, underlining the sexual division of labor, already mentioned above.

In this panorama of retreat from the adult world expressed by these children and adolescents, we believe that three phenomena or three civilizing processes come together that we will consider separately for the sole purpose of exposition. In this way, we avoid the stigmatization of adults by placing them as sons and daughters of their time. It should be noted that such processes are also developed in a historical context characterized by the highest historical levels of concentration of wealth,

increased poverty and loss of wages according to the Picketty report [9]. All this before the pandemic context, which what it has done is catalyze these structural and cultural processes already installed for decades: the cyclical crises of capitalism and the dilution of the borders between adults and children.

2.2 Investigating Oedipus and his present: the withdrawal of the father state

What we want to point out, basically given the length of the article, is where the legality of paternity/maternity is anchored in the present, in a context of substantive changes such as those that can be observed in the twentieth and twenty-first centuries? Will such paternity and maternity and their legality be up to sustaining the world of children and adolescents affected by the pandemic? Without intending to exhaust the subject, we bring to this dialog Tuber [10] who criticizes the Lacanian constructions regarding the name-of-the-father saying:

Thus, we have on the one hand the sublime father, the great man, the peacemaker and on the other, the father who demands blind obedience to his authority and an unquestionable absolute belief. Consequently, the paternal role cannot transmit only the principle of reason, without also bringing cruelty and irrationality. ([10], p.78).

This conception of the symbolic father is related to a context in which man symbolically appropriated the maternal origin of life. The Father of Oedipus [4] is the real bearer of the prohibition that prevents the union with the incestuous object [11] and that will return as a symbolic authority later in "Totem and Taboo" [12]. This father is not the third party at stake, the one that prevents the union with the incestuous object; on the contrary, it is his murder that prevents the symbolic union with the mother, as an object of desire. And that collective murder should have happened to go from the animal state to a social state, culturally integrated. While in the myth of Oedipus someone kills the father in an exceptional way, in Totem and Taboo we all did it, the Horde did it and that crime shared by all plays the role of a common cultural base: The Father becomes a symbol. This father coexists with the father jouissance, alive and sexual and the dead father and interdicting symbol [11, 12].

Later, in "Moses and the monotheistic religion" [13], both figures are synthesized in Moses, the man who, sweeping away superstitions, introduced monotheism and, if you like, the idea of a world governed by a single rational system. But this Moses lives with the figure of a furious and vengeful Moses if his people betray him. This vengeful God is not the Father Jouissance, but the one who introduces the prohibition of incest, the norm, the cultural and expels all jouissance or sexualization.

The Father of "Moses and monotheistic religions" is not the father of "Totem and Taboo" who imposed the prohibition knowing jouissance, but this Father prohibits without knowing it and with fury [11–13].

The modern bourgeois family, by unifying within its bosom the two paternal functions (object of identification and, on the other hand, subject of capital authority, that is, cutting function and nurturing function), has generated not only what some call a "crisis of Oedipus," but also a crisis of investiture of authority figures [11]. This historical process that Freud works on is historically associated with paternity and maternity in which the maternal origin of life has been expropriated by the father, from patrilineal lineage. In addition, it is clear that an affectionate, nurturing mother and a father who orders and sets limits reproduces the sexual division of industrial labor and the "spheres" of production and reproduction.

Although we do not have answers, we must think about how the nutritional and cutting functions are developed today, given the major changes observed at the family

level. There will always be someone in the family context who performs or complements them. More than the presence of one or the other parent, beyond their sexual identities, it is important that the function is fulfilled, but the truth is that the conditions for the exercise of these paternity and maternity duties must be thought from another perspective. In the general historical context of the loss of investiture of adult figures, it is especially the figure of the interdicting father that is in the process of withdrawal, with its diffuse limits [11].

What we want to invoke is that various authors for decades have questioned the validity of the Oedipus complex as a structuring element of the subject and/or pathology [14]. The advancement of women's rights, the different ways of being family, the diversity of cultural patterns to those that Freud observed, must be recognized and included in the analysis of the validity of the oedipal triad. Heritier [15] says that what is truly prohibited would be the contact of bodies and the transmission of flows from one body to another. In the tragedy, the Chorus does not tell Oedipus that he had a sexual relationship with his mother, but rather that he found his father in the womb. This is the unthinkable. If the limit disappears, we enter the terrain of the indiscriminate. And it is precisely in the field of sexuality and sexual relations and kinship, where the difference between the sexes and generations acquires greater relevance. Also in the paternal and maternal functions, due to the place they occupy in the construction of the human psyche, since when the barriers between the different and the identical are erased, there is a risk of falling into the indiscriminate, the unrepresentable.

What effects will all these changes have on future generations? We do not know yet, but perhaps it is a good indicator the greater presence of adolescents and young people with difficulties in the construction of a consolidated self or identity center. But these transformations, this return of the Father of the Horde, of a certain lack of differentiation between the adult and child world, could be speaking as much of an adulthood that denies one of its almost historical psychic conditions and demands, as of unrealized childhoods and adolescences.

Just as the Interdicted Father withdraws before the return of the Father of the Horde, the Welfare State withdraws, giving way to a light and distant state, as we will see later.

2.3 The disappearance of childhood

If paternity and maternity must be reread in this social context, so must child-hood. Neil Postman, an Australian sociologist, who has studied the ways in which technology impacts society, has a suggested book entitled The Disappearance of Childhood [16].

The central thesis of the work brings together three elements: the appearance and dissemination of TV, the collapse of the differentiation of information (adulthood and childhood) and the relativization of the adult and childhood categories.

Inescapable this author for our reflections. What the author maintains is that the lack of differentiation of information for adults and children, which TV shows indiscriminately to the adult world as well as to the children's world, would cause less distance between adults and children, or, in other words, the difference between them would begin to blur. We must not forget that Postman writes what he writes when TV was the quintessential mass medium in the USA.

In other of his works he tells us:

In each tool there is inscribed an ideological tendency, a predisposition to build the world in one way and not another, to value one thing more than another, to develop a sense or an ability or an attitude more than others ([17], p. 26).

Adding: This is how media technology works. A new technology does not add or take away anything. It changes everything ([17], p. 31).

But what relationship does the effect of TV have on the relationship between adults and children? For some authors, apart from Postman, TV blurs the line between adults and children. It is that TV does not require much instruction to be handled, the information it provides is simplified and does not discriminate against the audience. That is why the author indicates: "Without secrets, of course, there is nothing that can be called childhood" ([16], p. 80).

TV reveals secrets, children access issues previously reserved only for adults and TV makes public what was private [16]. This brings an unthinkable consequence in its repercussions: "The idea of shame is diluted and demystified" ([16], p. 85). But this does not mean that childhood is a "without shame" childhood; on the contrary, it means that it is a childhood that has not been able to build relationships of authority with adults [16].

Television forces the entire culture to "come out of the closet." In its desire to find new and sensational information to retain its audience, TV has to touch all the existing taboos within the culture—incest, divorce, promiscuity, corruption, adultery, sadism, masochism—one of them is now no more. Than a subject for a television show. And, of course, in this process, each of them loses their role as a secret that belongs exclusively to adults ([16], p. 93).

The author goes on to point out that, by using material from the adult world indiscriminately, TV is projecting a type of person that is completely new. We could call this person "the adult child." TV promotes as something desirable many of the attitudes that we usually associate with childhood. For example, an excessive need for immediate gratification, a lack of concern for the consequences, an almost promiscuous concern for consumerism, the author points out. Or the idea of eternal youth now facilitated by various cosmetic and surgical devices.

If you can conceive what it means to be an adult, you can conceive what it means to be a child. In any case, whatever description one wants to give this transformation that is taking place, it is quite clear that the behavior, attitudes, desires and even the physical appearance of adults and children are becoming less and less distinguishable ([16], p. 88).

If we say boy or girl, we almost immediately associate the future, an idea of the future. In communication terms, the author tells us that children are messages that we send to the future, living messages. But TV does not give a sense of history, past or future, it is eminently present. The author borders on cruelty when telling us that the message that TV can convey is that children do not matter, if they are exposed to the same sexual arousal, to the same consumerism as the adult world.

This leads us hand in hand to a fourth process that has crossed Western societies in the twentieth century and that has placed individualization processes and the exaltation of adult life projects as a form of biographical construction at the center of the scene. Successful.

2.4 The fading of the structures and the staging of risk management

Since the 1980s are observed, there has been a marked sociopolitical tendency to individualize social problems and understand them in terms of individual

"disabilities" and failed acts of the self, when it comes to socializing or entering the market in one way or another. Poverty, understood as individual traits or attributes, is already a characteristic assumption of social policies, not only in Uruguay but in the region. It must be noted, then, that the call to build reflexive biographies or narratives of the self has become an emblematic feature of late capitalism.

We are talking about a supposed sovereignty of the subjects, who, with more options at their disposal in the various social spheres, could choose between different "opportunities." At the same time, they will be held accountable for the decisions they make. The lifestyle is thus understood as a "project," based on a supposed freedom and individual conscience. The contemporary subject is understood as clearly reflective, supposedly informed and rational when making their decisions [18–20].

The Theory of Reflexive Modernization, led by Beck, is also built by Giddens [18] who broadly shares the following diagnosis: In the face of the crisis of traditional identity models, the constitution of the self becomes a personal reflective project. It is the individual who has to interpret his own past, justify his options, choose how to be and how to act, in order to reflexively build his own identity, in a narrative that unifies the experience and provides coherence to the self. This open nature of identity would open up new possibilities of autonomy for individuals and would democratize, according to the author, social relations.

Scott Lash [18], from the USA, for his part, completes the map of the main voices in the debate. In his proposal, in contemporary society, more than the preeminence of the actor over the structure, new structural conditions of reflexivity are noticed, which result in a complex dialectic between both instances. In this way, reflexivity is a relational concept that articulates two simultaneous meanings: On the one hand, it is the achievement of a greater independence of the subject with respect to tradition and ferreous social formations, and on the other, the constant demand for self-definition that is it presents to the subject once the classical sources that organize the social map have collapsed.

Unlike his theoretical companions, Lash [18] does believe in the validity of class polarization, although now the conflict would move from the place that the subject occupies in the work, family and political structure, to the question of the possibilities of access and the position occupied in the global structures of information and communication, given that today the capacities for the processing of the goods of reflexive production are acquired in replacement of the traditional social structures. Altered then, the canonical principle of accumulation appears a new class of excluded, who constitute what the author calls the losers of reflexivity, that is, "those who remain outside the limits of the informational society."

From a critical perspective regarding these authors, Bauman [21] recalls that initially work was the main tool for building one's own destiny. And, once chosen, the social identity accompanied the individual forever. However, the motto of the new times is flexibility. The author focuses on the term flexibility because he considers that whatever identity is sought and desired, it must have the gift of flexibility, and it is necessary that it can be changed in the short term. That is to say, the identities of late modernity are temporary and ephemeral for Bauman and their construction requires consumption. And in this sense it indicates: "the paths to reach one's own identity, to occupy a place in human society and to live a life that is recognized as significant require daily visits to the market" ([22], p. 49).

This individualizing tendency, which advocates biographical resolutions to structural problems, impregnates the conception of poverty of the current social protection systems in our country. And it places a successful challenge on the middle

and wealthy classes, with a clear emphasis on the vital projects of adults regardless of the responsibilities regarding the world of children and adolescents. At least this can be derived from reading the book Love and Eroticism in Modern Societies by A. Giddens [23].

The centrality of individual success, the understanding of the family as a network of individual adult projects, the biographization of structural social processes, place major responsibilities in the adult world, which could be interpreted as a change of course at the time. If the work, the directed and cumulative labor trajectory, the virtual absence of uncertainties, the modern, monolithic educational systems, in short, an "institutional program" of modernity that began to collapse, it is important to recognize the impacts of such transformations on the adult world. Adult world required to demonstrate its success in the construction of a healthy and rational biography. What spare time does the adult world, pressured by said demands, has to listen, care and be of emotional support for children and adolescents nowadays? [24].

3. Conclusions

Taking items 2.2, 2.3 and 2.4 as a starting point, we have tried to analyze some of the transformations experienced in the subjectivities of adults, children and adolescents. In these conclusions, we will try to close the items developed by trying to build a continuum of adult and non-adult subjectivities with the understanding that the adult's actions influence what a child or adolescent is like, since the adult is an important figure of authority and affection.

The challenges for parenthood in the twenty-first century can be summed up in the blurring of the adult that has consequences, not unique, in the new ways of being a child or adolescent. We have seen how, in item 2.2, we have questioned the psychological concept of adult that is traditionally handled.

This traditional psychological concept would seem to lose its legality once the Welfare State withdrew (item 2.3), demonstrating the links between the psychological and the economic, political and cultural aspects of the time. This mutation of the ways of being an adult and, therefore, of being a child, are the plot that allows us to understand the difficulties and challenges that fathers and mothers assume in the exercise of their function.

Functions that must be developed in the face of the new ways of being a child: being an adult child. In the continuum that we intend to synthesize, we will see how the context becomes a text in adolescent and child manifestations and how, in turn, this text is what allows, mistakenly or not, to find a place in this highly individualized society. It will bring to mind the corner where we grew up and the memories of our parents and grandparents. In this journey, we will discover resonances, either pleasant or unpleasant, that lead us to consider how we position ourselves as adults and parents.

In the first link of this continuum, we find the nineteenth century model of being an adult, marked by the incipient development of capitalism and the principle of authority (political, religious, by generation and gender). Capitalist development implied the domination of impulses (including sexual ones), the subordination of women to men and the subordination of children and adolescents to family power. We cannot leave aside the medical power that, in an organic alliance with the womanmother, allowed home control of hygiene and educational principles, which were instilled in children and adolescents. Belonging to a social class implies having ways

of being and acting; religion form of thinking and the incipient democratic life the principle of meritocracy.

Children and adolescents were something different from the adult world, who, based on the principle of authority, were transformed into beings to be governed, educated under a series of adult-centric devices. The adult's gaze was imposed, and the child's and adolescent's voice were not heard. The figure of the Father of the Horde (item 2.2) is timely. On today's patriarchal societies, the feminine, the childish, the adolescents became beings to be governed. As an example, it should be noted that the concept of youth appears in this period of the beginning of capitalism, which in Latin America can be located in the nineteenth century. The expression youth was associated in the first place with the young worker, who was part of the ranks of the proletariat and for whom the world of education was foreign to him. Youth was associated with a certain social "dangerousness."

According to what was expressed and from the various lines of subordination, affection was not frequently expressed, and when it was, it was done harshly. This nineteenth century model gives rise to other ways of being in the twentieth century. Century of life in strengthened democracies; of the rights of the individual and citizen; of humanized capitalism under the Welfare State. Century of the revolution of the rights of women and children and adolescents. Century in which they were considered subjects of rights and political subjects. Also, under the Parsonian model, the democratic school and family were the institutions that generated childhood and adolescence.

It is a new period, the three decades after the World War II, the decades of victorious capitalism and cushioned by the interclass pacts financed by the Welfare State. In this period, the school and the family were the institutions that created participating childhoods and adolescences, educated with new pedagogical forms that placed the child as the starting point and center of educational action. On the other hand, the families were democratized although the imputed gender roles still persisted.

The Father Legislator, who fulfills the court function and makes the Law enter the family nucleus, is the father of the first decades of this twentieth century. The social role of women was modified, demonstrating the advances in her rights, but within the family, the nurturing and affective function was what she sustained. With fathers and mothers with clear family locations and with an organic alliance, now between family and school, childhood and adolescence continued to be viewed from an adult-centric perspective.

These last three decades of well-being have allowed adults to plan their lives and their families based on certainties, in full-employment societies. With certainties and with institutions that give meaning (unions, companies, churches, universities, etc.), transforming the world into something known and safe guided by protective states, as we pointed out in item 2.4. Starting in 1973, from the crisis of the global accumulation pattern of capitalism, the so-called late capitalism was installed, or, from another perspective, the moment of globalization arrived.

In this Second Modernization, according to the Reflexive Modernization theorists, with the opening and mobility of the National States, with a strong weakening of traditional industrialization and with the highest indicators of poverty and concentration of wealth never seen before, fathers and mothers had to face uncertainty, that is, uncertainty in their job projections, in their salary indexes and in the access to goods and services. Its future was no longer assured, the Parent State had collapsed, and the figure of a weak and outsourced state was emerging. How to educate your children for an uncertain future? Did he reach the ideal of a university son? Was it enough now

to think of sexual identity as something binary? Could they continue with the dream of a home of their own? How to take charge of his own adulthood, whose parameters were no longer those of the decades of well-being?

On the other hand, how to be a child or adolescent in a family that has changed and attending an educational system that still looks to the past? Family and School were institutions that began to present problems to generate childhoods and adolescences in an uncertain and hostile world. Beyond the progress in the rights of women, children and infancy, family and school began to be weak entities to comply with a socialization process appropriate to the new era.

The disagreement between an adult world worried about uncertainty, including that of one's own love ties, concerned about building a biography that gives pleasure and a world of children and adolescents facing screens, is one of the biggest challenges for being parents today. The loneliness of children and adolescents, as we have seen in the empirical data that we have brought to the beginning of the chapter, the excessive medicalization for their anxiety, insecurities and loneliness and the pathologies generated by the difficulties in the construction of a consolidated self are some of the problems that childhoods and adolescences place today in the adult world.

For its part, it would seem that the adult world gives up its condition by not taking charge and sustaining the maturation process of infants and by not taking charge of the limits that it must impose. The child is treated as an adult. He is transformed into a consumer to obtain an ephemeral satisfaction for his needs. On the other hand, the advent of new communication technologies has transformed childhood and adolescence into consumers, too, of new realities. As we have said before, when the "secret" is erased and the child accesses what corresponds only to the adult world, the borders between both worlds are diluted.

Adults in crisis and uncertainty, childhoods and adolescents alone, those of whom have winded up without clear guidelines and have transformed into compulsive consumers and the object of medicalization, which raises multiple questions that go beyond the limits of this chapter. But what we are sure of is that the adult of the second half of the twentieth century and the twenty-first is an insecure adult. The process of high technification of parental functions, under the Welfare State, is not unrelated to this.

As we have already analyzed in item 2.3, the adult dependent on technical opinion to exercise her paternity is the product of the good intentions of the Welfare State, its unexpected result. The weakening of parental authority is an element to take into account when analyzing how to be parents in the twenty-first century. C. Lasch [25] comes to our aid again in a text in which he analyzes American culture after three decades of well-being, post-World War II.

The author analyzes various processes. It indicates that the last hundred years of family crisis, this is, the passage from the nineteenth to the twentieth century, have ended with family authority. But the author does not speak of a family crisis in conservative terms, although some believe so. Family crisis in terms of once more state interference in family affairs. In such a way that parental and maternal functions underwent a high process of technification, among other things thanks to the action of the care professions and other knowledge.

The precariousness of paternal and maternal authority has placed the adult world in a difficult situation when it comes to setting limits or transmitting values or visions of the world. The adult world is somehow infantilized before the state and professional authority, in a situation of dependency and subordination [26].

Desolate Childhoods: The Surrender of the Adult World – Notes from the Pandemic of SARS-CoV-2 DOI: http://dx.doi.org/10.5772/intechopen.110688

The author, with some pessimism, makes a description of American society from a psychological perspective. Beyond the cultural specificity, his ideas can make us think about the national and regional reality.

When talking about narcissism in that time period, he does not put selfishness in the first place. Instead, it creates dependence on others to found and sustain our selfesteem and our judgments as a defining feature.

Narciso drowns in his own reflection, without ever understanding that this is just a reflection. The essence of the story is not that Narciso falls in love with himself, but that he does not recognize his own reflection because he lacks any notion of a difference between himself and his environment ([25], p. 289).

This narcissistic perspective allows us to understand the exposure of and to the mass media, the importance given to youth, to the present and its enjoyment, the various new age techniques that teach it, to the detriment of the intimate, the secret, the "academic education "and of history and its meanings.

Narcissism is a metaphor for the current human condition. As Erich Fromm indicated, beyond his psychoanalytic notion, one must also speak of narcissism as vanity, admiration, satisfaction, that is, as individualism asocial, oblivious to any expression of cooperation, kindness and cooperative participation.

If we unite this narcissism as a metaphor for the adult human condition and we add the figure of the *adult child* and *child adult* Postman [16], it is worth asking: Who sets limits, healthy authority, informs traditions, etc.? [24]. Considering the data we were based on, we ask ourselves, who takes care of the *adult child* whom has lived throughout the pandemic as best as they could?

We are thinking about something that we call, provisionally today, the surrender of the adult world. Adulthood perhaps today indicates the dose of claudication: of authority, of placing limits on the little one, of saying no and of carrying out activities that adults historically did. And do they also give up on the nutritional and cutting functions? Up to some point, yes, and it takes just for us to take a look back into our introduction to see how the children and adolescents do not draw their parents or if they do, they portray them as sick or ill. If such functions are weakened, the path could lead us to the Horde, to the undifferentiated and to a world where the incest taboo is generally abolished. The raise in cases of child sex abuse is an indicator of what we stated above. It is that the child adult may have forgotten what it means to be an adult. But the one who cannot give up is the adult child, since he has not even known childhood when he is subjected to the pressures and perversions of us, supposedly conscious adults [24].

In this panorama that we designed we have placed some of the challenges that parents face in this twenty-first century. Under this period diagnosis, sustaining in the post-pandemic picture would still be seem to be a debt owed and of life.

Acknowledgements

We express our gratitude to the Sectoral Commission for Scientific Research and the Extension Commission of the Faculty of Social Sciences, for the financing and recognition, both from the University of the Republic, Montevideo, Uruguay.

Author details

Mónica De Martino* and Maia Krudo Republic University - Faculty of Social Sciences, Montevideo, Uruguay

*Address all correspondence to: monica.demartino@cienciassociales.edu.uy

IntechOpen

© 2023 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. (©) BY

References

- [1] De Martino M. Notes on families and pandemic. Social Work. 2021;24:123-141. Bogota. DOI: 10.15446/ts.v24n1.90377
- [2] Casas F. Childhood and social representations. Politics and Society. 2006;43:27-42. Available from: https://revistas.ucm.es/index.php/POSO/article/view/POSO0606130027A
- [3] Freud S. Introduction to narcissism. In: Complete Works. Vol. 14. Buenos Aires: Amorrortu; 1984. pp. 1-29
- [4] Freud S. The Oedipus complex. In: Complete Works. Vol. 19. Buenos Aires: Amorrortu; 1993. pp. 1-6
- [5] Vives M. Projective Test: Application to Clinical Diagnosis and Treatment. Barcelona: University of Barcelona; 2005. p. 379
- [6] Winnicott D. Transitional objects and phenomena. In: Pediatric Writings and Psychoanalysis. Barcelona: Laia; 1981. pp. 313-330
- [7] Foucault M. Nietzsche, genealogy, history. In: Varela J, Álvarez-Uría F, editors. Microphysics of Power. Madrid: La Piqueta; 1979. pp. 7-30
- [8] Bourdieu P. Power, Law and Social Classes. The Forms of Capital. Bilbao: Desclée; 2000. p. 345
- [9] Piketty T. Capitalism in the 21st Century. Mexico: FCE; 2013. p. 468
- [10] Tuber S. Figures of the Father. Valencia: Chair Editions; 1997. p. 3230
- [11] Žižek S. The Thorny Subject. The Absent Center of Political Ontology. Buenos Aires: Paidos; 2001. p. 434

- [12] Freud S. Totem and Taboo. Buenos Aires: Amorrortu; 1978. p. 147
- [13] Freud S. Moses and the Monotheistic Religion. Buenos Aires: Amorrortu; 2016. p. 223
- [14] Gil D, Nuñez S. Why Have you Abandoned me? Psychoanalysis and the End of the Patriarchal Society. Montevideo: Trilce; 2002. p. 234
- [15] Hertier F. On Incest. Buenos Aires: New Vision Editions; 1995. p. 174
- [16] Postman N. The Disappearance of Childhood. New York: Vintage books; 1994
- [17] Postman N. Technopolis. Barcelona: Gutenberg Galaxy; 1996. p. 266
- [18] Beck U, Giddens A, Lash S. Reflexive Modernization. Politics, Tradition and Aesthetics in the Modern Social Order. Alliance: Madrid; 1997
- [19] De Martino M. A Gramscian look at social policies in Uruguay since the 1990s. In: AA.VV. Public Policies in Argentina Today. Analysis and Experiences. Vol. 29-41. Buenos Aires: Cooperative Editions; 2007. p. 188
- [20] De M, Family M. State in Uruguay: Ruptures and Critical Continuities (1985-2009). Montevideo: Udelar – CSIC; 2014. p. 256
- [21] Bauman ZC. In Search of Safety in a Hostile World. Madrid: XXI century; 2003. p. 85
- [22] Bauman Z. Work, Consumerism and the New Poor. Barcelona: Gedisa; 2013. p. 75

- [23] Giddens A. The Transformation of Intimacy Sexuality, Love and Eroticism in Modern Societies. Madrid: Chair; 1992. p. 124
- [24] De Martino M. Social Work with Families. Montevideo: UdelaR-FCS-DTS; 2021. p. 354
- [25] Lasch C. The Culture of Narcissism. Barcelona: Andrés Bello; 1999. p. 330
- [26] Lasch C. The Family: Refuge in the World without a Heart? Rio de Janeiro: Peace and earth; 1991. p. 274

Chapter 5

Children's Rights (Absent) in Universal Parenting Training

Hetty Rooth

Abstract

In public health policies the child-parent relationship is regarded as a determinant for children's development and future health. During the first two decades of the twenty-first century Sweden implemented parenting training on a universal level as a proactive measure. Sweden ratified the UN Convention on the rights of the Child in 1990 by which follows that all policies involving children are entwined with children's rights. Still, research on society's universal manual-based parenting training interventions depicts that the concept of rights is not an intrinsic value. Adult norms and a preventive approach tend to rule over health promotion goals involving children. Moreover, research results show that children possess an awareness of democratic values and insight in how to handle a dynamic interchange in daily family life. Children's involvement in parenting training should be investigated and discussed as an ethical public health challenge for the future.

Keywords: universal parenting training, Sweden, UNCRC, liberty rights, participatory public health

1. Introduction: Institutionalized rights and their realities

Internationally political initiatives to implement the UN Convention on the Rights of the Child [1] have institutionalized children's right to participation and self-determination [2, 3]. In Sweden children's rights are established as a core rhetoric in social policies since the country's ratification of the UNCRC in 1990. In 2009 a national strategy to strengthen the rights of the child [4] outlined principles for the parliament, government, state authorities, county councils and municipalities, to ensure that the rights of the child were acknowledged in all activities. As a historic step the Swedish parliament initiated a bill to incorporate the UNCRC into Swedish law in 2020. Legislative measures were taken to ensure that the articles of the UNCRC were considered in all decisions made by public authorities. Literally this meant that other legislation must be interpreted in the light of the convention. Among the Nordic countries Norway had been a predecessor to this move towards consolidating children's rights, as the convention was incorporated into Norwegian law in 2003.

55 IntechOpen

2. Manual-based universal parenting training

In consequence with the Swedish political intention to respect children's rights in all things that concern children, a rights perspective was present when a broad governmental scheme to support the use of universal parenting training was implemented on a national level during the second decade of the twenty-first century. This parenting training scheme was realized through a national strategy for a developed parental support [5]. The strategy was initiated by a government proposal that suggested that parenting training should be available as a universal public health intervention for all parents with children aged 0–17. Local efforts for parental support interventions should be coordinated at central level in the municipalities. Collaboration between public, non-profit and private actors was also recommended, as was national and regional collaboration. Manual-based parenting courses were generally forwarded as an important intervention, but without favouring specific programs at this point [5].

The government's political step bore similarities to other European policy documents that informed that parents need to have access to expert advice in order to fulfil their parental tasks for the benefit of their children's best development [6, 7].

3. Statistics as a fundament for parenting interventions

With referral to the UNCRC, the National strategy for developed parental support stressed that care of children primary is a parental task but also a societal responsibility. Care and welfare issues were moreover described as closely related to parent–child relationships. That in turn motivated society's parental interventions. On one hand from a rights perspective on the other for societal sustainability. For better understanding of the communication between children and their parents, Sweden has since 1985 cooperated with the World Health Organization (WHO). Surveys have been launched to investigate children's welfare in family settings [8]. These surveys have contributed with important information for political decisions in family policies. With consistency the results have shown that a high percentage of girls and boys above 10 years approve of the relationship with their parents. Their parents share time with the children and listen to them [9, 10].

Other national Swedish surveys have pointed in the same direction. As an example, 90% of 10–15 years old children have said that they mostly can rely on their mother's attention, and a bit less so on their fathers [11].

It can be problematic to identify children as a group in statistics and even to make comparisons over time [12]. Evaluating statistical material for policy purposes about children's well-being has been described as based either on a developmental perspective, focusing on the future or a children's rights perspective [13–15]. A predominantly developmental perspective has steered public health interventions towards parenting training in the twenty-first century [5, 16].

The political initiative for public health-based universal parenting training interventions [5] relied on reports about increased mental health problems among young people. Statistics about mental illness among children and young people received a lot of attention in the early 2000s as a growing public health problem. Psychosomatic symptoms such as headache and stomach-ache and psychological symptoms such as depression and nervousness were reported to have become more frequent among school-aged youth since the 1990s.

Following the national strategy on parental support [5], Sweden chose to support parents on a universal level with group-based parenting training. Group-based parenting training can be implemented on different levels, as indicative, selective or universal interventions [11]. Indicative interventions are not relevant for this chapter as they are aimed at families with diagnosed problems and primarily implemented as individual consultations [17]. Selective group interventions are used for problemsolving with families who have perceived difficulties with child behavior. Selective programmes have been in use in Sweden since the late 1900s [16]. Universal group interventions are intended for all parents with children between 0 and 17 years [11] and were recommended for use by the National Strategy for Developed Parental Support [5].

A public health argument for universal interventions referred to the prevention paradox that the few in need could be reached by targeting all parents, without stigmatizing those at risk. Possible benefits of preventive measures to promote health without targeting specific groups were explained in the National strategy in 2008:

"By investing in universal prevention efforts, we are able to reduce the proportion of the population who would later have developed problems if no action was taken (...) with great opportunities to prevent ill health among a big group of children who have not yet shown any early symptoms" [5].

From 2009 manual-based parenting training courses were disseminated to parents for free. Municipalities all over the country were, from then onwards, encouraged to offer all parents with children between 0 and 17 training with manual-based group-based courses with educated leaders and without costs. A problem following this political decision was that few Swedish efforts had been made to produce course material that met the purpose for universal use [18]. In absence of universal courses Sweden resorted to selective manual-based courses designed to adjust children's perceived problem behavior. The selective courses, mainly called programmes, were structured, effect-oriented and demanded high manual fidelity [19]. Some Anglo-Saxon selective programmes were tried for universal use, for example, The Incredible Years [20], Parent Management Training (PMT) [21]; Triple P [22] and Connect [23]. Selective programmes could broadly be labeled either as interaction or communication programmes [16], or as relational or behavior modification programmes [24]. Some of them have theoretical underpinnings from two of these categories.

Behavior modification programmes that are associated with social learning theory [25] like COPE [26] and Triple P [22] are prevalent. These programmes focus on observable and measurable behaviors which are learned from the environment through the process of observational learning [27]. Relational programmes, like Connect, are associated with attachment theory focusing on relational development, security and parental response [28–30]. In a selective context the methodologies have been proven to give some positive results in effect studies measuring parental feedback.

Societal parenting training interventions aiming at all parents have met with criticism from scholars and professionals in the twenty-first century. Some scholars have disputed the applicability of normative manual-based programmes in multi-faceted postmodern settings [31]. The programmes have also been criticized for disempowering parents, children and facilitators, because of prioritizing professional expertise over lay knowledge [32, 33]. Some of these critical aspects were forwarded by the Swedish Council on Health Technology Assessment and Assessment of Social Services (SBU). The council expressed concerns about the efficiency of parenting training interventions as such, stating that traditional welfare issues like staffing levels, health

care in schools, social services and education are examples of structural changes that are "important to consider along with (or instead of) investing in programmes" [19].

The Swedish political effort to educate parents was summarized by the government as an "activity that gives parents knowledge of the child's health, emotional, cognitive and social development and/or strengthens their social networks, based on evidence-based models, methods and applications with a set of values based on the UN Convention on the Rights of the Child" [5].

With that statement in mind, it is the aim of this chapter to shed light on how Sweden's commitment to UNCRC was reflected and transmitted in manual-based universal parenting training implemented in a preventive/promotive setting from 2009 and onwards. Taking the discussion one step further, preventive top-down approaches as opposed to promotive bottom-up approaches could play out in parenting training as distinct practices of governing of parents and children [34].

4. Prevention and promotion

For more than a decade universal parenting training interventions have now been incorporated in Swedish public health policies. Manual-based courses are launched all over the country as a rhetorically rights-based preventive intervention in a health promotion context, formulated to

"promote mental health and prevent mental illness among children and young people, by reducing psychosomatic symptoms like anxiety, depression, sleeping disorders, tiredness and headache among school children, and thus promote health and prevent ill health among children and youth" [35].

Sweden has a long tradition of supporting parents, but parenting training was previously rooted in welfare policies with a heritage of liberal education in the early welfare state [36]. During the first decade of the twenty-first century a least two major changes made way for the shift, from traditional welfare policies that previously had promoted child rearing: A political demand for scientific rationality of caring and evidence-based practices paved the way for new structured manual-based parenting methods, mainly embedded in a Anglo Saxon context [5, 19, 37]. Secondly political target groups for parenting interventions expanded, from parents with babies and infants to parents with children between 0 and 17 years. Underlying theories about parental uncertainty in a risk-oriented society supported the new policies.

From an equity perspective the promotive side of public health polices was paired with prevention in a new structure for Swedish public health in 2018. The idea of connecting prevention and promotion with one another was motivated to enforce the possibilities for successful intervention results [38, 39]. Public health science had at that time long drawn on the same tradition as the positivistic biomedical paradigm where only classical experimental designs were acceptable. Still, since the 1980s, when the WHO proclaimed the importance of health promotion [40], the promotive side of public health policies has played an important part also in preventive interventions. This was apparent in the national strategy for parental support [5] when described by the government as a strategy for universal preventive parental support. The government thus wrote that the goal of universal preventive parental support was to promote children's health and positive development via parents and maximize the child's protection. Such a societal turn to govern the direction of parenting relates to behavioral modification models with roots in the 1960s. These methods were therapeutically used for indicated problem solving, working with the parent (mediator) to

improve the behavior of the child (target) [41]. Such methods were later developed and reconceptualized for group intervention in programmes like The Incredible Years [20], Triple P [22] and similar selective programmes.

As one of the fundamental principles in the UNCRC, the limits and possibilities of children's participatory rights have been discussed and analyzed since the UNCRC was adopted. The balance of power connected to adult's rights and children's participatory rights have thus emerged as an issue of scientific concern [42].

Politically a child's perspective was acknowledged as important when the Swedish parenting support strategy was processed in 2008. The inquiry searched for and found only four research studies and two student thesis involving children's views on parent child relationships in Sweden [5]. Such a white spot in research posed a challenge for the future. Even more so as children's status as informants in research had progressively changed since the 1990s. From being mere objects of socialization and adult interventions, children were now also regarded as subjects and knowledge providers in their own right. The concept of "childism", which emerged in the twenty-first century, can be regarded as a theoretical development from this change.

Some researchers propose that children's participatory rights could hollow out the rights that adults have to make decisions, [43]. What it boils down to is to acknowledge the existence of a power balance between adults and children.

In the following children's perspective as distinctive from a child perspective is considered. There is currently room for discussions about how parenting interventions involve a children's rights perspective.

5. Child perspective-children's perspective

The term "child perspective" has, influenced by children's rights, been frequently used in the twenty-first century, given different meanings depending on the context and who the senders are—politicians, lawyers, psychologists, researchers, social workers or children's rights advocates, etc. [44]. In science the term has been used by researchers both as a theoretical statement of adherence to a child rights paradigm and as an analytical tool. Due to often poorly performed definitions though, its usability in science has been disputed as lacking in clarity and distinction [45]. Some researchers have advocated for the use of the term "children's perspective" to clarify when children themselves are heard or have their say. But even this term has been disputed depending on how the word perspective is interpreted [46]. Today it seems hard, though, to bypass the ontological considerations affecting research about child–parent relations [47].

A natural consequence of ethical stances in child research is to incorporate children's views in research. Keeping in mind that capturing children's perspectives require adults, parents and health professionals to be attentive, sensitive and supportive of each child's expressions, experiences and perception [48, 49].

6. Adult views on children

The emergence of the "new sociology of childhood" [50] during the 1980s challenged socialization as an instrumental and one dimensional concept of child development. Theoretically "new" children emerged, declared as subjective actors in their own right—competent, interacting and socially constructed. Parenting was accordingly conceptualized as a bi-directional communication.

With insights about how interaction works in multifaceted relationships, the use of a child perspective, as an adult assessment of children rights and needs, has been disputed and debated. In the place of child perspective, children's own perspective is claimed as an important analytical stance. Children's perspective is motivated as a rights-based conception of children's status in parenting [46]. Today it seems hard, to bypass ontological considerations in research about child–parent relations [47]. A natural consequence of ethical stances in child research is thus to incorporate children's views in research.

This is not an easy adult task. It requires an open mind and dynamic apprehension of children's ways of expressing themselves. Basically, awareness of diverse constructions of childhood leads away from structures where children are diminished as receivers of instructions [49–51].

7. Childism as a new understanding of childhood

The concept of childism serves as a more recent theoretical base for childhood studies and has developed further from new social studies of childhood. Childism postulates that social theory needs to embrace the more fundamental potential for children's lived experiences, rather than just break down social norms. This requires reconstructing of interdependent social relations and generational orders.

Today childism theories are discussed in a global network with The Childism Institute as its base. Professor John Wall, founder of the institute, argues that by placing children at the centre of research, childhood studies can not only understand children's agency and experiences in their own right, but also develop critical understandings of child-adult relations and social practices [52].

8. Three research studies

As part of a larger research project on parenting training, the adherence to children's rights was studied sing material from two courses which were implemented universally in Sweden: The Canadian Connect programme and the Swedish All Children in Focus (ABC). Both courses were introduced to parents in 2011–2012 [53, 54].

As mentioned above, the Canadian Connect programme [23] is associated with attachment theory focusing on relational development, security and parental response [28–30]. Connect is originally designed as a selective programme, aimed at children between 8 and 12 with declared problem behavior. Connect has also been used universally in Sweden.

The Swedish course All Children in Focus (ABC) was developed by the Karolinska Institute and Stockholm Social Services in 2011 [55] for universal use only. ABC adheres to social learning theory but also uses some attachment theory. The developers were influenced by selective programme effectiveness as reviewed by Wyatt Kaminski et al. [56].

Both courses were chosen for the studies as they posed an alternative to the selective courses based on social learning theory that had been previously recommended in Sweden.

9. Course manuals

The Connect manual is extensive and consists of nine sessions coherently built up to fulfill the programs intentions. One optional follow-up session is also available. The manual contains four components with the aim to develop secure connections between parents and their children: sensitivity of parenthood, cooperation and reciprocity, ability to reflect as a parent and dyadic affect regulation. The group leader's lectures are mixed with role play and discussions with the part taking parents.

An overall purpose of the course manual is to inspire parents and guardians "to think about new ways to understand their children and their behaviour, new ways of understanding themselves and their behaviour as parents, as well as new opportunities in parenting" [23].

The ABC manual is presented in a binder with exchangeable information. The manual is less comprehensive with four basic sessions and one optional extra session. Each of the four parts has a theme: showing love (parental factors, five to one, focus on what works), being there (the interaction chain), showing the way (annoyance and anger) and picking your battles (natural consequences).

ABC seeks to strengthen parent-child relations. The part takers in the course get to learn more about relevant research on parenting and children's development. Role play is also used and short films about day-to-day family problems [55].

Both manuals were analyzed with qualitative content analysis which depicted that children's rights were not included as an aspect in either manual. The ABC manual's introduction mentioned rhetorically that the course values were founded on the UNCRC (1989). In the absence of referrals to children's rights in the manuals, the analysis concentrated on how the manuals displayed adult and child perspectives and if an ethos of children's welfare or liberty rights was present. The findings showed that the course manuals of Connect and ABC harmonized with a risk-prevention paradigm, prominent in Swedish public health policies [53].

The manuals forward advice and guidelines for parents in different ways. The Connect program uses adult preventive perspectives as well as child-oriented promotive methods. Children are to some extent seen as agents in a mutual developmental process with adults. The ABC affiliates to positive discipline based on authoritative prevention and control. The adherence to child autonomy and children's rights is weak.

In relation to the UNCRC, an ethos of children's well fare rights could be traced in the manuals, while a liberty rights perspective was restricted by adult preventive norms.

10. The course sessions

The sessions of Connect and ABC were observed, audio taped and transcribed. Between 12 and 16 parents took part in the Connect course and between 6 and 8 in the ABC course. Two leaders were present at each course. The recorded material from both courses was analyzed separately with discourse analysis [54].

The study confirmed that both courses relied on high manual fidelity. Normative goal-oriented methods were used which displayed adult centred child constructs and unidirectional parenting strategies. In general, parents were positioned as protagonists in the discussions between leaders and parents. Children, on the other hand,

were mainly positioned as secondary actors and subordinate others. Neither children's own perspectives on parent—child relations, nor the children's participatory expectations were explicitly met during sessions. The Connect course leaders used expressions describing human relationships—connection, lifelong attachment, natural conflicts, empathy and slow change. The ABC course leaders used expressions associated with regulation of human behavior show feelings, choose battles and of conflict handling like leadership in critical situations and use of natural consequences.

Findings from discussions during the two course sessions showed that neither Connect nor ABC dealt with children's rights as such. These findings confirm previous research with course leaders of the COPE programme and the Swedish universal course Family workshop. Leaders declared that they were aware of the intentions of the UNCRC but did not discuss children's rights during sessions [57].

Children's rights are highly regarded in Swedish social politics which makes the absence of children's rights as a guiding principle in universal parenting training courses notable. Today universal parenting training is embedded in public health politics which suggests that children's welfare rights and liberty rights have a different standing. Welfare rights, e.g., health and well-being, are an underlying value in adherence to a public health aim to secure children's health and welfare for the future. On the other hand, children's liberty rights, e.g., rights to participation can in a public health context be restrained by an adult position of parental strength, significant for a preventive public health paradigm [18]. The findings of these studies show that children's liberty rights can be limited by a preventive framework where children play a part as receivers of adult norms.

11. A research study with children

Advocacy for children's rights as expressed in the UNCRC [1] should be informed by a moral, ethical and political responsibility for giving strength to children's voices [58]. Consequently, a third study in the research project consisted of semi-open interviews with 11 children aged five to nine whose parents had taken part in universal parenting training courses (72). Qualitive content analysis was used in this case [51]. The research questions did not deal with children's rights as such but aimed to reflect the children's views on their own standing in family relationships. Hence the study intended to shed light on children's capability to reflect on interaction between parents and children—a core issue in universal parenting training.

Today adult power towards children is a matter of concern for many researchers dealing with children: Several ethical issues have been critically discussed in the literature [59, 60].

Where children are concerned, an adult perspective is to some extent inevitable. The findings of this study were divided in themes and categories. As an example, self-protection included to cherish one's own thoughts, withdraw from the adult company and to set limits for one's own space.

Togetherness' involved taking part in decision making, a will to compromise and a wish to behave in a good way. In general, the children expressed that they ideally took on a position that put them on equal terms with their parents in family discussion. Still, they listened to their parents and respected their knowledge. The children showed clearly that they were willing to compromise in decision making,

"I usually badger a bit but then it blows over and after a while we agree" (Lotta, age 8).

Moreover, the interviewed children sensed that they were qualified as active family members themselves. At the same time, they saw their parents as providers and protectors with good intentions.

The children's recollections and thoughts depicted strong desire to be listened to and to be taken seriously. They also displayed trust and belief in their parents' will and ability to meet their expectations [51].

12. Summary

Children's rights as expressed in the UNCRC have influenced Swedish social policies since the 1990s and are, in a European context, regarded as a primary consideration in parenting [61]. Consequently, the Swedish expansion of universal parenting training interventions to cover the whole period of child rearing could have great impact on rights of both children and their parents.

From a child rights perspective, this chapter has focused on how children's rights are realized in two parenting courses that are universally implemented. Two research studies of parenting course manuals and proceedings have shown a lack of adherence to children's liberty rights as expressed in the UNCRC. Moreover, children are placed as subordinate others in the course material and during sessions. These results do not suggest an absence of rhetoric acknowledgment of the fact that children have rights. Still it should be acknowledged that rights were not used as a fundamental principle in either of the studied courses.

Another suggestion has arisen from the findings from manuals and sessions. The possibility exists that a public health rule of preventive methods, based on adult norms, hinders the courses to forward children's liberty right and children as active agents in family relationships. Such a public health scenario would contradict the Swedish commitment to children's rights as a fundament for political interventions concerning children.

A possible future research area would be to investigate how children and parents regulate their joint family spaces in comparison with the parameters for family life that are outlined in today's parenting training programs. More knowledge about children's and parent's relational interactions can be gained by involving children themselves. Children's involvement could contribute to further development of policies for parent and child support. Health promotion as a theoretical base for parenting training interventions could also be further researched in relation to children's agency and participatory rights. Conclusively there still is a need for society to ethically and morally consider children's position and dependence on adult norms and wishes. Children's involvement in parenting training needs to be investigated and discussed as an ethical public health challenge for the future. As Wall claims, "rights have to be forged in social contexts that are always too narrow. The social whole has constantly to recreate itself" [62]. UNICEF has more cautiously described a way forward as "adults' evolving capacity and willingness to listen to and learn from their children (...) re-examine their own opinions and attitudes and to envisage solutions that address children's views [63].

Author details

Hetty Rooth^{1,2}

1 PhD Academy of Health Care and Social Welfare, externally affiliated to ChiP research group, Mälardalen University, Västerås, Sweden

2 ChiP Research Group, MDU, Sweden

*Address all correspondence to: hetty.rooth@protonmail.com

IntechOpen

© 2023 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. [CC] BY

References

- [1] UNICEF. UN Convention on the Rights of the Child. Geneva: UNICEF; 1989
- [2] Nolas SM. Children's participation, childhood publics and social change: A review. Children & Society. 2015;**29**:157-167
- [3] Sevón E. Who's got the power? Young children's power and agency in the childparent relationship. International Journal of Child, Youth and Family Studies. 2015;**6**(4.1):622-645
- [4] Government's official investigations. Strategy to Strengthen the Child's Rights in Sweden. Rpr 2009/10:232. Stockholm: Department of Social Affairs; 2010
- [5] Government's official investigations. Parental Support – A Win for All. National Strategy for Community Support and Assistance to Parents in their Parenting. SOU 2008:131. Stockholm: Department of Social Affairs; 2009
- [6] Smith R. Total perenting, Educational Theory. 2010;**3**:357-369
- [7] Vansieleghem N. The residual parent to come: On the need for parental expertise and advice. Educational Theory. 2010;**60**(3):341-355
- [8] WHO. Social Determinants of Health and Well-being Among Young People. Health Behaviour in School-Aged Children (HBSC). Geneva: International report from the 2009/2010 survey; 2011
- [9] Survey of Mental Health. Stockholm: Department of Social Affairs. R 2011:**9**;2011
- [10] The Statistics Database. Children's Relations Parents by Indicator, Accounting Group and Gender. Year

- 2014-2015 2018-2019. Stockholm: Central Bureau of Statistics (SCB); 2020
- [11] Public Health Agency of Sweden. How Municipalities, County Councils, Municipalities and Regional Associations and Ideally Organized Organizations Work with Parental Support Work. Östersund; 2014
- [12] Söderlind I, Sköld J, Malmberg B.Children and Young People TheirRole and Meaning Before and Now.Stockholm: The National Board of Health and Welfare; 2013
- [13] Pollard EL, Lee PD. Child wellbeing: A systematic review of the literature. Social Indicators Research. 2003;**61**(1):59-78
- [14] Statham J, Chase E. Child Well-being. A Brief Overview. Briefing Paper 1. Childhood Well-being Research Centre. Loughborough University. University of Kent; 2010
- [15] Cottm S, Espie J. DiscoursesUnderpinning Parenting TrainingProgrammes: Positioning and Power.Children & Society. 2014:25(6):465-477
- [16] Bremberg S, editor. New Tools for Parents – Proposals for New Forms of Parental Support. Stockholm: Swedish National Institute of Public Health; 2004. p. 49
- [17] Kiesler DJ. Beyond the Disease Model of Mental Disorders. Greenwood Publishing Group; 2000
- [18] Rooth H. All Parents in Focus. Governing Parents and Children in Universal Parenting Training. Mälardalen University: Academy of Health, Care and Social Welfare; 2018

- [19] Swedish Council on Health Technology Assessment and Assessment of Social Services. Programmes for Preventing Mental Illness in Children. A Systematic Literature Overview. Stockholm: Swedish Council on Health Technology Assessment and Assessment of Social Services; 2012
- [20] Webster-Stratton C. A randomized trial of two parenting training programmes for families with conduct disordered children. Journal of Consulting and Clinical Psychology. 1984;52:666-678
- [21] Costin J, Chambers S. Parent management training as a treatment for children with oppositional defiant disorder referred to a mental health clinic. Clinical Child Psychology and Psychiatry. 2007;12(4):511-524
- [22] Sanders MR, Markie-Dadds C, Tully LA, Bor W. The triple P-positive parenting program: A comparison of enhanced, standard, and self-directed behavioral family intervention for parents of children with early onset conduct problems. Journal of Consulting and Clinical Psychology. 2000;68(4):624-640
- [23] Moretti MM, Obsuth I, Craig SG, Bartolo T. An attachment-based intervention for parents of adolescents at risk: Mechanisms of change. Attachment and Human Development. 2015;**18**:1-17
- [24] Barlow J, Stewart-Brown S.
 Understanding parenting programmes
 Parents' views. Primary Health
 Care Research and Development.
 2000;2:117-130
- [25] Skinner BF. Science and Human Behavior. New York: Macmillan; 1954
- [26] Thorell LB. The Community Parent Education Program (COPE): Treatment effects in a clinical and a community-based sample. Clinical Child Psychology and Psychiatry. 2009;**14**:373-387

- [27] Bandura A. Social Learning Theory. Englewood Cliffs, N.J: Prentice Hall; 1977
- [28] Ainsworth MS. Infant-mother attachment. American Psychologist. 1979;**34**(10):932-993
- [29] Bretherton I. The origins of attachment theory John Bowlby and Mary Ainsworth. Developmental Psychology. 1992;28:759-775
- [30] Cassidy J, Shaver PR, editors. Handbook of Attachment: Theory, Research, and Clinical Applications. New York: Guilford Press; 2007
- [31] Lister R. Children (but not women) first: New Labour, child welfare and gender. Critical Social Policy. 2006;**26**(2):315-335
- [32] Furedi F. Paranoid Parenting: Why Ignoring the Experts May be Best for Your Child. London: Continuum; 2008
- [33] Cottam S, Espied J. Discourses underpinning parenting training programmes: Positioning and power. Children & Society. 2013. DOI: 10.1111/ chso.12011
- [34] Naidoo J, Wills J. Health Promotion: Foundations for Practice. 2nd ed. Edinburgh: Baillière Tindall; 2000
- [35] Swedish National Institute of Public Health. Implemented measures. Knowledge base for Children and young people 2013 The development of the factors that affect health and implemented measures. R;2013(05):2013
- [36] Sandin B. The parent: A cultural invention. The politics of parenting. European Journal of Developmental Psychology. 2017;14(6):733-746
- [37] Smith R. Parent education: Empowerment or control? Children & Society. 1997;**11**:108-116

- [38] Breslow L. From disease prevention to health promotion. Journal of the American Medical Association. 1999;**281**:1030-1033
- [39] O'Connor TG, Scott SBC. Parenting and Outcomes for Children. King's College London: Joseph Rowntree Foundation; 2007
- [40] WHO. The European Health Report 2015. Targets and Beyond – Reaching New Frontiers in Evidence. WHO Regional Office for Europe; 2015
- [41] Shaffer A, Beth A, Kotchick BA, Shannon Dorsey S, Forehand R. The past, present, and future of behavioral parent training: Interventions for child and adolescent problem behavior. The Behavior Analyst Today. 2001. DOI: 10.1037/h0099922
- [42] Reynaert D, Bouverne-de-Bie M, Vandevelde S. A review of children's rights literature since the adoption of the United Nations convention on the rights of the child. Childhood. 2009;**16**(5):18
- [43] Pupavac V. Misanthropy without borders: The international children's rights regime. Disasters. 2001;25(2):95-112
- [44] Department of Social Affairs. Child abuse To prevent and remedy. Final report from the committee against child abuse. SOU. 2001:72
- [45] Halldén G. Child perspective. An ideologically charged concept and imprecise as an analytical tool. Locus. 2009;**3-4**:20
- [46] Qvarsell B. Children's perspective and human rights. Maximal knowledge or building knowledge? Pedagogisk Forskning i Sverige. 2001;8(1-2):101-113
- [47] Johansson E. Approaching children's perspectives. Meetings of Researcher's

- and educator's meetings with children's perspectives. Pedagogisk Forskning i Sverige. 2003;8(1-2):42-57
- [48] Sommer D. Views of the child in developmental psychology. Is a child perspective possible? Pedagogisk Forskning i Sverige. 2003;8(1-2):85-100
- [49] Söderbäck M, Coyne I, Harder M. The importance of including both a child perspective and the child's perspective within health care settings to provide truly child-centred care. International Journal of Child Health Care. 2011;15(2):99-106
- [50] Prout A. Taking a step away from modernity: Reconsidering the new sociology of childhood. Global Studies of Childhood. 2011;**1**:4-14
- [51] Rooth H, Forinder U, Piuva K, Söderbäck M. Being a child in the family. Young children describe themselves and their parents when their parents have attended universal manual-based parenting courses. Journal of Family Studies. 2020. DOI: 10.1080/13229400.2020.1860112
- [52] Wall J. From childhood studies to Childism: Reconstructing the scholarly and social imaginations. Children's Geographies. 2019. DOI: 10.1080/ M733285.2019.1668912
- [53] Rooth H, Forinder U, Piuva K, Söderbäck M. An assessment of two parenting training manuals used in swedish parenting interventions. Children & Society. 2017;31(6):510-522
- [54] Rooth H, Piuva K, Forinder U, Söderbäck M. Competent parents with natural children: Parent and child identities in manual-based parenting courses in Sweden. Childhood. 2018;25(3):369-384
- [55] Lindberg L, Ulfsdotter M, Jalling C, Skärstrand E, Lalouni M,

Lönn Rhodin K, et al. The effects and costs of the universal parent group program – All children in focus: A study protocol for a randomized wait-list controlled trial. BMC Public Health. 2013;13:688

- [56] Wyatt Kaminski J, Valle LA, Filene JH, Boyle CL. A meta-analytic review of components associated with parent training program effectiveness. Journal of Abnormal Child Psychology. 2008;36(4):567-589
- [57] Widding U, Olsson J. A parental support based on the Children's Convention? An analysis of how parent support materials relate to different dimensions of the Child Convention. Tidskriftet Barn. 2014;1:41-56
- [58] Mazzoni V, Harcourt DS. An international experience of research with children: Moving forward on the idea of children's participation. Qualitative Research. 2013;14(2):252-268
- [59] Christensen P, James A, editors. Research with Children. New York: Falmer Press; 2008
- [60] Clark A, Moss P. Listening to Young Children: The Mosaic Approach. London: National Children's Bureau; 2011
- [61] Daly M, editor. Parenting in Contemporary Europe: A Positive Approach. Strasbourg: Council of Europe Publishing; 2007
- [62] Wall J. Ethics in the Light of Childhood. Washington DC: Georgetown University Press; 2010
- [63] UNICEF. Fact Sheet: The Right to Participation. 2011

Section 2 Migration and Child Rearing

Chapter 6

Communication in Times of Uncertainty and Loss: Refugee Parents in Interaction with Kindergartens and Schools

Anne Marit Vesteraas Danbolt

Abstract

Most refugee parents experience uncertainty and loss transitioning to a new society. In this situation, the family is ascribed an increased value, as a social capital and a point of reference in an unstable world. At the same time, the complexity of the new society creates new hierarchical structures that place the family in a vulnerable position. Educational institutions like kindergartens and schools play a major role in the lives of families, since their practices impact the wellbeing and prospects for the future of the children. This chapter explores how newly arrived refugee parents perceive their interactions with kindergartens and schools in their new context. Data is collected by observations in a group of refugee parents over one year and by individual interviews, using interpreters. The study shows that there are good intentions on both sides, but the complexity of the situation causes misunderstandings, thus creating new instances of uncertainty. There is, however, a good potential for improved communication and better understanding in the relationship between refugee parents and kindergartens and schools.

Keywords: refugee parents, education policy, initial literacy, diversity, communication

1. Introduction

In a globalised world, millions of families are on the move and settling in new surroundings. This migration often means major disturbances to family life as well as to the families' relationship with educational institutions like kindergartens and schools. Migration produces new hierarchical social positions, where parents must re-orient themselves and find new ways of coping with daily life. Families account for between one-fourth and one-half of the migrant population in the OECD countries [1], and forced migration due to war and conflict is part of this picture. Many refugees parents experience uncertainty and loss when transitioning to a new society. They have left behind family and friends, workplaces and social relations, in short, life as they knew it. Different societal norms and regulations may form barriers to integration and inclusion in a new community. There are thus good reasons for municipalities, local

71 IntechOpen

authorities as well as teachers and managements in kindergartens and schools to take into consideration how families experience challenges when they are settling into a new local community.

The migration-driven diversification of societies in the Western world has been described as super-diversity [2]. This entails new patterns of inequality and prejudice, racism and segregation ([2], p.2), and the term is widely used across disciplines. This concept is fruitful for shedding light on communication patterns that arise in the interaction between migrant parents and representatives of educational institutions. In such interactions, the uneven distribution of power is noticeable, but may all the same be concealed in subtle ways. Settling into a community that is quite unfamiliar as regards ways of behaving, socialising, interacting and communicating with institutions that have a bearing upon their life as a family, the family is placed in a vulnerable position. In such times of uncertainty, family values, as well as the members of the family itself, are ascribed an increased value, as a point of reference in an uncertain and unfamiliar world [3]. The children in the family represent in this situation a social capital [4], increased in symbolic value, since other forms of social capital in the form of network, education, employment as well as economic capital are diminished or lost. All matters relating to the welfare of their children are therefore of the utmost importance to the refugee parents [3].

This chapter will focus on how refugee parents perceive their relationship with kindergartens and schools in a new community. The research questions guiding this research are: How do refugee parents experience their interactions with educational institutions like kindergarten and schools? What perceptions do they have regarding language and literacy when it comes to their children's education?

2. Immigration and education policies for refugees in Norway

2.1 Settlement regulations for refugees

Norway is among the 145 countries that have recognised the Geneva Convention of 1951. This convention protects the rights of refugees [5, 6]. Immigration to Norway has increased during the last decades, and especially after the influx of people fleeing from the war in Syria, the number of refugees increased considerably. Many refugees have also arrived from countries in Eastern Africa and have settled in Norway. The pandemic caused a sharp decrease in immigration, but since the outbreak of the war in Ukraine in 2022, nearly 40,000 refugees have sought asylum in Norway. Still, the percentage of people with refugee background is low: while about 15 percent of the Norwegian population have immigrant background, only 4,5 percent have a refugee background [7].

Refugees may come either as asylum seekers, or they may be included as resettlement refugees in a quota agreed upon with the United Nations High Commission for refugees (UNHCR). They are then brought directly to Norway from refugee camps abroad [8]. Refugees are granted a temporary residence permit for one or up to three years. Family members may also be granted a family immigration permit based on certain criteria. It is normally not possible to apply for a permanent residence permit before three years' stay in the country. Those who are granted a temporary residence permit are settled in designated municipalities, based on a request to the municipality by the Norwegian Directorate for Integration and Diversity. The municipality receives economic support from the government based on the number of refugees received

by the municipality, and they must offer an introduction programme including these elements: Norwegian language classes, courses providing knowledge about the Norwegian society, and educational work-related activities. In this study, data was collected among refugee parents who took part in a course on parenting practices, based on the Incredible Years programme.

2.2 Education policy for immigrant children in primary and lower secondary schools

Education for pupils with another language background than Norwegian or Sami (which is an indigenous language in Norway) is regulated by Section 2.8 in the Education Act. This section states:

Pupils attending the primary and lower secondary school who have a mother tongue other than Norwegian or Sami have the right to adapted instruction in the Norwegian language until they are sufficiently proficient in Norwegian to follow the normal instruction of the school. If necessary, such pupils are also entitled to mother tongue instruction, bilingual subject teaching, or both [9].

The Education Act, Section 2.8, also states that the municipality must map the Norwegian language skills of the pupils before it is decided to give them adapted language education. This mapping must continue while they receive adapted Norwegian instruction, to assess whether the pupils are sufficiently skilled in Norwegian to follow the regular teaching in the school. The Norwegian Directorate for Education has developed curricula for adapted instruction in Norwegian and for mother tongue instruction and a tool for mapping of language development for second language learners of Norwegian [10].

It is the municipality, as the owner of the primary and lower secondary schools within their community, who is responsible for providing teaching for pupils who are second language learners of Norwegian. However, even though the Education Act states that mother tongue instruction is among the rights of a pupil who is a second language learner of Norwegian, mother tongue instruction is rarely used in Norwegian classrooms. In fact, even though the number of pupils with immigrant background has increased, the number of pupils receiving adapted instruction in Norwegian and mother tongue instruction has decreased [11]. This development is interesting, since the core curriculum for primary and secondary education states that all pupils "shall experience that being proficient in a number of languages is a resource" [12]. The same positive attitude to different languages is also mentioned in the Framework Plan for Kindergartens, which says that the kindergarten staff should help ensure that "linguistic diversity becomes an enrichment for the entire group of children and encourage multilingual children to use their mother tongue" [13]. The documents giving the mandate for kindergartens and schools thus promote a positive view on multilingualism and diversity, while the Education Act is open for interpretations and gives few directions for implementation.

2.3 Initial literacy and second language learning

International research recommends that initial literacy should be taught in a language that is familiar to the learner [14, 15]. Since reading is to make meaning of written language, the familiarity with the language is crucial. However, many learners

around the world receive education in an unfamiliar language [15]. This also applies to Norway, where second language learners of Norwegian rarely receive initial literacy instruction in their mother tongue [11]. As mentioned above, only those pupils who are considered not to be able to follow the regular instruction in the classroom are granted adapted instruction and mother tongue instruction or bilingual subject teaching only when it is considered "necessary". And when granted, normally only a restricted number of hours per week are allocated.

This means that pupils with another language background than Norwegian will have a different outset for their literacy development than their peers who are first language users of the language of instruction. They have a double learning task in the way that they must learn the language of instruction and the content of the curriculum at the same time. This is a demanding role and requires much support from those around the emergent bilingual pupil.

All children, irrespective of language background, have experiences with written language before they start school [16, 17]. When it comes to second language learners, who are emergent bilinguals, the literacy practices they encounter in their homes form a basis for the continued development in school. But if the literacy instruction rests on a monolingual bias, without any attention to other languages or other literacy practices than those of the language of instruction, the experiences emergent bilinguals bring to school will not be recognised. In such a situation, the second language learners' experiences will be invisible and an unused resource. To draw on the resources of the emergent bilinguals requires a partnership between the school and the home.

Trends in research on literacy and language learning open for more flexible ways of building emergent bilinguals' literacy. There are several examples of such partnerships where the school successfully builds bridges to the home and acknowledges the complexity of the situation for migrant families [17–19]. More flexible ways of drawing on the resources multilingual pupils bring to school can be a potential for a more positive development in the language learning process and for more confident multilingual practices in the family.

Many immigrant parents want their children to maintain the language(s) of their family, while they at the same time are dedicated to their children's learning of the language of instruction. Making visible the language resources of an emergent bilingual pupil also confirms the identity of the family [17, 18]. To open up for a view on multilingualism as a resource requires competence among teachers in kindergartens and schools and is ultimately the responsibility of the municipality.

3. Research design

The research design for this study was an exploratory qualitative case study [20]. Qualitative research seeks to explore and understand "the meaning individuals or groups ascribe to a social or human problem" [21], which describes well the aim of this research. Data was collected in a group of refugee parents who took part in an introduction course in a semi-rural municipality in Norway. Data consisted of observations in classes when the participants talked about their family, by semi-structured interviews individually and in pairs and by reflection notes collected in a meeting towards the end of the school year. The data was analysed by a thematic analytical approach, developing codes to look for emerging themes.

The group consisted of six mothers and three fathers, representing six different families. Three couples came from an Arab-speaking country, two mothers came from East African countries, and one single mother came from an Asian country. All participants except two were newly arrived in Norway with less than 8 months' stay. Their education varied between six and twelve years, whilst one had professional training.

Since all the participants were newly arrived in Norway, the interviews were conducted by help from interpreters. The topics for the interviews were: Experiences from their interactions with the teachers in their children's kindergarten or school, and their hopes and expectation for their children's future, including their education. The interviews were recorded and transcribed by the author. The participants were given pseudonyms to preserve anonymity, and all recordings were deleted after the transcription.

3.1 Limitations of the study

This study is an observation and interview study of refugee parents, where the main purpose was to bring forth the voices of the parents. The school management and the teachers have not been involved in the study. This means that statements about the schools' teaching practices come solely from the parents, and the intentions behind the choices the schools have made will remain unknown to the researcher. This must be taken into consideration when interpreting the data. Furthermore, the number of participants is low, so findings from this study cannot be generalised to the wider population. However, this study may give a deeper understanding of how refugee parents experience their interactions with institutions that are of vital importance for their children's well-being and future in the new country.

Another limitation is that all communication between the researcher and the participants has been conducted with interpreters as mediators. This means that nuances and even important information in the participants' statements may have been lost. However, to capture the perceptions from newly arrived refugee parents who have not yet learnt the Norwegian language, the use of interpreters is indisputably necessary. The interpreters were experienced and knew the Norwegian school system well, which was an asset given the topic of this study.

Last, but not least, it is a limitation to the study that the researcher was an ethnic Norwegian and as such a representative of the new society. Even though the researcher came from the outside and had no links to the parenting programme nor to the kindergartens and schools in the municipality, this fact may have led to restraints in criticism or misplaced praise. The longitudinal design of the study, with frequent meetings over one year, was intended to counteract this tendency and build trust between the partners, but it must nevertheless be considered in interpreting the responses from the parents.

3.2 Ethics approval and consent

Newly arrived refugee parents, not yet proficient in the majority language, must be considered a vulnerable group ([21], p. 95). Their situation must therefore be taken into account when doing research of this kind that will involve talking about private matters like family and children. This topic could also increase the parents' sense of loss and cause difficult feelings. Ethics approval was obtained from the Norwegian Centre for Research Data (NSD), including approval of interview guides and information letters. The interpreters read the information letter to the participants in a

meeting, and the researcher was available to answer questions. It was made clear that participation in the research was voluntary and that the parents could withdraw from the study at any time, without any repercussions for themselves. The participants signed the consent form in the first meeting. No one wanted to withdraw during the period of data collection or afterwards. On the contrary, it seemed that they appreciated having the opportunity to talk about their children and to vent issues regarding their children's schooling.

4. Results and discussion

Three themes emerged from the coding of the data: Communication in a mode of friendliness and trust; the complexity of the language policy; the value of education.

4.1 Communication in a mode of friendliness and trust

The parents all expressed appreciation of their children's teachers, some more explicitly that others. Several of the families came from refugee camps in Lebanon, where the children were assigned to classes in the afternoon, after the resident children had left. This experience of a non-inclusive practice seemed to have left its mark on one of the parents, who saw the Norwegian school system in a stark contrast: "It is first of all a good and nice system, and I am so grateful there is no discrimination nor racist attitudes. My children can attend school in the morning and not in the afternoon, like in Lebanon". Given the huge pressure on the education system in Syria's neighbouring countries, the comparison may seem quite unfair; however, the experience of inclusion of their children in the mainstream classroom obviously meant a lot. The parents also mentioned that the teachers did not allow bullying, and that instances of unacceptable behaviour were sanctioned quickly. Were sanctioned quickly. The teachers were described as "humane" and friendly towards the children.

The parents conveyed appreciation of this friendliness and positive interest from the teachers. Some of the teachers obviously tried to draw on the linguistic diversity in the classroom. One of the mothers described her daughter's teacher in this way: "For example Sylvia (pseudonym), who is the teacher in my daughter's class, I love her so much. She has learnt a lot of words, like 'good morning' and 'good day' and such things, she has learnt these words in all the languages of the children". The parents also mentioned teachers in their own language classes for adults who showed interest and a positive attitude towards their mother tongues.

Schools are required to have individual meetings with the parents of each pupil twice a year. These meetings focus on the child's individual progress in all areas, as well as the child's social adaptation to other pupils and to the school as a whole. The parents told that in these meetings, an interpreter was always present and available for them. The parents appreciated this practice. One couple mentioned, however, that in meetings with the whole class (parents' meeting), which are not mandatory, there were no interpreters available. When asked what he felt about being in such a situation, one father said: "It was like being inside a black bag". This lack of opportunity to talk with the other parents made him feel left out of the group.

Through the parents' descriptions, the teachers put much emphasis on friendly relations with the pupils and the parents. In the mandatory individual meeting with the parents twice a year, the school provides interpretation for the parents. But this

does not take place in social arrangements and meetings that are not mandatory, which indicates that there are limits to the inclusiveness. The friendliness and inclusive attitude seem to be a widespread but individual practice that was not yet fully institutionalised.

4.2 The complexity of the language policy

The parents were asked whether their children received any mother tongue instruction, cf. 2.2 above. One of the mothers responded that there was an Arabic teacher in the school and that there were four pupils in his class, but her daughter was not among those who were assigned to his class. She said that she had asked why her daughter was not included, but she did not receive an answer, or at least not an answer that she could make sense of. Her interpretation of the situation was that her daughter first should focus on learning Norwegian, and then, she could have classes in Arabic. This misconception was corrected by the interpreter, who asked if he could explain to this mother. He told her that the case was that her daughter was considered too proficient in Norwegian to receive mother tongue instruction. This was obviously new information to the mother, so the communication from the school seemed not to have fulfilled its purpose.

A report from another mother demonstrated that the mother tongue instruction was not always given under conducive circumstances. Her son received some help in the mother tongue from an interpreter who explained the teaching to him in class. He was placed at the back of the classroom, probably to avoid disturbances. Her son had told her that the class teacher sometimes said "Please don't disturb" to the interpreter, since the help was given in the same classroom. The mother was obviously frustrated by this practice. "How can my boy learn if the teacher says, 'Please don't disturb'? And he is the weakest, and he is placed in the back of the classroom!" The mother also wondered why her son was put at the back of the classroom the whole day, while the interpreter was there with him only for a few hours. She had tried to ask the school about it but did not receive an answer that made this practice clear to her.

Since the provision of mother tongue instruction was relatively scarce, some of the parents tried to teach their children themselves. One of the mothers described how she tried to support her child at home: "I try to be the teacher myself, but it is hard. I have to learn Norwegian myself". It is time-consuming to learn a new language, and the parents' capacity is naturally stretched, when they try to do two quite difficult tasks at the same time.

The school seemed to strive for an inclusive practice and had allocated mother tongue instruction to some of the newly arrived second language learners. However, when it came to the realisation of this practice, there were obviously pitfalls. For instance, it was not clear whether the pupils actually met a qualified mother tongue teacher, and the use of the word "interpreter" indicates that the school might have opted for mother tongue support rather than mother tongue instruction or bilingual subject teaching. In this regard, the limitations of this study must be taken into account. It is also a fact that many schools struggle to find qualified mother tongue teachers. However, it seems quite clear that communicating the specific rights of pupils with another language than Norwegian and Sami, laid down in the language policy, seemed to be a challenge. Considering the somewhat complicated regulations and the rather unclear indicators of what is "sufficiently proficient", it is not surprising that the school seemed to have a difficult task in explaining these features of the language policy to newly arrived refugee parents.

4.3 The value of education

The parents were overall very proud of their children's progress in learning to speak the new language. They expressed no concern when it came to their children's progress in learning spoken Norwegian. However, when it came to basic literacy skills, like reading and writing, some of them expressed worries, like this mother, speaking about her 7-year old son: "He can speak Norwegian, but he struggles with reading and writing". When they were asked whether they had been given any advice as to how they could support their children's literacy development, one mother said that the teacher had asked them if they had audio books or a laptop or a computer. This made her feel uncomfortable, since she herself had very little experience with digital tools. In the meetings, the parents told that they had been strongly encouraged to visit the library and borrow books to read to their children. One mother confirmed that there were books in many languages in the school library, and her daughter had borrowed a book with Arabic letters. The book turned out to be written in Farsi, a language they did not know, and the mother laughed at the mix-up. But she clearly expressed her willingness to read to her daughter, if only the school could provide her with books an Arabic.

The parents put strong emphasis on the value of education in more general terms. One of the fathers explained in this way how he encouraged his children to work hard in school: "You must read, you must learn, you must just go on and work hard, we expect that you are doing your best". And he continued: "We tell them that being able to read and write is equal to seeing". In the final meeting in the group, the parents were asked to express their hopes for their children, and one of them wrote: "I want my children to have a good education and find a job, but first and foremost I want them to have a good and happy life in a peaceful society".

By the parents' accounts, there is an awareness among teachers in kindergartens and schools of the value of other languages than Norwegian. The encouragement the parents receive to visit the library and read books to their children points towards an interest in the language resources in the family. The school had also acquired books in many languages in the library, to be available for the newly arrived pupils. In the conversations about the children's education, there was a clear concurrence of values between the parents and the school.

5. Conclusion

Refugee parents meet many challenges when transitioning to a new society. The transition is mostly involuntary and includes many experiences of loss. They have nevertheless proved great strength in enduring difficult circumstances and mastered the transition. In their new community, they want the best possible future for the children and show positive attitudes towards their children's teachers. However, they meet many expectations and regulations related to kindergartens and schools, which implies adjusting to new social complexities [2].

Findings in this study show that parents expressed trust in the teachers in the kindergartens and schools and experienced a warm and friendly attitude in their communication with the teachers. They encouraged their children's education and wanted to support the school. The concurrence of values as regards education can have a great potential in building a strong relationship between the home and the school.

However, there are challenges to the relationship. The communication gaps in explaining the language policy as well as the lack of a systematic support to the

Communication in Times of Uncertainty and Loss: Refugee Parents in Interaction... DOI: http://dx.doi.org/10.5772/intechopen.110687

children's multilingual development put strain on the family. The parents felt that it was hard for them to really contribute, since they were unfamiliar with the language of instruction, and some of them did not have much education themselves. But there are bright spots also to this picture. Some of the teachers practice an openness to the linguistic diversity in the class, and the schools have multilingual books in the library and encourage pupils and parents to borrow books. By the parents' report, the school seems to be interested in the parents' literacy practices, though this is probably not communicated in the best possible way to the parents.

Communication with refugee parents can be difficult for many reasons [17]. These difficulties can be mitigated by good preparation of the kindergarten and school staff, including capacity building, and to establish educational partnerships with parents.

In communication with refugee parents, it is important to be aware of the special circumstances surrounding their life in the new country. The underlying imbalance of power requires a sensitive approach in the interactions. Good intentions are important, but not sufficient in securing full understanding of the educational offer to second language learners of Norwegian, given the complexities of the language policy. Empowering the parents by acknowledging their resources and finding ways of how they can support their children will build their partnership with the educational institutions and give better prospects for their hopes and dreams for their children in a new society.

Conflict of interest

The author declares no conflict of interest.

Author details

Anne Marit Vesteraas Danbolt Inland Norway University of Applied Sciences, Elverum, Norway

*Address all correspondence to: anne.danbolt@inn.no

IntechOpen

© 2023 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. CCD BY

References

- [1] Organisation for Economic Co-Operation and Development (OECD). A portrait of family migration in OECD countries. In: OECD. International Migration Outlook 2017. p. 107-166. DOI:10.1787/ migr_outlook-2017-en.
- [2] Vertovec S. Talking around super-diversity. Ethnic and Racial Studies. 2017;**42**(1):125-139. DOI: 10.1080/01419870.2017,1406128
- [3] Danbolt AMV. Cultural responsiveness in the incredible years parenting programme for refugees: A case study. International Journal of Child Care and Education Policy. 2020;**14**(6):1-19. DOI: 10.1186/s40723-020-00071-5
- [4] Bourdieu P. The forms of capital. In: Halsey A, Lauder H, Brown P, Wells A, editors. Education, Culture, Economy, and Society. Oxford: Oxford University Press; 1997. pp. 46-58
- [5] European Commission. Common European Asylum System. n.d. Available from: https://home-affairs.ec.europa. eu/policies/migration-and-asylum/ common-european-asylum-system_en
- [6] United Nations High Commission for Refugees. The 1951 Refugee Convention. Available from: https://www.unhcr. org/1951-refugee-convention.html
- [7] Statistics Norway (SSB). Persons with Refugee Background. 2022. Available from: Personer med flyktningbakgrunn (ssb.no)
- [8] Norwegian Directorate of Immigration. Resettlement Refugees (quota refugees). n.d. Available from: Resettlement refugees (quota refugees) -UDI

- [9] The Education Act. Available from: https://lovdata.no/dokument/NLE/lov/1998-07-17
- [10] Norwegian Directorate for Education. Mapping Tool for Basic Norwegian. Available from: https://www. udir.no/laring-og-trivsel/lareplanverket/ fagspesifikk-stotte/kartleggingsverktoyi-grunnleggende-norsk/
- [11] Key Numbers for the Education Sector. Norwegian Association of Teachers. 2021. Available from: Nøkkeltall for grunnskolen t.o.m. skoleåret 2020/21 (utdanningsfor bundet.no)
- [12] Norwegian Directorate for Education. Core Curriculum. Identity and Cultural Diversity. Available from: https://www.udir.no/lk20/overordnetdel/opplaringens-verdigrunnlag/1.2identitet-og-kulturelt-mangfold/
- [13] Norwegian Directorate for Education. Framework Plan for Kindergarten. Available from: https:// www.udir.no/laring-og-trivsel/ rammeplan-for-barnehagen/
- [14] Snow CE, Burns MS, Griffin P, editors. Preventing Reading Difficulties in Young Children. Washington, DC: National Academy Press; 1998. p. 432. DOI: 10.1002/pits.10011
- [15] UNESCO. Policy Paper 24, 2016. If You Don't Understand, How Can You Learn? Available from: https://en.unesco.org/gem-report/if-youdon%E2%80%99t-understand-how-can-you-learn
- [16] Heath SB. Ways with Words: Language, Life, and Work in Communities and Classrooms.

Communication in Times of Uncertainty and Loss: Refugee Parents in Interaction... DOI: http://dx.doi.org/10.5772/intechopen.110687

Cambridge: Cambridge University Press; 1983

[17] Moll LC, Amanti C, Neff D, Gonzalez N. Funds of knowledge for teaching: Using a qualitative approach to connect homes and classrooms. Theory Into Practice. 1992;**XXXI**(2):132-141. DOI: 10.1080/00405849209543534

[18] Cummins J, Early M. Identity texts. In: The Collaborative Creation of Power in Multilingual Schools. Stoke on Trent: Trentham Books; 2011

[19] Ragnarsdóttir H. Multilingual childhoods of refugee children in Icelandic preschools: Educational practices and partnerships with parents. European Early Childhood Education Research Journal. 2021;29(3):410-423. DOI: 10.1080/1350293X.2021.1928719

[20] Yin R. Case Study Research. Design and Methods. 5th ed. Thousand Oaks, CA: Sage; 2014

[21] Creswell JW. Research Design: Qualitative, Quantitative, and Mixed Methods Approaches. 4th ed. Los Angeles: Sage; 2014

Chapter 7

Mitigating the Unmet Psychosocial Support Needs of Left-behind Learners in Child-Headed Households: Exploring 'Educare' Support at a Zimbabwean School

Ricanos Jaure and Alfred Henry Makura

Abstract

Following parental labor migration, the family unit is disrupted contributing to the formation of a unique form of child-headed households (CHH) among the left-behind children (LBC). Parental migration creates a gap in terms of children support, care, and protection leading to unmet psychosocial support needs among the LBCs in child-headed households. This research chapter sought to explore the fusion of education and care (Educare) in schools as a holistic system of support for the left-behind at a school where LBC expend most of their time. This elevates and transforms the school from just being centers for education excellency but also centers for care and learner well-being. The chapter draws from qualitative data in which twelve left-behind learners and ten education and community informants participated in a focus group discussion. Findings revealed that the school has great potential to support learners in adverse living and learning situations through the curriculum, extra-curriculum clubs, peer groups, and school personnel (teachers in loco parentis). However, the same school environment was found to be associated with negative typecasting of LBC's that sometimes affects the support system. This chapter recommends building the support base for LBC within school through sensitizations, capacitation, and taping support from community resources.

Keywords: Educare, left-behind learners, psychosocial, learning, child-headed households, Zimbabwe

1. Introduction

"Schools are not just place where we learn facts and numbers but also place where we learn how to live" Dewey (1899).

The vision shared by John Dewey in the nineteenth and twentiethth centuries about the school is still relevant today, more so, when world order and, in particular, parenting practices are under threat from global trends and most recently the

83 IntechOpen

COVID- 19 pandemic. Globalization and the economic crises have triggered labor migration trends in Africa and part of the developing world [1, 2]. Adults, who are also parents in the developing world, take the option to migrate to local and international destinations seeking employment opportunities. This process has significantly transformed, destabilized, and reconfigured the traditional parenting and family structure [3]. Following parental labor migration, the key tenets that define parenting such as co-residence, care, and guidance are sometimes delegated or neglected, thus exposing the left-behind learners to unmet psychosocial needs. The parent is made to perform parenting duties from a distance, which limits them in terms of carrying out the expected interaction, care, and guidance responsibilities [4]. In some cases, the migrating parents were also incapacitated in terms of providing for the needs of LBC's due to challenges associated with the then prevailing COVID-19 restrictions. It is in view of these complications that this book chapter sought to explore the potential of the school in supplanting parental roles and become not only a center of education but also care and learner well-being (Educare).

This research paper acknowledges extensive research on left-behind children and child-headed households [3–5]. There is, however, limited scholarly attention when the two components are combined. The combination of LBC and CHH settings could be described as an abysmal combination as regards the psychosocial functioning and educational outcomes. This book chapter sought to expose the potential of the school as a stopgap measure to address the unmet psychosocial needs of the LBC's. The utilitarian definition and understanding of the school as a center for life lessons and pillar for care for learners is used. A more holistic understanding of education is used in which performance is not only measured by performance in traditional subjects such as Mathematics, languages, or Sciences but life lessons. Schools become centers for both education and care (Educare) in which issues to do with learner well-being are prioritized.

2. Background

Parental labor migration has been part of the history of many of the developing world countries, and in Africa, traced back to the days of colonization. This has been accepted as a means for survival and millions of adults migrate largely from rural to urban areas in search of job and life opportunities. Driven by globalization, social and economic challenges migration trends in Zimbabwe, for instance, have taken an international flare [3, 6]. The search for life opportunities is no longer limited by geographical and national boundaries, but parents move to where opportunities are. Parents in Zimbabwe have found settlement in countries such as Australia, United Arab Emirates, the United Kingdom, Qatar, and China for employment. In Africa, South Africa hosts the bulk of the labor migrants, (up to 3.5 million) while others have moved to countries such as Botswana, Mozambique, Zambia, Namibia, and among others [6].

The migration of a parent or parents is often met with lots of joy and anticipation [1]. In Zimbabwe, this is usually driven by the hope for a better life provided by the migrating parent. Lots of sacrifices are made by the family in the attempt to meet the migration cost for the parent. In some cases, some families disposed of their family houses to pay for the air tickets and the visa requirements. The hope is that all that is sold to finance the migration processes will be replaced when the parent settles, especially in the global north. The settling of the parent in the global north marks the onset of "survival circuits" characterized by remittances that keep the left-behind children surviving the economic challenges associated with the global south [2].

The remittances are not only significant for the family but they also contribute to the economy of the global north countries such as Zimbabwe [2].

Out of the more than 15 million people in Zimbabwe close to 7.7 million are children [7]. The children constitute close to half the population of Zimbabwe meaning they are important in terms of future planning. They need the proper upbringing so that they would be able to take up leading roles in the future. The UNHCR defines a child as someone immature below the age of 18. The immaturity may imply that the child is unable to feed and care for himself/herself. Thus, when the adult parent is not present in the life of the children there is indeed a chance that the children would be adversely affected in terms of their psychosocial functioning.

While the community makes attempts to address the plight of vulnerable children in CHH, the kind of support is described as not cohesive and comprehensive [8]. Support that is ad hoc fails to meet the psychosocial support needs of the left-behind learners. This understanding makes institutions such as the school significant. This study proposes to utilize the school as a center in the bulk, which support for children in adverse living and learning conditions could be coordinated. Outcomes, such as well-being and learner conduct, would be imparted to learners through both the curriculum and co-curriculum in school setups.

3. Left-behind learners in child headed household settings

As parents migrate for labor, several restricting factors to do with immigration and the destination country sometimes makes it difficult for parents to migrate with their children [9]. Traditionally, when parents migrate, children were left in the care of the spouse or the extended family such as maternal or paternal grandparents were asked to assist. However, the breakdown of the extended family network and the prevalence of single parenting made this option difficult to make. Conflict over remittances between the migrating parents and the extended family is cited as one of the major factors leading to children being left in CHH [3]. Conflict usually arises because of remittances; some of the extended family members would feel they receive inadequate remittances, while the migrating parents would feel they are giving more. The result is some children considered "old enough" are left behind in their home country in child only families when parents migrate [10]. It is against this background that in Zimbabwe, there is a new growing population of children in CHH not because they are orphans but because the parents have migrated for labor. Left-behind learners (LBC) refers to children who are left behind for a period exceeding 6 months when parents migrate either to the cities or outside the country for work [11]. Following the breakdown of extended family networks and prominence of the nucleus family, many migrating parents end up leaving children in CHH.

In the developing world, children are more likely to be left in CHH because of single parenthood [12]. The number of female-headed households, in particular, is said to have increased dramatically driven by factors such as divorce and death of spouse [12]. This outcome eliminates the possibility of leaving the children with the other spouse when migrating. Females have also joined the bandwagon of labor migrants from Zimbabwe, and when they are single, they sometimes leave children in child only families. Zimbabwe, for example, has up to 40.6% female-headed house-holds [13]. The feminization of labor particularly in the care field has triggered a massive exodus of women contributing to children being left behind in CHH. This to some extent contributes to the creation of this unique form of CHH [14].

Certain factors are considered by the migrating parents [5] before leaving their children in CHH. Such factors are primarily social and economic (gender, age, accommodation, and security, etc.). Jaure and Makura [1] assert that the gender of the LBC's is a critical factor that the migrating parent takes into consideration. In a study at a school in Zimbabwe and South Africa, Jaure, [1] for instance, observed that boys are more likely to be left in CHH when compared to girls. The possible reason is that they feel boys are less vulnerable as compared to girls [5, 12]. The reason may also be motivated by the general perception that regards sexual abuse as graver as compared to other forms of abuse. The idea that girls more than boys are known victims of abuse, justify this line of thinking.

The age and level of maturity of the LBC's is a factor [5] for consideration in determining the negative effects associated with parental migration. In a study conducted among LBC's in Zimbabwe, most of the LBC's are relatively mature, above the age of 15, at adolescence stage [10]. Nevertheless, the adolescence stage of development has noted developmental challenges that become compounded by lack of parental guidance. A study, in China, also affirms that the age and gender of the LBC were key predictors of negative psychological effects such as anxiety and depression [13]. Left behind girls tend to be affected by parental migration more than boys because in accordance to African culture, they are made to shoulder the household duties [7–24]. In that regard, girls are said to be affected more by parental migration as compared to boys.

4. Unmet psychosocial support needs

As parents migrate, LBC find themselves experiencing unmet psychosocial support needs that in turn affect their learning outcomes. The LBC attending school becomes left-behind learners in the context of a school. Left-behind learners find themselves with unparalleled psychosocial challenges attributable to lack of parental supervision, set boundaries, love, guidance, and support [16]. This elucidates the significance of the adult parent in the home. Literature to date associates such learners with social and emotional challenges at home and at school making them a vulnerable group [1]. Findings drawn from several studies on LBC indicate that such children are associated with challenges such as hyperactivity, emotional, and peer problems [5, 17, 18]. This implies that the children fail to function properly and fail to relate with peers and siblings due to parental absence.

The lack of support systems contributes toward psychological problems that affect mental health including depression, poor psychological well-being, and health problems [19]. This is further elucidated by comparative analysis on the LBC compared to children living with their parents. The comparisons reveal that LBC's fare badly in terms of mental health as compared to those children who reside with their parents [20].

LBC's have also been associated with negative outside behaviors that are attributed to the absence of their parents. Separating internalizing from externalizing effects of parental migration is complicated because they are interdependent [10]. It is usually the inside effects such as loneliness, depression, and anxiety that lead to negative outside behaviors, for example, conduct challenges. The social and emotional maladjustments significantly affect their outward behavior. The outside behaviors noted in several studies on LBC include violent, aggressive behavior, drunkenness, and risky sexual behaviors [21–23]. Such behavior and conduct problems become significant learning problems that affect educational outcomes. In Zimbabwe, the LBCs are

associated with wild parties commonly known as *Vuzu* parties in which they abuse drugs, alcohol, and experiment with sex [10]. Children with both parents as migrants were noted as having lower scores for psychological well-being and higher scores for emotional symptoms and literacy problems when they are compared to conventional families [14]. This implies that parental care is an important factor that can influence well-being. The absence of the parent can trigger unmet psychosocial support needs that manifest through internalizing- and externalizing behaviors.

5. The unique form of CHH

LBC in child-headed households are a recent development that partly owes their development to the demise of the extended family system [1–4, 6–13, 15–24]. This unique form of CHH has with its similarities and differences with the traditional CHH. Similarities and differences are noted in the composition and effects on children. A close analysis of the similarities and differences is significant in the process of trying to expose the unmet psychosocial support needs of the LBC's.

The traditional definition of a child-headed household carries with it connotations of orphans, poverty, or misery [8]. Discourse on CHH refers to children whose parents would have died, and the children live and fend for themselves [24]. Fending for themselves in this case implies that they provide their own meals, with no adult to care for them. The immediate family and the alternate carers who are usually drawn from the extended family are not available to provide care [8]. A qualitative study, conducted in Zimbabwe by Maushe and Mugumbate, [8] reveals that children who live and care for themselves have challenges with basic provisions such as food education, and clothing. There is also no adult to guide, love, and supervise the children. This, in turn, explains the poverty that the children are often associated with. Because of parental death the children left in child only homes become destitute and experience challenges in terms of school fees payment and are deprived of basic amenities [8].

Parental migration has redefined the child-headed household family setup. Firstly, the parents or parents are very much alive but because of labor migration, they reside separately from their children [10]. The parent can elect to continue playing a significant parental role in the lives of their left-behind children. In that regard, the parent is unable to perform parental duties full-time. When parents migrate, there is so much anticipation on the part of the left-behind children and other relatives. The idea that the parent would have migrated to better economies is the major reason for hope. The children sometimes become the envy of the community as the parent can provide the basics and luxuries. The migrating parents are said to make up for their absence by providing even more than what the children require. The migrating parent can also continue to interact with the LBC's through internet or mobile communication tools. This to some extent retains a level of intimacy between the LBC and the migrating parent.

Owing to parental labor migration, parenting is transformed from being a full-time responsibility to being done remotely or part-time. Research indicates that in countries such as Zimbabwe and South Africa, labor migration is also responsible for the creation of this unique form of child-headed household [10].

LBCs represent parenting reconstituted and reconfigured. The traditional model of parenting and family model is represented by cohabitation and proximity [16]. The family resides under the same roof and shares meals. In the family setup, there are set

responsibilities for the parents and for children. However, when parents migrate, the parental roles are sometimes performed at a distance. The parent could still provide for the children through remittances. The parent can also share love over the phone or other social media platforms. In some cases, the eldest child would assume some of the parenting roles for younger siblings [10].

6. Traversing the COVID-19 pandemic complications/challenges

Circumstances following the declaration of non-governmental pandemic have unsettled and disrupted order among this growing population of left-behind learners in Zimbabwe. The negative impacts are even more pronounced among those left behind in child-headed households (CHH). Such children are made to do without parental social and emotional support, parental guidance, parental love, and monitoring over prolonged periods of time. These ensuing environments are perceived to disadvantage them in terms of their psychosocial functioning and educational proficiency [14]. The plight of such learners is further compounded by the COVID-19 pandemic that to some extent, have crippled the traditional sources of their support, that is, return visits and remittances. Cross-border mobility became limited and made it difficult for the parent to make return visits [4]. This further prolonged the period of separation between the migrating parent and the LBC's.

A survey conducted in South Africa during the national lockdown induced by the COVID-19 pandemic, reveals that two out of five adults lost their source of income and ran out of money to buy food [25]. South Africa is of interest in this study because they house the bulk of the labor migrants from Zimbabwe [6]. When the migrating parent is incapacitated, it would also follow that remittances would be limited. Thus, children left behind in their homes are equally affected by what happens in the host country. The COVID-19-induced lockdowns affected the well-being of LBC's.

This makes it prudent to explore other schemes of support that would improve the psychosocial functioning and prevent or modify conduct problems among left-behind learners. The goal is to ensure that left-behind learners and other learners in adverse living and learning conditions learn how to 'live.' In this case, 'live' entails coping and striving for improved well-being in the face of challenges associated with parental absence due to migration. The goal is to promote the acquisition of requisite social and emotional competencies (of LBC) for improved well-being and improved learning outcomes.

6.1 Methodology

This study followed a qualitative route and had the following objective (i) explore schemes of psychosocial support for LBC's in the context of the school. Focus group discussions were held with purposively selected participants drawn from one secondary school in the Midlands province of Zimbabwe. The participants were seven left-behind learners (n=7) and eight education and community informants (n=8). Separate sessions were held for learners and adults (n=7+8). The education informants were teachers, the school head, a senior teacher, and a school psychological services representative. The community contingent comprised a representative of the police community and liaison office and a representative of local nongovernmental organizations. To contextualize the envisaged support,

participants were asked to outline psychosocial challenges at school and at home. Thereafter, participants were asked to identify support systems within the school that could be utilized in building psychosocial support or mechanisms for the LBC's. Permission for the study followed an ethical clearance from the Central University of Technology (FRIC 21.18/2), permission was also granted by the Zimbabwe Ministry of Primary and Secondary Education. Written consent forms were completed by the adult participants. Participants' names were coded to protect their privacy. The discussions were recorded on a digital recorder by the main researcher and later transcribed verbatim. This was to ensure an accurate record of the proceedings. Data analysis followed the thematic approach (Thematic analysis) in which codes and themes were used. The collected information was coded and categorized after it was transcribed. Coding entailed systematically analyzing the transcribed data and disaggregating it into similar constituent parts (code clustering of data sets). After coding, data were organized or categorized into themes. Each theme entailed a unique aspect, which we elaborate in the results section. The analyses were used to arrive at conclusions, notwithstanding the goal of the study and the study's research objectives [26].

7. Results

Findings from this study revealed the different school-based avenues that have potential to be optimized for the support of left-behind learners affected by parental migration. Participants in the focus group discussion saw the essence of psychosocial support for left-behind learners in view of the unmet psychosocial support needs attributed to lack of parental support, monitoring, and discipline hand. By inference, this envisaged support is considered relevant in the face of threats posed by the COVID-19 pandemic. Participants saw potential support as being housed in the curriculum, extra curriculum, school leadership, school culture, and in teacher-learner relationships.

Teacher participants acknowledged the significant role that is shouldered by the school in support of learners in adverse learning conditions. Teacher participant TR3 "We are teaching students from diverse backgrounds, some are orphans, some have parents in the diaspora and some are in poverty meaning to say now more than ever, as teachers we need to be innovative. The school is their hope, and the community expects the school to step in and make a difference." The school, in this case, is made up of diverse learners with diverse challenges. What stakeholders in the school expect is the transformation of the institute through innovations so that the diverse learners are accommodated and supported.

Participant TR1, who is an experienced teacher remarked, "So much is expected from the school in today's world, the teacher must play different roles depending on circumstances. Sometimes you are the father, mother, teacher, or even aunt, learners come to class with many different problems, and as a teacher, they look up to you for solutions." This implies that the teacher sometimes must play caring, teaching, and advisory roles among the learners. In playing the different roles the school as an institution goes through a transformation. The teacher is expected to be innovative and come up with solutions that would facilitate teaching and learning. Such an understanding of the teacher's role is significant in exploring the utilitarian role of the school in the support of learners. The key roles expected of parents, such as guidance and protection, are assumed by the teacher.

7.1 Curriculum-based support

Participants revealed the critical role that the curriculum play in the support schemes of learners disadvantaged by adverse living conditions such as left-behind learners. The school curriculum in Zimbabwe has undergone several changes possibly to be apt to the transformation of society. More recently (2021) the Zimbabwe Ministry of Primary and Secondary Education launched the updated national curriculum. Participants also saw great potential in the support of left-behind learners within updated school curriculum. This was particularly pronounced in the guidance and counseling (G&C) course, and generally in crosscutting themes in all the learning areas. In this regard, a school head participant (SH) shared, "the new curriculum gives hope to disadvantaged learners because of its emphasis on guidance and counseling. True, left-behind learners need guidance and protection. If not, some end up abusing drugs and alcohol. We have several cases of learner dropouts because of pregnancy among these children of diaspora parents. This, G&C course, gives a fighting chance for left-behind learners because of its emphasis on life skills. Learners acquire decision-making skills, problem-solving skills among other skills." In other words, the school becomes a place where learners learn how to live according to Dewey [27].

Teacher participant TR2 also shared, "Indeed the G&C course is an important component of the curriculum, the only problem is that teachers fail to take it seriously and so learners fail to benefit much from its teaching. My suggestion is the Ministry need to train at least one full-time teacher." The learning area would teach and tackle real-life challenges and promote well-being among the learners.

"Cross-cutting issues within the updated curriculum can also help in resolving behavior problems among learners negatively affected by labor migration" (TR1). Cross-cutting themes in most of the learning areas in the syllabus include teaching on human rights, sexuality, Ubuntu, good citizenship, and social responsibility. Such themes when inculcated among learners would go a long way in assisting with coping strategies when challenges arise. Ubuntu is a guiding principle and an African philosophy that lays emphasis on humanity to others ('I am because you are'). The word Ubuntu comes from Xhosa and Zulu languages of Southern Africa. This social justice philosophy emphasizes love, generosity, collectivism, sharing of resources for the common good, and respect for one another. It is essentially a guiding principle in the realms of societal leadership, governance, and leadership. Fundamentally, it is premised on the notions of kindness, compassion, and individuality. The curriculum becomes a tool for capacitation, empowerment, and inculcation of Ubuntu principles. Learners, who are educated on human rights and sexuality, are more likely to protect themselves from issues such as abuse. The Ubuntu philosophy can also foster collaboration, respect, and social responsibility among the learners. It is through such courses that LBC's are empowered to deal with their unmet psychosocial support needs.

7.2 Teachers in the support schemes

The teachers, who deliver the curriculum, were also revealed as a special group that could be utilized in the psychosocial support schemes of left-behind learners. Teacher support was suggested to be through lesson delivery, initiating a conducive learning environment, and through fostering positive relationships with learners.

Teachers have capacity to introduce classroom management philosophies that foster positive behavior among learners. Participants brought to perspective the concept of loco parentis. According to SH, "teachers at college level are trained on the concept of loco parentis. This means that they are the local parent and can step into the shoes of the parent." This view suggests that the teacher is sometimes expected to play parental roles that may include emotional and physical support for learners going through challenges. Teachers in this regard take a leading role in educating the whole child as they do not concentrate not only on academic issues but also on the socioemotional development of the learner.

7.3 Positive school environment

Participants shared that when the school environment is positive, it may occupy the young minds of learners thereby minimizing on negative behaviors. In that regard, the school leadership plays a major role in cultivating a positive school environment. School psychological services representative (SPS1) shared, "...every school has its tone, the school administrators are supposed to develop the correct school tone so that learners from diverse backgrounds would be comfortable at the school."

To achieve that end, participants, for example, TR1, TR2, SPS1, and SH1 identified key virtues that need to be nurtured in creating a positive school environment, for example, empathy, respect, working together, and discipline. In such a school environment, learners negatively affected by parental migration could easily get support. When the correct school environment is nurtured at school and develops to be a school culture, the support base for learners negatively affected by parental migration widens. "When the school has developed the correct environment, everyone has potential to support those who are down" (SPS1). The school would have a wide support base with the learners and teachers equipped with virtues such as empathy and respect.

However, participants noted that when the school environment is not positive the left-behind learners end up being negatively affected in terms of their psychosocial functioning. According to participant SPS1, "the problem at most schools is that left-behind learners are negatively type casted and vilified by teachers, school administrators, and even by their peers, they are labeled troublemakers, truant, and spoilt." This labeling sometimes affects the learners socially and emotionally.

8. Discussion

In the face of challenges associated with parental migration and compounded with the COVID-19 pandemic, the school was revealed as a potential source of support for left-behind learners. The school as an institution has capacity to define what needs to be learned and, in this regard, can teach well-being and coping strategies for learners affected by parental migration. This would be part of what Dewey (1938) described as teaching on life. Appropriate support schemes would facilitate improved psychosocial functioning among learners negatively affected by parental migration.

The school was also revealed as having potential to step into the shoes of the migrating parents. The concept of loco parentis shows that the teacher can step in and provide emotional and social support for learners. Social relationships are an important factor that influences well-being. This concept resembles the responsive

classroom approach in which the teacher facilitates a joyful learning environment in which learners thrive both socially and academically. Such an environment would ensure that the learners thrive socially and academically. The goal is for the school to move from concentrating only on academic subjects but to also facilitate learner well-being.

When learners are equipped with social, emotional, and academic learning skills, the whole learner is educated [28, 29]. It is important that appropriate capacity development programs be in place to professionally develop the teachers for such roles. Negative typecasting of the LBC's negatively impacts on the learners' well-being. The disruptions associated with left-behind learners among other social challenges warrant the need to adopt the concept of educating the whole learner. A socially deprived learner would not thrive academically. Thus, the goals of teaching and learning would not be met. This makes it impossible to separate learner well-being with education.

8.1 Educare in schools

A school is often defined as an institution for the acquisition of knowledge in a formal way under teachers [10]. However, over the years the defining features of the school have been transformed beyond this limit as the expectations of society from it widened. More so, many began to see potential in the school that goes beyond just teaching and learning. According to Dewey [29] it is at a school that we "learn how to live," implying that the school is for the greater good for society. This view regards the school as a dynamic institution that also facilitates the adaptation process among learners. The school is transformed to become a center for both education and care for learners (Educare). Learners are capacitated through the school to cope with their changing circumstances. Educare describes a holistic approach in which education and care are equally important and inseparable [25]. Care is defined as a process of responding to needs, and according to Noddings [28] "needs do not stop at the entrance of a classroom door." The child, who comes to school with unmet needs, would still be deficient when he/she enters the classroom. It is the onus of the teacher to totally ignore such needs or to try and facilitate measures to mitigate. Addressing the psychosocial needs of the learners would also enable the teacher to pursue his/her own teaching objectives. The school environment should be supportive and be ready to take over support when need arises.

For learners left in CHH, the school would provide the much-needed care. The school steps into the roles of the absent parent and provide guidance and supervision for the LBC. The school steps in to provide socially and politically desirable skills, knowledge attributes, and skills among learners [27]. John White in response to his own question on what schools are said schools should be institutions that promote human happiness and well-being [27]. By implication, when life, happiness, and well-being are under threat, the school is expected to step in. The societal expectations of the school are not fixed as change in society is not constant. New threats and challenges to happiness, well-being, and life continue to emerge, and the school is expected to be apt to the changes and challenges. The school is also defined as a self-determining institution. That is to say: it has capacity to "define what needs to be learned and unlearned, and in what ways" [27]. Thus, the school has potential to facilitate the acquisition of skills and outcomes that it wants among its learners. This potential and capacity of the school are what this study believe is essential when support for left-behind learners is envisaged. Different avenues were scouted within

Mitigating the Unmet Psychosocial Support Needs of Left-behind Learners in Child-Headed... DOI: http://dx.doi.org/10.5772/intechopen.111450

the school culture, curriculum, extra-curriculum, school leadership, and classroom management for well-being support for learners affected by parental absence due to migration.

9. Conclusions

The COVID-19 pandemic yielded far-reaching ramifications on left-behind learners who were already made vulnerable by parental migration. The lack of parental protection and support exposes left-behind learners to vulnerable conditions. This, in turn, affects the educational outcomes of the left-behind learners. This, therefore, calls to the return to the philosophy of Dewey in which education goes beyond words and figures but should equip the learner with life skills. It also calls on society to embrace Ubuntu in relating the LBC. This study revealed support in the school being housed within the curriculum, the teachers, and the school environment. It is of importance that the school environment be positive and enabling for learners to thrive. When the environment is positive, and enabling support would be available to everyone. Conversely, when the environment is not conducive and characterized with negative typecasting and labeling of left-behind learners, negative outcomes would be expected. It becomes the key responsibility of the school leadership to nurture a conducive learning environment through continuous engagements. In the face of adversity, the school has potential to be a center for hope.

A widened understanding of the school is vital and relevant in view of the threats posed by globalization, parental migration, and compounded by the COVID-19 pandemic. This paper calls for going back memory lane and adopt Dewey's lenses. The school becomes a place where there is hope, as learners learn life lessons. This transformation is significant considering that most children spend three-quarters of their waking time at school. The teaching staff needs to be capacitated through professional development to effectively play their loco parentis roles in the school realm. The teacher becomes a specialist for both education and care (Educare). Such approaches will surely mitigate the unmet psychosocial needs of left-behind learners in child-headed households in contemporary times.

Acknowledgements

The researchers acknowledge financial support from the Central University of Technology, which funded the research from which this paper was conceived.

Conflict of interest

The authors declare no conflict of interest.

Author details

Ricanos Jaure* and Alfred Henry Makura Central University of Technology, Free State, South Africa

*Address all correspondence to: jaurericanos2019@gmail.com

IntechOpen

© 2023 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited.

References

- [1] Jaure R, Makura AH. Child learners of parents in employment sojourn: Their lived experiences in separation. Journal of Psychology in Africa. 2021;31(1):82-86
- [2] Batisai K. Transnational labour migration, intimacy and relationships: How Zimbabwean women navigate the diaspora. Diaspora Studies. 2016;9(2):165-178. DOI: 10.1080/09739572.2016.1185236
- [3] Jaji R. Transnational migration and reconfiguration of the family in Zimbabwe. REMHU. 2022;**30**(66):227-242. DOI: 10.1590/1980-85852503880006613
- [4] Harima A. Transnational migration entrepreneurship during a crisis: Immediate response to challenges and opportunities emerging through the COVID-19 pandemic. Journal of the W Michael Hoffman Centre for Business Ethics at Bentley University. 2021;127:223-251
- [5] Tong L, Yan Q, Kawachi I. The factors associated with being left-behind children in China: Multilevel analysis with nationally representative data. PLoS One. 2019;14(11):e0224205
- [6] Nyikahadzoi K. Enhancing the welfare of migrant-sending households in Zimbabwe. In: Migrating out of Poverty Policy Brief September, 2019. England: University of Sussex; 2019
- [7] ZIMSTAT. Population preliminary Household Census. 2022. Available from: https://www.zimstat.co.zw/wpcontent/ uploads/2022/07/Census2022_ Preliminary_Report.pdf
- [8] Maushe F, Mugumbate J. "We are on our own": Challenges facing child headed households (CHH), a case of Seke rural area in Zimbabwe. African Journal of Social Work. 2015;5(1):33-60

- [9] Asuncion F-F. The bumpy landscape of family reunification: Experiences of first and 1.5-generation Filipinos in France. Journal of Ethnic and Migration Studies. 2015;**41**(7):1152-1171
- [10] Jaure R. Psychosocial Support for Learners in Child and Youth Headed Households in the Context of Parental Labour Migration: Exploring Stakeholder Based Support in Zimbabwe and South Africa [Thesis]. Bloemfontein: Central University of Technology; 2021
- [11] Fan F, Su L, Gill MK, Birmaher B. Emotional and behavioral problems of Chinese left-behind children: A preliminary study. Social Psychiatry and Psychiatric Epidemiology. 2010;45(6):655-664
- [12] Yoosefi Lebni J, Mohammadi Gharehghani M, Soofizad G, et al. Challenges and opportunities confronting female-headed households in Iran: A qualitative study. BMC Women's Health. 2020;**20**:183. DOI: 10.1186/s12905-020-01046-x
- [13] Cheng J, Sun YH. Depression and anxiety among the left-behind children in China: A systematic review. Child care Health and Development. 2015;**41**(4):515-523
- [14] Manyeruke G, Çerkez Y, Kiraz A, et al. Attachment, psychological wellbeing, and educational development among child members of transnational families. Alpha Psychiatry. 2021;22(1):49-55
- [15] United Nations. Conventions on the Rights of the Child. 1989. Available online: http://www.unicef.org/crc/ [Accessed: October 19, 2022]
- [16] Seepamore BK. Distance parenting implications for social work practice.

- Social Work/Maatskaplike Werk. 2015;**52**(4):7
- [17] Mengtong C, Ling CK. Parental absence, child victimization and psychological well-being in rural China. Child Abuse and Neglect. 2016;59:45-54
- [18] Adhikari R, Jampaklay A, Chamrstrithirong A, Richter K, Pattaravanich U, Vapattanawong P. The impact of parental migration on the mental health of children left behind. Journal of Immigrant Minority Health. 2014;**16**:781-789
- [19] Agere LM, Tanga PT. A critical examination of the nexus between psychosocial challenges and resilience of child headed households: A case of Zola, Soweto. Child Abuse Research in South Africa. 2017;18(2):59-67
- [20] Mazzucato V, Cebotari V, Veale A, White A, Grassi M, Vivet J. International parental migration and the psychosocial wee-being of children in Ghana, Nigeria and Angola. Social Science and Medicine. 2015;132:215-224
- [21] Bakker C, Elings-Pels M, Reis M. The Impact of Migration on Children in the Caribbean. UNICEF Paper No 4. Barbados: UNICEF; 2009
- [22] Cortes P. Adolescents' Rights, Gender and Migration Challenges for Policy Makers. Boston: Boston University; 2011
- [23] Moshiri L. Impact of Labour Migration on Children Left behind. Dushanbe: UNICEF; 2011
- [24] Chademana KE, van Wyk B. Life in a child-headed household: Exploring the quality of life of orphans living in child-headed households in Zimbabwe. African Journal of AIDS Research. 2021;**20**(2):172-180

- [25] Wills G, Patel L, van der Berg S, Mpeta B. Household Resource Flows and Food Poverty during South Africa's Lockdown: Short-Term Policy Implications for Three Channels of Social Protection. National Income Dynamics Study (NIDS) – Coronavirus Rapid Mobile Survey (CRAM). South Africa: Stellenbosch University; 2020. pp. 1-46
- [26] Harima A. Transnational migration entrepreneurship during a crisis: Immediate response to challenges and opportunities emerging through the COVID-19 pandemic. Business and Society Review. 2022;**127**(S1):223-251
- [27] Cole P. Brilliant insights about education from John Dewey. 2016. Available from: https://www. emergingedtech.com/2016/09/5brilliant-insights-about-teaching-fromjohn-dewey/
- [28] Noddings N. The Challenge to Care in Schools: An Alternative Approach to Education. 2nd ed. New York: Teachers College Press; 2005
- [29] Kochhar-Bryant C, Heishman A. An Effective Collaboration for Educating the Whole Child. Thousand Oaks, CA: Corwin Press; 2010

Chapter 8

The Effectiveness of Foster Care Arrangements for Unaccompanied Migrant Children in Zimbabwe: A Case of Children under Foster Care in Dulibadzimu, Beitbrigde

Mulwayini Mundau and Maxwell Chikowe

Abstract

The chapter examines the effectiveness of foster care arrangements for unaccompanied migrant children in Zimbabwe, with a specific focus on children in foster care in Dulibadzimu, Beitbridge. Through a qualitative research approach and an interpretive paradigm, the chapter highlights the formal and informal foster care arrangements for unaccompanied and separated migrant children that are existing as a place of safety, temporal home, community care, and as well an alternative shelter for the border town between Zimbabwe and South Africa. Furthermore, the chapter highlights the successes and possible challenges in dealing with this vulnerable group that is a product of economic migration between the two countries.

Keywords: children, foster care, migrant children, unaccompanied, Zimbabwe

1. Introduction

The phenomenon of unaccompanied and separated migrant children is global, alarming, and escalating. Children on the move are widely confronted with multiple vulnerabilities and nations across the globe have a legal duty to provide social support and care. According to the UNHCR [1], 89, 3 million people around the world have been forced from their homes, and among these, 27, 1 million are refugees with over half of them being unaccompanied and separated children under the age of 18. Additionally, in 2019 there were also millions of children across the globe living outside their nuclear family environment [2].

In relation to Zimbabwe, the prevailing unfavorable socioeconomic circumstances have further made migration of children across national borders a serious challenge. In this regard, it was noted that Orphaned and Vulnerable Children (OVCs) in Zimbabwe have constituted a large number of children on the move [3]. The case with Beitbridge involves an increasing number of migrant children without company and living outside the family environment. And such children are confronted with

97 IntechOpen

multiple vulnerabilities such as sexual and economic exploitation, and trafficking, so their situation requires special attention.

The escalating vulnerabilities for separated children without company raise questions on the effectiveness of the relevant alternative care option which suits their scenarios. This yawning gap in social protection demands an appreciation from social workers and all practitioners in the helping professions. In order to have a better appreciation of foster care, it is necessary to have an exploration of the effectiveness of foster care in the protection of unaccompanied and separated migrant children. The provision of care and support to OVCs has long been a global mandate also for Zimbabwe and therefore are high expectations that such children should be looked after within a family environment and foster care exists as the more recommended care priority option under the six tier system. And also, the international United Nations Convention on the Rights of the Child (UNCRC) and the UN Guidelines for the Alternative Care of Children stand strongly on the placement of children without company into foster care.

In line with the preambles of the UNCRC and the African Charter on the Rights and Welfare of the Child (ACRWC), the constitution in Zimbabwe echoes that children should grow up in a family environment. It is the responsibility of the state to make sure that such entitlements are fulfilled and Zimbabwe has a long history of providing care and support to such children under a cherished tradition of Ubuntu. The strong tradition regards a child as everyone's child although such children belong to their nuclear families. The common adage such as "it takes the whole village to raise a child", presents the strong commitment of Zimbabwean villages to providing care and support to OVCs under various community structures. Foster care, therefore, stands as the predominant form of care in Africa and also in Zimbabwe, however much of such arrangements were informal.

Zimbabwe is confronted with the challenge of separated children without company in migration. These children are prone to facing unprecedented challenges in the face of the migration process. In particular, Beitbridge as a border district town has also received a large number of children transiting through the border. In relation to children on the move, according to International Organization for Migration (IOM) [4], foster care, or family-based care, is widely regarded as the best form of care for unaccompanied migrant children, as fewer children go missing from foster care compared to reception centers or institutions. Additionally, in relation to fostering vulnerable child migrants, IOM [4] notes that it is imperative to consider their special vulnerabilities and needs that are based on their triple characteristics of being children, being in migration, and without company of adults who know them well (their family or legal guardians).

2. The circumstances of unaccompanied and separated migrant children in Beitbridge

The context of Beitbridge, being a border town makes it ideal for fostering services to shelter children who are separated through migration between Zimbabwe and South Africa whilst family tracings and reunifications will be awaited. Beitbridge is also a receiving district of unaccompanied minors coming up from high-sending districts such as Chipinge, Chimanimani, Chiredzi, Gwanda Bulawayo, and Mutasa. Apart from the above, the district is widely known for attractive commercial

activities, the illicit activities include rampant smuggling of contraband such as cigarettes and alcohol to South Africa and the importation of groceries, building materials, and many other supplies into Zimbabwe [5]. Such activities also attract high volumes of children looking for working opportunities. Concurrently the unfolding activities also make Beitbridge a risky place for children as it is highly characterized by high rates of prostitution, crime rates, and drug and substance abuse [3]. Additionally, IOM [5] noted both regular and irregular cross-border movements between Zimbabwe and South Africa for employment, trade, and commerce, seeking health services, family reunions, or illicit activities, which include trafficking in persons, smuggling of goods, and other illicit activities.

It is also imperative to state that Beitbridge is a major transit route for migrants from the Democratic Republic of the Congo who pass through Zambia or Mozambique and then Zimbabwe with the destination intention of South Africa. According to IOM [5], some migrants move from the Horn of Africa, including Ethiopia, Somalia, and Tanzania, moving from Zambia or Malawi and passing through the same Beitbridge district in Zimbabwe. It was estimated that 30–40% of the travelers traveling to South Africa, traverse through this route and hence it is estimated that nearly 6000 travelers use this route on a peak day when the Beitbridge border post is fully operational [5]. The nature of Beitbridge being both a transit, receiving, and destination district for unaccompanied and separated minors makes it a town of migrant children who have attempted to cross the border and failed and also for those who have been deported from South Africa. Such children are found stranded, and resultantly find shelter in foster care centers. This makes it more of an area of focus in relation to child protection and therefore, justifies the importance of foster care arrangements for unaccompanied and separated migrated children.

3. Fostering services and legislation

Fostering unaccompanied and separated migrant children is both a legal and moral responsibility. It is an essential form of alternative care option for OVCs across the globe. The UNCRC and the UN Guidelines for the Alternative Care of Children stand strongly on the placement of children without company into foster care homes. In line with the preambles of the UNCRC and the ACRWC, the constitution in Zimbabwe places a premium on children to grow up in a family environment [6]. In such scenarios, the foster care option is the recommended immediate safety net for children with unclear immediate family status.

In this regard, Zimbabwe's Orphan Care policy (1999), introduced a six-tier system and considers foster care as a fourth priority in the protection of children outside the family environment, with biological family being the first priority followed by extended family then community care, foster care, adoption and institutionalization as the sixth option and a last resort which should be avoided. These environments should be given consideration in their priority order. Despite the significant investments and efforts for years to increase the demand and promote foster care as an ideal form of alternative care, it is associated with low uptake and institutionalization has maintained a grip over foster care for over 20 years after the adoption of the six-tier system in 1999. The study agrees with the definition of the six-tier system as defined by the National Orphan Care policy (1999). The six-tier care environments should be

adopted in their priority order, with the first being the best and the sixth being the last resort and to be avoided.

4. Theoretical framework: fostering and children in need assessment framework

The Children in Need Assessment Framework is an established systematic approach to the process of gathering information about children who may be in need and their families [7]. The assessment framework captures the three main pillars such as family and environmental factors, parental capacity, and children's developmental needs. The needs are looked at as interrelated parts that depend on each other and are critically essential for child development [7]. This framework was developed from the functionalist theory which regards the family as an equally essential system. This framework recognizes that individual needs are unique and therefore interventions should be individually tailored rather than having a universal care plan for OVCs. And also vulnerabilities varies across different age group and circumstance, and also much of them stretch through adult life. Sabates-Wheeler [8], rightly argued that vulnerabilities facing children differ from those of adults and there are child-specific vulnerabilities derived from asymmetrical power relations and inequalities in relation to resources, responsibilities, opportunities, voice, rights, and constraints. In the same view, these scholars further indicated that there are child-intensified vulnerabilities, which refer to vulnerabilities that may affect whole populations, such as migration, famine, or undernutrition, but have effects that are more damaging for children than

In relation to the fostering of unaccompanied and separated migrant children, this model gives a view of how foster care arrangements effectively respond to the safeguarding and protection needs of these children. This assessment model is also essential to social work professionals working with families to inform their decision-making and foster parents in identifying the individual needs of each child. According to Turney et al. [7], studies indicate that good assessment is grounded in a thorough understanding of the child and family's situation, needs, and strengths, and to gain this knowledge, practitioners need to work directly with the child and their family. However, it needs to be supported by appropriate practice and research-based knowledge and tools to support practice [7]. The authors further state that keeping the child or young person 'in view' is fundamental to good assessment, and failure to do so can have severe consequences, as analyses of serious case reviews have consistently demonstrated. This provides essential guidelines in identifying the effectiveness of foster care in the protection of these children such that the effectiveness can be judged in relation to this framework.

Using this Children in Need Assessment Framework in the case of Beitbridge is likely to highlight multiple vulnerabilities that these children are likely to encounter. These may include among others, sexual and economic exploitation, and trafficking. Such proceeding circumstances have posed serious problems and exacerbated risky situations for such minors. The escalating vulnerabilities for these children question the effectiveness of the relevant alternative care option which suits their scenarios. This yawning gap in social protection demands appreciation from social workers. In order to have a better appreciation of foster care, it is necessary to have an exploration of the foster care arrangements in Beitbridge and appreciate their effectiveness in safeguarding and protecting this vulnerable group of children.

5. Methodology

In an endeavor to examine the effectiveness of foster care arrangements for unaccompanied children in the Beitbridge District of Zimbabwe, a qualitative research approach and an interpretivist paradigm were adopted. In this study, an indepth interview guide and a Focus group discussion guide, and a semi-structured Key informant interview guide were used. The adopted research approach and paradigm gave the researchers the freedom to conveniently establish the sample size which they could effectively handle in the context of constrained time and financial resources. The researchers were convinced that 6 Focus group discussions with children, 25 participants for in-depth interviews with foster care parents and caregivers that were purposively drawn from various foster care homes, and 6 key informants would constitute a significant population for the study. Key informants for this study were members of the District Child Protection Committee that were drawn from the Department of Social Welfare (Renamed Department of Social Development) Child Protection Society, Family Support Trust, International Committee of the Red Cross, International Organization for Migration, and the Immigration department.

6. Results

This section of the chapter presents the research findings and the discussion thereof. These are centered on the foster care arrangements in Beitbridge, services available for unaccompanied and separated migrant children under foster and the effectiveness of foster care in the protection of unaccompanied and separated migrant children.

6.1 Foster care arrangements in Beitbridge

This study had a specific objective of identifying foster care arrangements for unaccompanied migrant children in Dulibadzimu, Beitbridge. This study noted that there are formal and informal foster care arrangements for unaccompanied and separated migrant children in Beitbridge. Some of the foster care arrangements are formal and recognized by the Department of Social Development (DSD) while others are informal and unrecognized.

6.1.1 Informal foster care

The study established that there are informal foster care arrangements for unaccompanied and separated migrant children in Beitbridge. In this type of foster community care, DSD is the main actor involved in the placement of child under foster care. Informal foster care identified in this study included kinship and community care whereby migrant children identified without an immediate family status are looked after by relatives, churchmates, and local community leaders traced and identified in Beitbridge, whilst waiting for a permanent solution. The findings capture informal foster care as also pre-stage to formal foster care, as such formal care initially starts as informal arrangements of foster care. In the process, most of the foster parents indicated that they registered after they felt the joy of bonding and blending with migrant children as theirs. From the findings, the cases of unaccompanied and separated children have involved children placed in foster care after

attempting to cross the border and failed. One of the foster care parents from the in-depth interviews said,

"I'm staying with 2 children, the children were received from South Africa, and volunteered to look after them, and these children are also our church mates".

The arrangement identified also involved unaccompanied and separated migrant children being looked after by community volunteers such as Community Case Care workers, local leaders such as councilors, and also Church leaders. The arrangements are widely popular as foster care in Dulibadzimu. One of the caregivers from the in-depth interviews said,

"I volunteered to look after a child after I felt that there is a need, the child was roaming around my house and I felt pity as the child indicated that she couldn't find her parents in South Africa, she decided to come back".

This indicated that migrant children before placement under foster care were identified whilst stranded and roaming on the streets. In addition, many of the children under informal foster care were registered as having biological families in South Africa and untraceable relatives in Zimbabwe. This indicates that children from unstable families are at risk of separation from caregivers and migrate unaccompanied across the national border. The vulnerabilities to which unaccompanied and separated migrant children are exposed motivate the community volunteers to take such children and look after them. This was also echoed by one of the in-depth interview participants, who said,

"I picked the child when he was roaming in the streets and I felt so pity, he looked stranded and reported that he does not know the whereabouts of his parents, he was dumped by omalaicha (transporters) at the border, I decided to look after the child".

From the extract, it can be noted that the motivation behind fostering unaccompanied and separated migrant children in informal care is rooted in some values of love, peace, and caring, in a society in which some caregivers have a positive desire for the betterment of children's lives. Another caregiver from the in-depth interviews confirmed that,

"Sometimes, a child can be identified as unaccompanied and placed under foster through community and then the case will be reported after the child has been placed under foster care."

The findings indicate that some caregivers are also motivated towards fostering non-relatives children as long they share some cultural values and language. Venda people prefer fostering a child speaking Venda, the same also applies to Shona and Ndebele people in Beitbridge. The issue of totem and religious beliefs are also behind the motivation to foster unaccompanied and separated children. Additionally, the study further noted that the nature of Beitbridge being a multi-cultural and also multi-lingual border town makes it a resource for fostering migrant children as they emerge from diverse cultural backgrounds with different languages. The town comprised people speaking different languages such as Venda, Shona, Ndebele, English,

The Effectiveness of Foster Care Arrangements for Unaccompanied Migrant Children... DOI: http://dx.doi.org/10.5772/intechopen.111674

Migrant children placed under informal foster care arrangements also experience love and care from unregistered caregivers. The findings noted that some children have lost bonding connection with their biological parents abroad and tend to develop strong bonds and healthy relationships with their foster caregivers such that they feel that they prefer staying with caregivers their entire life. One of the children that participated in the focus group discussions reported that,

"I see my foster parent as my real mother as she picked me when I had nowhere to go. I attempted crossing over to South Africa and failed and mom took me to stay with her, this is my real mother, with regard to my biological mother in South Africa, I do not even know her number"

Apart from the above, the findings also indicated that some children under informal foster care are subjected to unsafe environments and risky situations such as forced labor. Children raised issues such as forced labor, sexual exploitation, and emotional and physical abuse. This was also echoed by a key informant from DSD, "cases of females minors sexually abused under foster care have been reported and also the informal foster care arrangement are at most not known and undocumented by DSD. This suggests that informal foster care arrangements in Beitbridge also expose children to risky situations and unsafe environments characterized by multiple forms of abuse despite being a resource to shelter unaccompanied and separated migrant children.

6.1.2 Formal foster care

The study findings also show that formal foster care is widely recognized by law and less adopted by the community. It is mostly considered an option after family tracings would have failed or in cases whereby the child is unwilling to be reunified with extended family or under community care. The findings from this study also noted that in Dulibadzimu, Beitbridge, formal foster care is a legal duty for social workers under the Department of Social Development and it is also a specific process with seven main stages. The first stage involves the identification and assessment of unaccompanied migrant child in need of foster care by a social worker, the child can be identified inside border area, or in South Africa by the immigration department or police and then referred to DSD for further assistance. Secondly, the child is placed in a place of safety at Beitbridge reception center (popularly known as IOM Centre) where they are not supposed to stay for more than 7 days without a court order. The third stage then involves the identification of a registered foster parent, a parent can be identified from the DSD register of foster care parents. Afterward assessments of foster the parent will follow as the next stage, looking at the suitability of the environment to the needs of the child. And then they will be an application of court order for placement of a child under foster care, prepared by a social worker writing to the high court. They will be an implementation of a court order through the placement of the child under foster care, this placement comes with an expiry date but can be extended in the event that it expired without a permanent in place. And finally case follow-ups by social work and they should be at least three within a year period.

In the study, it was also noted that the assessments in formal foster care capture critical issues relating to safeguarding and protection of the child such as the needs of the child and also parental capacity of the foster parent. It also captures the child's

developmental needs and capacity to adapt to the environment. This is in agreement with the Children in Need Theoretical Framework which also informed this study that, assessments should look at three main factors such as domains of the children's developmental needs, parenting capacity, and family and environmental factors.

Children under formal foster care are largely known, and recognized by DSD. One of the in-depth interview participants said,

"For me to be a foster parent, I started by going to the Department of Social Development, police services took my fingerprints and the social worker did my home assessments, that's when I qualified to be a foster parent. At first, I first I used to take children without the Department's knowledge and sometimes, the Department would bring children until I had the desire to register."

This was echoed by another interviewee who said that "the Probation officer is the one who gave us children for fostering, and at most I can be given a child together with a court order or I sign an affidavit form" This suggests that foster care involves the registration of a foster parent by the Department of social development and followed by the placement of a child under a foster parent through a court order. From the above probation officer carry out the duties. This agrees with the Children's Act [56.01] which characterizes foster care as involving situations where children are placed by a competent authority, through a court order.

The findings also show that Foster parents received training and support from the government and partners, in order for them to provide quality care to migrant children. The forms of support they received targeted foster parents and also children there are fostering, this included home visits, foster care packages, and access to free health and education services. In this regard, one in-depth interview participant reported that,

"from our training, we learned that formal foster care involves registration of foster parent through the department of social development and placement of a child under foster care by a probation officer through a court order"

Such arrangements identified are in line with the provisions in the country's Children's Act [Chapter 5:06] [9], which states that foster parents should be selected, qualified, approved, and supervised for providing such care. However, one of the foster parents indicated that the training content does not equip them with relevant knowledge of how they should respond to multiple challenges faced in fostering unaccompanied and separated migrant children such as administering corporal punishment and discipline to the child. This was also echoed by all the key informants indicating that such training often does not cover critical issues on how to administer corporal punishment for such children.

6.2 Findings on circumstances of unaccompanied and separated migrant children under foster care

The findings of the study show that children in informal foster and formal foster require special attention. Participants in this study expressed that children under informal foster care were reported to be more vulnerable than children in formal foster care arrangements. In addition, the circumstances of unaccompanied and separated migrant children living under both types of foster care are not similar, as

children under informal foster care were reported to be a high chance of illegal adoption, sexual exploitation, and forced labor. Whilst children under formal foster care are reported to be under better circumstances as their living arrangements are closely monitored by state agencies and partners. In contrast, it was also noted that children under informal foster care have funny and friendlier relationships with their caregivers as compared to children under formal care where many of the participants from the in-depth interviews expressed that "you even fear to laugh or discipline the child, everything is very formal". Notwithstanding that 90% of the key informants indicated that children under formal foster care have better access to child protective services than those under informal foster care.

6.3 Services available for unaccompanied and separated migrant children in foster

The study noted there are specific services available to unaccompanied and separated migrant Children under foster care such as psychosocial support, mental health support, foster care packages, free educational and free health service.

6.3.1 Foster care settling packages

The findings from the study noted that foster care settling packages are part of the once-off package available for children being placed and the package consists of \$75 USD, which caters to hygiene and dignity kits for the child. One in-depth interview participant said, "there are organizations that support foster families with \$75, and such an amount compliments the support needed for the child."

It was also noted that the foster care settling packages were reported to be essential in supporting the reintegration of unaccompanied and separated migrant children into the community. In relation to the number of foster parents accessing such packages, the findings noted 60% of the foster care parents indicated that they have not yet received the packages but they just heard about them. The other 40% of the foster parents who received the packages raised complaints that such packages are given only once off packages and cannot effectively respond to the basic needs of children alone, but rather they will be a need to complement such services. The majority of the participant confirmed that foster care settling packages complement the effort of foster families in looking after fostered children.

6.3.2 Free educational support

The findings also show that children under foster care lose many years of schooling whilst in migration. From this study, it was noted that children who have dropped out of school constitute a large number of children under foster care with a total of six children enrolled in primary schools and then three secondary-going children. The foster care arrangement for unaccompanied and separated migrant children under foster care allows such children to be enrolled back in school. There are educational support services to support in circumstances where the foster parent cannot afford school fees. From these services the findings noted that such children are entitled to free education which covers school fees and uniforms through Basic Education Assistance Module (BEAM) and public assistance programs, it was indicated that children under foster care are a first priority to such services. One of the in-depth interviewees reported that "The government meets educationally related expenses such as school fees and uniforms, but there is low uptake of such services due to children's

unwillingness to return back to school". This extract indicates that children under foster care have access to specific services such as free education and this is consistent with what was identified by Chibwana [6] that the children go to public schools and they rely on either BEAM or school fees paid by the government. Such services are stipulated in the social welfare assistance Act and are largely relevant.

6.3.3 Free health services

From interviews with foster parents, it was noted that children under foster care are entitled to free health services through Assisted Medical Treatment Orders (AMTO). One of the foster care parents in the in-depth interviews said, "With regards to health services, once a child gets sick, that is not an issue, we simply apply for the AMTO at the Department of Social Development, we are likely to be assisted for free" In line with the foster parent's assertion, one child in the focus group discussions echoed that, on one occasion, I got sick, I went to the hospital with my foster mother and she confirmed that social welfare is responsible for payment of such a service.

The extracts indicate that foster families also depend on free services provided by the government to access health. On the other side, foster parents also highlighted accessing health services is a challenge for unaccompanied and separated migrant children under foster care as AMTO was said not to cover the essential medication and pills, at most the medication that can be accessed through AMTO is often unavailable in local clinics.

The findings also show that some foster families have the financial capacity to afford health services for unaccompanied and separated migrant children under foster care. From this study, it was also noted that there are foster families which afford health bills for children under foster care at their own cost. One of the interviewees said, "I consider the child I have as mine and I have the capacity to meet the required bills just like what I do with my biological children under my parental care". This indicates variations in the parental capacities of foster parents. Indicating that foster care parents have different parental capacity, some depends on state resources whilst some compliments were state services cannot reach.

The findings therefore suggest that foster homes have different capacities and offer different quality of life to children depending on the household income, some households depend on public assistance whilst some can sustain within their means to look after the child under foster care. This feeds well with what was noted by Chibwana [6] that, some foster families procure private medical facilities therefore they would not need the AMTOs it is well within their means to take care of the health needs of the fostered child. Henceforth foster care's responses to basic child protection needs vary from household to household, some families can afford to provide for the child whilst some families depends on public assistance.

6.3.4 Public assistance

The study further noted that families fostering children also have a first priority in public programs and also the Drought Relief Programmes (DRP). Such provisions were identified as complimentary as fostering unaccompanied and separated migrant children is still a voluntary act than a well-paid activity. In support of this one of the in-depth interviews, participants said, "Sometimes we are given rice and a bag of mealies through the drought relief program, we are not left behind, the Department of social development has promised us to receive a portion."

From the findings, it can be noted that nutritional provision comes from both foster care households and also government and its partners. The quality of nutrition in foster homes varies from household to household depending on the household's capabilities. The findings of 80% of the children indicated that they do not access balanced nutrition whilst under foster care and at most they eat twice a day, whilst the other 20% indicated that they access food at any given time under foster care. The nutritional support provided does not fully suit children, particularly those under the age of 10 who demands special meals such as porridge and other supplementary meals recommended for children.

6.4 Foster care arrangements for unaccompanied children from the Children in Need Assessment Framework's lens

At the heart of the triple focused Children in Need Assessment Framework's principles of family and environmental factors, parental capacity and children's developmental needs is the safeguarding and promotion of the child's welfare. The framework provides a holistic and interrelated view of the parts that are critical to a child's development. In relation to the study at hand, this framework was of critical relevance in bringing out what exactly to focus on when exploring these foster care arrangements for unaccompanied and migrant children's welfare needs.

It is imperative to highlight that the framework promotes the use of relationships with other assessment processes and tools in the assessment of the child's needs. In this regard, the study relied on various sources to obtain a clear understanding of the inter-relationship between a child's needs, parents' capacities, and the impact of family and environmental factors according to the United Kingdom's Department of Health [10]. For the purposes of this study, information was sourced from foster parents, caregivers as well as key informants that also included key government departments like the DSD.

In line with this framework, the dire need situation of these children has been made clear. The various forms of care that children are exposed to are likely to affect them either positively or negatively. The study's findings show that there are two common forms of care in Beitbridge and these are informal and formal care systems. Within the context of the Children in Need Assessment Framework, the informal care system is likely to expose children's various developmental needs, parental capacity as well as family and environmental factors.

Furthermore, the framework provides the linkages between the child welfare providers. In this regard, there are collaborative efforts between the foster care homes and the DSD (Public Assistance program), Ministry of Health (providing AMTOs), Ministry of Education (BEAM program), and other private organizations that are committed to the provision of child welfare services. However, the findings point to the shortage of resources and failure to attend to special needs like for those below the age of 10 years. There is therefore a need to broaden the levels of collaboration in order to cater to all the children's needs.

The study is therefore of great significance in highlighting the multidimensional needs for children's development. It is, therefore, critical for the Zimbabwean government to address the emergency of informal care systems for unaccompanied and migrant children in Beitbridge and all other districts with the same challenges. In addition, the study also highlights the inadequacy of services provided to these children. It is therefore imperative for the country to ensure the provision of all the

basic needs that contribute to a child's holistic development in collaboration with all critical stakeholders in the child welfare sector.

7. Conclusion

In summation, there are formal and informal foster care arrangements for unaccompanied and separated migrant children. The arrangement in Beitbridge involved children placed under foster care as a temporal arrangement whilst waiting for family tracing outside Beitbridge and in some scenarios involved children being fostered by extended families, local leaders, and churchmates. From the arrangement, it was established that foster parents prefer to foster children with whom they share similar totem, church, and languages. This is well in line with existing literature that, most of the urbanites who foster children are not bound by the traditional beliefs which are inhibitive to the concept of foster care [6]. Foster care arrangements differ extensively in the quality of care they offer to children. The foster care arrangements depend on government-sponsored services and also on their own resources to look after the children there are fostering. From the services it was noted that foster families that depend on state provisions are compromised on the quality of care they offer to unaccompanied and separated migrant children, whilst foster families that depend on their own provide better quality care to fostered children. The study also finds out that services around foster care arrangements are fragmented, less coordinated, and undocumented, this tends to make foster care unaccountable and compromise the quality of care for children in foster care.

Acknowledgements

The researchers would like to express their heartfelt acknowledgments to the Ministry of Social Development in the Beitbridge District, all the child welfare organizations, and the individual participants that volunteered to participate in this study.

Conflict of interest

The authors declare no conflict of interest.

| The Effectiveness of Foster Care Arrangements for Unaccompanied Migrant Children DOI: http://dx.doi.org/10.5772/intechopen.111674 |
|--|
| |
| |

Author details

Mulwayini Mundau^{1*} and Maxwell Chikowe²

- 1 Department of Sociology and Social Work, National University of Lesotho, Maseru, Lesotho
- 2 Department of Social Work, University of Zimbabwe, Harare, Zimbabwe
- *Address all correspondence to: lmundau@gmail.com

IntechOpen

© 2023 The Author(s). Licensee IntechOpen. This chapter is distributed under the terms of the Creative Commons Attribution License (http://creativecommons.org/licenses/by/3.0), which permits unrestricted use, distribution, and reproduction in any medium, provided the original work is properly cited. [CC] BY

References

- [1] United Nations High Commissioner for Refugees (UNHCR). Global Trends: Forced Displacements in 2016. 2016. Available from: https://www.unhcr. org/5943e8a34.pdf
- [2] United Nations Children's Fund (UNICEF). State of the World's Children: Excluded and Invisible. New York, USA. 2019. Available from: https:// www.unicef.org/sowc06/pdfs/sowc06_ fullreport.pdf
- [3] Save The Children. Protecting Children on the Move: A Guide to Programming for Children Affected by Migration and Displacement. 2018. Available from: https://resourcecentre.savethechildren.net/node/13638/pdf/children_on_the_move_programme_guide.pdf
- [4] International Organization for Migration (IOM). Handbook on Unaccompanied and Separated Children. 2017. Available from: https://www.iom. int/sites/default/files/HANDBOOK-WEB-2017-0322.pdf
- [5] International Organization for Migration (IOM). Population Mobility Mapping Zimbabwe | Beitbridge. 2020. Available from: https://displacement.iom.int/system/tdf/reports/POPULATION%20 MOBILITY%20MAPPING.%20 ZIMBAWE%20%20BEITBRIDGE. pdf?file=1&type=node&id=10300
- [6] Chibwana MWT. Foster care ontologies: A qualitative study in Zimbabwe. Journal of African Studies and Development. 2019;**11**(6):118-131
- [7] Turney D, Platt D, Selwyn J, Farmer E. Social Work Assessment of Children in Need: What do we know?

- Messages from Research. Research Brief. School for Policy Studies, University of Bristol. 2011
- [8] Sabates-Wheeler R. Migration and Social Protection: A Concept Paper. Development Research Centre on Migration, Globalisation and Poverty Arts C-226, University of Sussex. Brighton BN1 9SJ. 2010
- [9] Children's Act [Chapter 5:06] of 1997. From: https://www.justice.gov/sites/default/files/eoir/legacy/2013/11/08/childrens_act.pdf
- [10] Department of Health, UK. Framework for the Assessment of Children in Need and their Families Guidance Notes and Glossary for: Referral and Initial Information Record, Initial Assessment Record and Core Assessment Records The Stationery Office Ltd. 2000. Available from: https://bettercarenetwork.org/sites/default/files/Framework%20for%20the%20 Assessment%20of%20Children%20 in%20Need%20and%20Their%20 Families%20-%20Guidance%20 Notes%20and%20Glossary.pdf



Edited by Teresa Silva

This book debates the high impact modern societies have on the way we raise children. Although problems such as family dysfunction, work-family imbalance, and migration due to war, violence, and poverty are not new, their consequences for children's well-being and mental health are aggravated by the lack of effective social support networks affecting many children and families living in contemporaneous urban areas. The proverb "It takes a village to raise a child" is as valid now as it was in the early history of less complex communities. However, extended families and the social environment of villages have been substituted by a system of welfare and childcare institutions that, in many cases, fail to provide the appropriate care, education, and support the children need. Job-demanding competitive societies, where career achievement and wealth become the definition of success, force parents to the duality of choosing between family and career and depending on others to parent their children. Likewise, social inequality compels many parents to work in neverending shifts that add to the hours they spend commuting to their workplaces. Sometimes, parents are forced to migrate, leaving their children behind. Children learn to survive in the absence of their parents and to deal with small or inexistent parental emotional investment. The parent-child relationship and attachment necessities are impacted in ways that will affect children for the rest of their lives. Alternatively, migrant children accompanying their parents to a new host country may feel the shock of a normative society with cultural values different from the ones they left behind. Parenting behaviour and style may then be considered inappropriate, challenging parents' ability to educate and pass their values to the offspring. This book is an academic reflection on these controversies.

Katherine K.M. Stavropoulos, Education and Human Development Series Editor

Published in London, UK

- © 2023 IntechOpen
- © NeoLeo / iStock

IntechOpen



